UNITED STATES BANKRUPTCY COURT NORTHERN DISTRICT OF GEORGIA ATLANTA DIVISION

In re:)	Chapter 11
OTB HOLDING LLC, et al.,1)	Case No. 25-52415 (SMS)
Debtors.)))	(Joint Administration Requested)

DEBTORS' EMERGENCY MOTION FOR INTERIM AND FINAL ORDERS
(I) AUTHORIZING THE DEBTORS TO OBTAIN SENIOR SECURED
SUPERPRIORITY POSTPETITION FINANCING; (II) GRANTING (A) LIENS AND
SUPERPRIORITY ADMINISTRATIVE EXPENSE CLAIMS AND (B) ADEQUATE
PROTECTION; (III) AUTHORIZING USE OF CASH COLLATERAL;
(IV) MODIFYING THE AUTOMATIC STAY; (V) SCHEDULING A FINAL HEARING;
AND (VI) GRANTING RELATED RELIEF

The above-captioned debtors and debtors in possession (collectively, the "<u>Debtors</u>") file this Emergency Motion for Interim and Final Orders (I) Authorizing the Debtors to Obtain Senior Secured Superpriority Postpetition Financing; (II) Granting (A) Liens and Superpriority Administrative Expense Claims and (B) Adequate Protection; (III) Authorizing Use of Cash Collateral; (IV) Modifying the Automatic Stay; (V) Scheduling a Final Hearing; and (VI) Granting Related Relief (this "Motion"). The Debtors seek entry of an interim order, substantially in the form attached hereto as <u>Exhibit A</u> (the "Interim DIP Order"), and a final order (a) authorizing the Debtors to obtain senior secured, superpriority, postpetition financing (the "<u>DIP Financing</u>");

The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number, include: OTB Holding LLC (3213), OTB Acquisition LLC (8500), OTB Acquisition of New Jersey LLC (1506), OTB Acquisition of Howard County LLC (9865), Mt. Laurel Restaurant Operations LLC (5100), OTB Acquisition of Kansas LLC (9014), OTB Acquisition of Baltimore County, LLC (6963). OTB Holding LLC's service address is One Buckhead Plaza, 3060 Peachtree Road, NW, Atlanta, GA 30305.



(b) authorizing the Debtors to use Cash Collateral (as defined below); (c) granting liens and affording superpriority claims status with respect to such postpetition financing; (d) approving the form of adequate protection to be provided by the Debtors; (e) modifying the automatic stay to the extent necessary to effectuate the terms of the Proposed Orders (as defined below) and the DIP Loan Documents (as defined below); (f) scheduling a final hearing (the "Final Hearing") to consider entry of a final order (the "Final Order" and, together with the Interim DIP Order, the "Proposed Orders") approving the DIP Financing on a final basis; and (g) granting related relief in connection with the DIP Financing. In support of this Motion, the Debtors have filed contemporaneously herewith (a) the Declaration of Jonathan Tibus in Support of the Debtors' Motion for Interim and Final Orders (I) Authorizing the Debtors to Obtain Senior and Junior Secured Superpriority Postpetition Financing; (II) Granting (A) Liens and Superpriority Administrative Expense Claims and (B) Adequate Protection; (III) Authorizing Use of Cash Collateral; (IV) Modifying the Automatic Stay; (V) Scheduling a Final Hearing; and (VI) Granting Related Relief (the "Tibus Declaration" and, together with the First Day Declaration (as defined below), the "Declarations"), a copy of which is attached hereto as **Exhibit B**, and (b) the First Day Declaration, each of which are incorporated herein by reference.

JURISDICTION AND VENUE

1. This Court has jurisdiction over these cases (these "Chapter 11 Cases") and this matter pursuant to 28 U.S.C. §§ 157 and 1334. This is a core proceeding within the meaning of 28 U.S.C. § 157(b)(2)(A). Venue is proper in this district pursuant to 28 U.S.C. §§ 1408 and 1409(a).

2. The bases for the relief requested herein are sections 105(a), 345(b), 363, 364, 1107, and 1108 of title 11 of the United States Code (the "Bankruptcy Code"), Rules 6003 and 6004 of the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"), and Rules 7007-1, 9013-1, 9013-4, and 9014-2 of the Local Bankruptcy Rules (the "Local Rules") for the United States Bankruptcy Court for the Northern District of Georgia (this "Court"), and Procedure K(1) of the Second Amended and Restated General Order 26-2019, Procedures for Complex Chapter 11 Cases, dated February 6, 2023 (the "Complex Case Procedures").

RELIEF REQUESTED²

- 3. By this Motion, the Debtors request entry of the Proposed Orders:
- A. authorizing the Debtors to incur senior secured postpetition obligations on a superpriority basis in respect of a senior secured superpriority delayed multiple-draw term loan facility (the "DIP Facility") in the aggregate principal amount of \$14 million plus all interest, fees and expenses that will be capitalized as part of the aggregate outstanding principal amount (the "DIP Loans"), comprised of: (a) upon entry of the Interim DIP Order, \$11.5 million, including \$7.5 million of new money DIP Loans (the "Interim Advance"), plus a roll-up and conversion into DIP Loans of \$4 million of the outstanding principal balance (plus accrued interest) under the Prepetition Secured Note (as defined below) (the "Roll-Up"), plus capitalization of the DIP Facility Fee (defined below) and (b) upon entry of a Final DIP Order, an additional \$2.5 million of new

A description of the Debtors' businesses, the reasons for commencing these chapter 11 cases, the relief sought from the Court to allow for a smooth transition into chapter 11, and the facts and circumstances supporting this motion are set forth in the *Declaration of Jonathan M. Tibus in Support of Chapter 11 Petitions and First Day Pleadings* (the "First Day Declaration"), filed contemporaneously herewith. Capitalized terms used in this Relief Requested section of the Motion but not otherwise defined therein shall have the meanings ascribed to such terms later in the Motion, and any other capitalized terms used herein but not otherwise defined shall have the meanings ascribed to such terms in the First Day Declaration or the DIP Credit Agreement, as applicable.

money DIP Loans (for a total of \$10 million of new money DIP Loans (including amounts deemed to have been borrowed to pay the DIP Facility Fee)), pursuant to the terms and conditions of the Interim DIP Order, the Final DIP Order, and a Secured Super-priority Debtor-in-Possession Credit Agreement, substantially in the form annexed hereto as Exhibit A to the Interim Order (as the same may be amended, restated, supplemented, waived, or otherwise modified from time to time, consistent with the terms of the Interim DIP Order and any Final DIP Order, the "DIP Credit Agreement"), by and among OTB Acquisition LLC, as borrower (the "DIP Facility Borrower"), and its affiliates who are Debtors in these Chapter 11 Cases, as guarantors (such guarantors together with the DIP Facility Borrower, the "DIP Facility Loan Parties" and, each, a "DIP Facility Loan Party"), and OTB Lender, LLC, an affiliate of Pappas Restaurants, Inc. (the "DIP Lender");

- B. authorizing the Debtors to execute and deliver the DIP Credit Agreement in form and substance consistent with the terms of the Interim DIP Order, and any other agreements, instruments, pledge agreements, guarantees, fee letters, control agreements, and other ancillary documents related thereto (including any security agreements, intellectual property security agreements, or notes) (as amended, restated, supplemented, waived, and/or modified from time to time, collectively with the DIP Credit Agreement, the "DIP Loan Documents") and to perform such other acts as may be necessary or desirable in connection with the DIP Loan Documents;
- C. authorizing the Debtors to borrow the Interim Advance, which shall be available upon entry of the Interim DIP Order to avoid immediate and irreparable harm;
- D. approving an upfront commitment fee in an amount equal to 0.50% of the \$14 million DIP Facility, which such amount shall be deemed to have been borrowed concurrently

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 5 of 171

with the Interim Advance and such amount will be capitalized and added to the principal amount outstanding under the DIP Facility (the "DIP Facility Fee");

- E. subject to and subordinate in all respects to the Carve-Out (as defined below), granting the DIP Facility and all obligations owing thereunder and under, or secured by, the DIP Loan Documents, to the DIP Lender (collectively, and including all "Obligations" to be defined in the DIP Credit Agreement, the "DIP Obligations") allowed superpriority administrative expense claim status in each of the Chapter 11 Cases or in any case under chapter 7 of the Bankruptcy Code upon the conversion of any of the Chapter 11 Cases, or in any other proceedings superseding or related to any of the foregoing (collectively, the "Successor Cases");
- F. subject to the Carve-Out, granting to the DIP Lender automatically and validly perfected security interests in and liens on all of the DIP Collateral (as defined below), including all of the Debtors' cash constituting cash collateral (the "Cash Collateral");
- G. authorizing and directing the Debtors to pay the principal, interest, premiums, fees, expenses, and other amounts payable under the DIP Loan Documents as such become earned, due and payable, including the DIP Facility Fee, any administrative agent's fees, the reasonable and documented fees and disbursements of the DIP Lender's attorneys (including Porter Hedges LLP), advisors, accountants, and other consultants, all to the extent provided in, and in accordance with, the DIP Loan Documents;
- H. authorizing the Debtors to use the Prepetition Collateral (as defined below), including the Cash Collateral of the Prepetition Bridge Lender (as defined below) under (a) each of the agreements, instruments, and other documents executed in connection with the Prepetition Secured Note, and (b) the Prepetition Secured Note;

- I. authorizing the Debtors to use the assets ("CrossFirst Collateral") subject to junior liens held by CrossFirst Bank ("CrossFirst"), including the cash collateral of CrossFirst, and providing adequate protection to CrossFirst for any diminution in value of the CrossFirst Collateral from and after the Petition Date to the extent such diminution in value occurs on account of the Debtors' sale, lease or use of the CrossFirst Collateral, the priming of the liens securing the obligations to CrossFirst (the "CrossFirst Liens"), and the imposition of the automatic stay pursuant to section 362 of the Bankruptcy Code ("Diminution in Value");
- J. subject to and effective upon entry of the Final DIP Order, authorizing a waiver of (a) the Debtors' right to surcharge any collateral pursuant to sections 105(a) and 506(c) of the Bankruptcy Code or otherwise and (b) the equitable doctrine of marshaling and other similar doctrines;
- K. modifying the automatic stay imposed by section 362 of the Bankruptcy

 Code to the extent necessary to implement and effectuate the terms and provisions of the DIP Loan

 Documents and the Interim DIP Order; and
- L. scheduling a final hearing (the "<u>Final Hearing</u>") to consider the relief requested in the Motion and approving the form of notice with respect to the Final Hearing.

Concise Summary of Terms of DIP Facility

4. Under the disclosure requirements of Bankruptcy Rule 4001(b), (c) and (d) and Section G.1.b of the Complex Chapter 11 Procedures, the following tables concisely summarize the significant terms of the DIP Facility and the Interim Order:³

This summary, and all other descriptions contained in this Motion, including the defined terms it uses (whether or not defined within the summary), are qualified in their entirety by the provisions of the DIP Loan Documents and the Interim Order, as applicable. To the extent that there are any conflicts between this Motion, on the one

Summary of Relevant Provisions		
Material Terms	DIP Facility	
Borrowers Bankruptcy Rule 4001(c)(1)(B)	OTB Acquisition LLC See DIP Credit Agreement Schedule 1.1 (Definition of "Borrower")	
Guarantors Bankruptcy Rule 4001(c)(1)(B)	All Debtors, other than the DIP Facility Borrower See DIP Credit Agreement Schedule 1.1 (Definition of "Guarantor")	
DIP Agents Bankruptcy Rule 4001(c)(1)(B)	N/A	
DIP Lenders Bankruptcy Rule 4001(c)(1)(B)	OTB Lender, LLC See DIP Credit Agreement Schedule 1.1 (Definition of "DIP Lender")	
Amount and Facilities Bankruptcy Rule 4001(c)(1)(B)	A super-priority multi-draw term loan credit facility in an aggregate principal amount of \$14 million in term loan commitments (plus all interest and fees that will be paid in kind), comprised of: (a) upon entry of the Interim DIP Order, \$11.5 million, including \$7.5 million of new money DIP Loans (the "Interim Advance") plus a roll-up and conversion into DIP Loans of \$4 million of the outstanding principal balance (plus accrued interest) under the Prepetition Secured Note (as defined below) (the "Roll-Up"), and (b) the remainder \$2.5 of new money available upon entry of the Final Order.	
	See DIP Credit Agreement § 2.1; Interim Order ¶1.	
Funding Use of Proceeds Bankruptcy Rule 4001(c)(1)(B)	Subject to the Approved Budget (subject to permitted variances and other exclusions set forth in the DIP Loan Documents), the DIP Term Facility shall be available to (a) pay transactional fees, costs, and expenses incurred in connection with the DIP Loan Documents and the transactions contemplated thereby, (b) fund general corporate needs, including, without limitation, working capital needs, (c) pay administrative expenses of the Chapter 11 Cases, including fees and expenses of professionals, and (d) to repay the Prepetition Bridge Loans in full, together with all accrued and unpaid interest under the Prepetition Secured Note. See DIP Credit Agreement § 5.13 (Use of Proceeds)	

hand, and any DIP Loan Documents or the Interim Order, as applicable, on the other, the terms of such DIP Loan Documents or the Interim Order, as applicable, shall govern.

Summary of Relevant Provisions		
Material Terms	DIP Facility	
Interest and Fees Bankruptcy Rule 4001(c)(1)(B)	Under the DIP Loan Documents, the Debtors have agreed, subject to Court approval, to pay certain fees to the DIP Lender. In particular, the Debtors have agreed to pay:	
	• <u>Interest Rate</u> : The loans under the DIP Facility will bear interest at a rate of 12% percent per annum. All accrued interest will be paid-in-kind and capitalized on the first business day of each month (and thereby added to principal, which shall thereafter accrue interest).	
	Default Rate: Effective immediately upon the occurrence of an Event of Default, interest on the outstanding loans under the DIP Facility shall accrue interest at a rate that is 15% per annum.	
	• <u>DIP Facility Fee</u> : 0.50% paid in kind.	
	See DIP Credit Agreement §§ 2.4 (Interest Rate: Rates, Payments, and Calculations) and 2.7 (Commitment Fee)	
Maturity Bankruptcy Rule 4001(c)(1)(B)	The maturity date of the DIP Term Loans under the DIP Facility is the earliest to occur of: (i) May 30, 2025; and (ii) the acceleration of all DIP Obligations in accordance with the terms of the DIP Facility.	
	See DIP Credit Agreement Schedule 1.1 (Definition of "Maturity Date")	
Mandatory Prepayments Bankruptcy Rule 4001(c)(1)(B)	Asset sale Net Cash Proceeds in excess of \$500,000 and Net Cash Proceeds from the incurrence of any debt not permitted by the DIP Loan Documents or issuance of shares of Capital Stock. Prepayments shall be in an amount that is an integral multiple of \$25,000 and not less than \$50,000.	
	See DIP Credit Agreement §§ 2.3(c)(i) (Voluntary Prepayments) and 2.3(c) (ii) (Mandatory Prepayments)	
Security and Priority Bankruptcy Rule 4001(c)(1)(B)(i)	All DIP Obligations shall be secured by the following liens and security interests (the " <u>DIP Liens</u> ") on all assets and properties of the Debtors (whether tangible, intangible, real, personal or mixed), whether now owned or hereafter acquired and wherever located, before or after the Petition Date, including, without limitation, all cash, cash equivalents, accounts, inventory, equipment, equity interests or capital stock in subsidiaries, investment property, instruments, chattel paper, real property, leasehold interests, contracts, patents, copyrights, trademarks	

Summary of Relevant Provisions		
Material Terms	DIP Facility	
	and other general intangibles, receivables, all claims or causes of action (excluding avoidance actions arising under Chapter 5 of the Bankruptcy Code but including all proceeds thereof) and all products, offspring, profits and proceeds of each of the foregoing (collectively, the "DIP Collateral"; provided that notwithstanding the definition of DIP Collateral, (i) DIP Liens shall only attach to the Debtors' liquor licenses to the extent permitted by applicable non-bankruptcy law, (ii) to the extent applicable non-bankruptcy law does not permit the DIP Liens to attach directly to the Debtors' liquor licenses, the DIP Liens shall be valid, binding, continuing, enforceable, non-avoidable, and automatically and properly perfected security interests and liens in any proceeds from any sale or other disposition of the Debtors' liquor licenses and such proceeds shall constitute DIP Collateral and (iii) in no event shall DIP Collateral include any leasehold interest in non-residential real property that prohibits or restricts the granting of liens thereon (except as permitted pursuant to applicable non-bankruptcy law), but DIP Collateral shall include the proceeds of the sale or other disposition of such leases):	
	(a) secured pursuant to section 364(c)(2) of the Bankruptcy Code, by a first-priority perfected lien on, and security interest in, all present and after-acquired property of the Debtors, wherever located, that is not subject to a perfected lien or security interest on the Petition Date;	
	(b) secured pursuant to section 364(c)(3) of the Bankruptcy Code, by a junior perfected lien on, and security interest in, all present and after-acquired property of the Debtors, wherever located, that is subject to a valid, perfected, enforceable and unavoidable lien or security interest (excluding the CrossFirst Liens) on the Petition Date or subject to a lien or security interest (excluding the CrossFirst Liens) in existence on the Petition Date that is perfected subsequent thereto as permitted by section 546(b) of the Bankruptcy Code (including, without limitation, the Carve-Out (as defined below)), in each case, that is expressly permitted to be senior to the DIP Liens pursuant to the Interim DIP Order or the DIP Credit Agreement;	
	(c) secured pursuant to section 364(d)(1) of the Bankruptcy Code, a first-priority, perfected senior priming lien on, and security interest in, all present and after-acquired property of the	

Summary of Relevant Provisions		
Material Terms	DIP Facility	
	Debtors, wherever located, that is or was subject to a perfected lien or security interest on the Petition Date (including the CrossFirst Liens but excluding the liens and security interests set forth in clause (b) of this sub-paragraph); and	
	(d) other than as set forth herein (including with respect to the Carve-Out), the DIP Liens shall not be made subject to or <i>pari passu</i> with any lien or security interest heretofore or hereinafter granted in the Chapter 11 Cases or any Successor Cases and shall be valid and enforceable against any trustee appointed in the Chapter 11 Cases or any Successor Cases, upon the conversion of any of the Chapter 11 Cases to any Successor Case, and/or upon the dismissal or conversion of any of the Chapter 11 Cases or Successor Cases. The DIP Liens shall not be subject to any of sections 510, 549, or 550 of the Bankruptcy Code. No lien or interest avoided and preserved for the benefit of the estate pursuant to section 551 of the Bankruptcy Code shall be <i>pari passu</i> with or senior to the DIP Liens.	
	The DIP Liens shall not extend to, and the DIP Collateral shall not consist of, avoidance actions brought pursuant to Chapter 5 of the Bankruptcy Code or applicable state law equivalents but they shall extend to proceeds thereof. The DIP Liens and security interests of the	
	DIP Lender shall not prime any Permitted Liens (as defined in the DIP Credit Agreement).	
	See DIP Credit Agreement Schedule 1.1 (Definition of "Collateral") and § 2.8 (Security Interest; Superpriority Claim and Liens; Collateral); Interim Order ¶5	
DIP Budget Bankruptcy Rule 4001(c)(1)(B)	The use of proceeds of the DIP Facility is subject to an agreed upon 13-week budget as extended in accordance with the DIP Documents (as updated from time to time, the "Approved Budget"), subject to permitted variances and other permitted exclusions as set forth in the DIP Loan Documents.	
	See DIP Credit Agreement §§ 5.13 (Use of Proceeds), 5.19 (Approved Budget; Variance Reports) and Schedule 1.1 (Definition of "Approved Budget"); Interim Order ¶12)	

Summary of Relevant Provisions		
Material Terms	DIP Facility	
Reporting Bankruptcy Rule 4001(c)(1)(B)	The DIP Credit Agreement requires compliance with certain periodic reporting covenants including monthly and quarterly financial statements, the Approved Budget, and variance reports.	
	See DIP Credit Agreement § 5.1 (Reports and Information)	
Borrowing Conditions Bankruptcy Rule	The DIP Credit Agreement includes customary conditions of borrowing, the satisfaction of which are a condition precedent to the obligations of each DIP Lender to make DIP Loans, including:	
4001(c)(1)(B)	(a) The Petition Date shall have occurred.	
	(b) The Interim Order shall have been entered.	
	(c) No trustee under chapter 7 or chapter 11 shall have been appointed in the Bankruptcy Cases.	
	See DIP Credit Agreement § 3.1 (Conditions Precedent to Effective Date and the Advances)	
Stipulations as to Prepetition Claims and Liens Bankruptcy Rule 4001(c)(1)(B)(iii)	After consultation with their attorneys and financial advisors, and without prejudice to the rights of parties in interest, the Debtors, on their behalf and on behalf of their estates, admit, stipulate, acknowledge, and agree to certain stipulations regarding the validity and extent of the Prepetition Bridge Lender's claims and liens.	
	See Interim Order ¶ F	
Effect of Stipulations	The Debtors' Stipulations shall be binding on the Debtors in all circumstances upon entry of the Interim Order.	
Bankruptcy Rule 4001(c)(1)(B)(iii)	See Interim Order ¶ 28	
Adequate Protection Bankruptcy Rule 4001(c)(1)(B)(ii)	CrossFirst is entitled, pursuant to sections 361, 362, 363(c)(2), and 363(e), of the Bankruptcy Code, to adequate protection of its interests in the applicable CrossFirst Collateral, including Cash Collateral, solely to the extent of any Diminution in Value of its interests in the CrossFirst Collateral.	
	As adequate protection (the " <u>Adequate Protection</u> "), CrossFirst shall receive, effective and perfected upon the Petition Date and without the necessity of the execution by the Debtors of security agreements, pledge agreements, mortgages, financing statements, or other agreements, a valid, perfected replacement security interest in and lien on the CrossFirst Collateral (the " <u>Adequate Protection Liens</u> "), which	

Summary of Relevant Provisions		
Material Terms	DIP Facility	
	security interests and liens shall be subject to and subordinate in all respects to the Carve-Out, the DIP Liens, and any liens or security interests that are valid, perfected, enforceable and unavoidable as of the Petition Date. The Adequate Protection Liens shall be in addition to all valid and enforceable liens and security interests now existing in favor of CrossFirst and not in substitution therefor. The Adequate Protection Liens shall not attach to Avoidance Actions or the proceeds thereof.	
	See Interim Order ¶ 11	
Covenants Bankruptcy Rule 4001(c)(1)(B)	Affirmative Covenants: Usual and customary covenants for financings of this type, including, without limitation, (a) reporting requirements, (b) delivery of certain compliance certificates, notices, reports and filings, (c) preservation of existence, (d) compliance with applicable laws, (e) payment of post-petition obligations, (f) maintenance of property and insurance, (g) keeping of books and records, (h) use of proceeds, (i) further assurances regarding collateral and guarantors, (j) compliance with the Milestones (as defined below), and (k) delivery of the Approved Budget and variance reporting.	
	See DIP Credit Agreement Art. 5 (Affirmative Covenants)	
	Negative Covenants: Usual and customary covenants for financings of this type, including, without limitation, restrictions on: (a) indebtedness, (b) liens and guaranties, (c) investments, (d) disposal of assets, (e) restricted payments and payments in respect of other indebtedness, (f) transactions with affiliates, (g) use of proceeds, (h) post-petition claims, and (i) compliance with the Approved Budget (subject to permitted variances and exclusions).	
	See DIP Credit Agreement Art. 6 (Negative Covenants)	
Events of Default Bankruptcy Rule 4001(c)(1)(B)	The DIP Credit Agreement and Interim Order contain events of default that are usual and customary for debtor-in-possession financings, including without limitation, the failure to comply with the Approved Budget covenants, and any Milestone.	
	See DIP Credit Agreement Art. 8 (Events of Default); Interim Order ¶ 16	
DIP Termination Date Bankruptcy Rule 4001(c)(1)(B)	Upon the occurrence of an Event of Default and after the transmission of written notice thereof to counsel for the Debtors, any Committee, and the U.S. Trustee stating an occurrence of an Event of Default, the DIP Lender, in its sole discretion, may: (i) cease making DIP Facility	

Summary of Relevant Provisions			
Material Terms	DIP Facility		
	advances to the Debtors; (ii) terminate the Debtors' use of the DIP Collateral (including, without limitation, Cash Collateral), and/or (iii) immediately terminate the DIP Facility and demand immediate repayment, in cash, of the DIP Facility obligations then outstanding. See Interim Order ¶ 23		
Carve-Out	The Carve-Out shall be, collectively,		
Carve-Out Bankruptcy Rule 4001(c)(1)(B)	 (i) all fees required to be paid to the Clerk of the Court and to the U.S. Trustee pursuant to 28 U.S.C. §1930(a) plus interest at the statutory rate, if any, pursuant to 31 U.S.C § 3717 (without regard to the Carve-Out Trigger Notice (as defined below)), (ii) reasonable fees and expenses incurred by a trustee and payable under section 726(b) of the Bankruptcy Code in an aggregate amount not to exceed \$100,000 (without regard to the Carve-Out Trigger Notice), and (iii) to the extent allowed at any time and only as specifically permitted under the Budget, all unpaid fees and expenses of the professionals retained by the Debtors and any Committee appointed in the Bankruptcy Cases, (A) that are incurred on or prior to the third business day succeeding the date of delivery of the Carve-Out Trigger Notice, or (B) subject to an aggregate cap of \$100,000 for the Debtors' professionals and a separate aggregate cap of \$25,000 for the Committee's professionals that are incurred after the third business day succeeding the date of delivery of a Carve-Out Trigger Notice (the caps set forth in this clause (B), the "Post Carve-Out Trigger Notice (Cap"), (iv) (iv) to the extent allowed by order of the Court, the fees and expenses payable to Hilco Corporate Finance in connection with the consummation of a sale of substantially all of the Debtors' assets; and (v) (v) to the extent allowed by order of the Court, the completion fee payable by the Debtors to Alvarez & 		
	Marsal. See Interim Order ¶ 24		

Summary of Relevant Provisions		
Material Terms	DIP Facility	
Milestones Bankruptcy Rule 4001(c)(1)(B)(v) and (vi)	he Debtors shall be required to comply with the following milestones he "Milestones"), which may only be modified by the written consent of the DIP Lender in its sole discretion: (a) No later than five (5) days after the Petition Date, the Court shall have entered the Interim DIP Order in form and substance satisfactory to the DIP Lender; (b) No later than 30 days after the Petition Date, the Court shall have entered the Final DIP Order in form and substance satisfactory to the DIP Lender; (c) No later than two (2) days after the Petition Date, the Debtors shall have filed a motion (the "Sale and Bidding Procedures Motion") to sell, and to approve bidding procedures for the sale of substantially all of the Debtors' assets (the "Assets"); (d) No later than 30 days after the Petition Date, the Court shall have entered an order in form and substance acceptable to the DIP Lender, approving the bidding procedures set forth in the Sale and Bidding Procedures Motion; (e) No later than 66 days after the Petition Date, the Court shall have entered an order (the "Sale Order") in form and substance acceptable to the DIP Lender, approving the sale of all or substantially all of the Assets; and (f) No later than 15 days after the entry of the Sale Order, the Debtors shall have consummated the Sale Transaction for all of Debtors' assets to the DIP Lender in a manner consistent with the Sale Order.	
	See DIP Credit Agreement ¶ 5.18(c) (Bankruptcy Related Affirmative Covenants; Milestones); Interim Order ¶ 17	
506(c) Waiver Bankruptcy Rule 4001(c)(1)(B)(x)	Subject to and upon entry of the Final DIP Order, except to the extent of the Carve-Out, no costs or expenses of administration that have been or may be incurred in the Chapter 11 Cases at any time shall be charged against the DIP Lender, the DIP Collateral, the Prepetition Bridge Lender, or the Prepetition Collateral pursuant to sections 105 or 506(c) of the Bankruptcy Code, or otherwise, without the prior written consent of the DIP Lender or the Prepetition Bridge Lender, as may be applicable, and no such consent shall be implied from any action, inaction, or acquiescence by any party. See Interim Order ¶ 30	

Summary of Relevant Provisions		
Material Terms	DIP Facility	
Section 552(b) Waiver Bankruptcy Rule 4001(c)(1)(B)(x) Liens on Avoidance Actions Bankruptcy Rule 4001(c)(1)(B)(xi)	N/A The DIP Collateral shall not consist of avoidance actions brought pursuant to Chapter 5 of the Bankruptcy Code or applicable state law equivalents but they shall extend to the proceeds thereof (the "Avoidance Action Proceeds"). See Interim Order ¶ 5	
Waiver of Marshaling Bankruptcy Rule 4001(c)(1)(B)(x)	Subject to and upon entry of the Final DIP Order, in no event shall the DIP Lender be subject to the equitable doctrine of "marshaling" or any other similar doctrine with respect to any of the DIP Collateral or the Prepetition Collateral, as applicable. See Interim Order ¶ 31	
Indemnification Bankruptcy Rule 4001(c)(1)(B)(ix)	The Debtors are authorized to indemnify and hold the DIP Lender and each such entities' current and former affiliates, and each such entity's current and former directors, officers, managers and equityholders (regardless of whether such interests are held directly or indirectly), predecessors, successors and assigns, and direct and indirect subsidiaries, and each of such entity's current and former officers, members, managers, directors, equityholders (regardless of whether such interests are held directly or indirectly), principals, members, employees, agents, independent contractors, representatives, managed accounts or funds, management companies, fund advisors, investment advisors, financial advisors, and partners (including both general and limited partners) (each an "Indemnified Party") harmless from and against any and all claims, damages, losses, liabilities and expenses (including, without limitation, all fees and disbursements of attorneys and other professionals) to which any Indemnified Party may become liable or which may be incurred by or asserted against any Indemnified Party, in each case in connection with or arising out of or by reason of any investigation, litigation or proceeding arising out of or relating to or in connection with the DIP Facility, any obligation, or any act, event or transaction related or attendant thereto or any use or intended use of the proceeds of the DIP Facility, except to the extent the same is found in a final, non—appealable judgment by a court of competent jurisdiction to have resulted from such Indemnified Party's own gross negligence or willful misconduct.	

Summary of Relevant Provisions		
Material Terms	DIP Facility	
	See Interim Order ¶ 33; DIP Credit Agreement § 11.3 (Indemnification)	
Automatic Stay Waiver/Modification Bankruptcy Rule 4001(c)(1)(B)(iii)	The automatic stay imposed under section 362(a)(2) of the Bankruptcy Code is hereby modified as necessary to effectuate all of the terms and provisions of the Interim DIP Order, including, without limitation, to: (a) permit the Debtors to grant the DIP Liens, Adequate Protection Liens, and DIP Superpriority Claims; (b) permit the Debtors to perform such acts as the DIP Lender may reasonably request to assure the perfection and priority of the liens granted herein; (c) permit the Debtors to incur all liabilities and obligations to the DIP Lender; and (d) authorize the Debtors to pay, and the DIP Lender to retain and apply, payments made in accordance with the terms of the Interim DIP Order. See Interim Order ¶ 13	
Waiver/Modification of Applicability of Nonbankruptcy Law Relating to Perfection or Enforceability of Liens Bankruptcy Rule 4001(c)(1)(B)(vii)	N/A	

Provisions to Be Highlighted

- 5. The Debtors hereby disclose the below provisions (collectively, the "<u>Highlighted Provisions</u>") pursuant to Complex Case Procedures (G)(1)(a):
 - Cross-Collateralization Provisions: No such provision is contained in the Interim Order. See Complex Case Procedures (G)(1)(a)(i).
 - Findings of Fact: Pursuant to paragraph F of the Interim Order, the Debtors have provided the Debtors' Stipulations concerning the Prepetition Secured Note, the Prepetition Secured Note Obligations, and the Prepetition Secured Note Liens, and a release and indemnification of the Prepetition Bridge Lender, among other things. See Complex Case Procedures (G)(1)(a)(ii).

- **506(c)** Waiver: Pursuant to paragraph 30 of the Interim Order, the Debtors are seeking approval of a waiver of rights under Bankruptcy Code section 506(c) against the DIP Collateral and, upon entry of the Final Order, the Prepetition Collateral. Accordingly, parties will have notice of (and an opportunity to object to) the Debtors' proposed waiver. *See* Complex Case Procedures (G)(1)(a)(iii).
- Liens on Avoidance Actions: Pursuant to paragraph 5 of the Interim Order, the Debtors are seeking approval to grant liens on the proceeds of Avoidance Actions. Accordingly, parties will have notice of (and an opportunity to object to) the liens. See Complex Case Procedures (G)(1)(a)(iv).
- Provisions Authorizing "Roll Up" or Repayment of Prepetition Debt: Paragraph 3 of the Interim Order authorizes the Debtors to refinance in full the Prepetition Secured Note using proceeds of the initial borrowing under the DIP Facility upon entry of the Interim Order and the satisfaction or waiver of all other closing conditions in the DIP Credit Agreement, without any further action by the Debtors or any other party. See Complex Case Procedures (G)(1)(a)(v).
- **Disparate Treatment of Debtor and Creditors' Committee Professionals in Carve-Out**: Pursuant to paragraph 24(a) of the Interim Order, professional fees are subject to an aggregate cap of \$100,000 for the Debtors' professionals and a separate aggregate cap of \$25,000 for the Committee's professionals, that are incurred after the third business day succeeding the date of delivery of a Carve-Out Trigger Notice. *See* Complex Case Procedures (G)(1)(a)(vi).
- Non-Consensual Priming of Prepetition: While the DIP Facility provides for "priming" liens, as discussed below, the Debtors anticipate that they will obtain the consent of CrossFirst, the only known holder of liens subject to the priming liens. *See* Complex Case Procedures (G)(1)(a)(vii).

The Debtors' Prepetition Capital Structure

6. As of the Petition Date, the Debtors are party to two outstanding prepetition secured credit facilities. In addition, one Debtor, OTB Acquisition LLC, has granted an all-asset security interest to its primary food vendor, US Foods. The Debtors' outstanding prepetition obligations in connection with these secured claims are in the amount of approximately \$19,602,000 in the aggregate:

Facility	Approximate Principal Outstanding
CrossFirst Credit Facility	\$11,750,000
Prepetition Bridge Financing	\$4,032,000
US Foods	\$3,820,000
Total	\$19,602,000

a. CrossFirst Credit Agreement

- 7. On September 9, 2022, debtor OTB Acquisition LLC, CrossFirst Bank, as Lender, and each of the other co-Debtors, entered into that certain Credit Agreement (as amended or otherwise modified from time to time, the "Prepetition Credit Agreement") which provides for a revolving loan facility with an aggregate commitment of \$15,000,000, including a \$3,000,000 sublimit for letters of credit. The Debtors used the proceeds advanced under this facility to, among other things, fund (i) working capital and other general corporate expenses, (ii) the issuance of letters of credit and (iii) the acquisition and development of restaurants. The outstanding obligations under the Prepetition Credit Agreement are secured by a lien (the "CrossFirst Liens") on substantially all of the Debtors' assets (the "CrossFirst Collateral").
- 8. Additionally, following certain defaults under the Prepetition Credit Agreement, including the Debtors' failure to make required payments, the Debtors negotiated a forbearance (the "Forbearance Agreement") with CrossFirst. The Debtors and CrossFirst entered into that certain Forbearance Agreement, dated as of February 14, 2025, pursuant to which CrossFirst agreed to temporarily forbear from exercising their remedies under the Prepetition Credit Agreement until March 5, 2025.

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 19 of 171

9. As of the Petition Date, CrossFirst is owed approximately \$11,750,000 (inclusive of fees and expenses) under the Prepetition Credit Agreement.

b. Prepetition Bridge Loans

- 10. In January 2025, the Debtors required additional working capital to (i) continue operations and (ii) evaluate and pursue certain strategic alternatives in order to maximize the value of the Debtors' assets for the benefit of all stakeholders. The Debtors were in default under the Prepetition Credit Agreement, and CrossFirst was unable to advance additional funds. However, CrossFirst was willing to subordinate its claims and liens to new "rescue" financing and potential debtor-in-possession financing (subject to a cap). As a result, the Debtors and their professionals commenced a process to identify potential bridge funding to provide the Debtors with sufficient liquidity to continue operations while they evaluate and pursue certain strategic alternatives, including a potential chapter 11 filing, to maximize the value of the Debtors' assets for all stakeholders.
- 11. Following negotiations with two parties, on February 14, 2025, OTB Lender, LLC, a Delaware corporation (the "Prepetition Bridge Lender"), extended a term loan in an aggregate principal amount of \$4,000,000 (the "Prepetition Bridge Loan Amount") to the Debtors pursuant to that certain Secured Promissory Note, dated as of February 14, 2025 (as amended or otherwise modified prior to the Petition Date, the "Prepetition Secured Note" and the loans thereunder, the "Prepetition Bridge Loans"), plus fees, expenses (including advisors' and professionals' fees and expenses, in each case, that are chargeable or reimbursable under the applicable agreements), disbursements, charges, claims, indemnities and other costs and obligations of whatever nature incurred in connection therewith which are chargeable or otherwise reimbursable under the

applicable agreements or applicable law (collectively, the "Prepetition Secured Note Obligations"). Concurrently with entering into the Prepetition Secured Note, the Prepetition Bridge Lender, entered into that certain Subordination and Intercreditor Agreement, dated as of February 14, 2025 (as amended or otherwise modified, the "Subordination Agreement") pursuant to which CrossFirst consented to the priming of the CrossFirst Collateral and the subordination of the obligations arising under the Prepetition Credit Agreement. The Senior Secured Note matures on April 5, 2025 and is secured by a first-priority lien (the "Prepetition Secured Note Liens") on all assets of the Debtors (the "Prepetition Collateral").

- 12. The proceeds of the Prepetition Bridge Loans were used to (i) fund operations while assessing strategic alternatives to maximize the value of the Debtors' assets and (ii) repay an overdraft balance in the Debtors' cash management system. The Prepetition Bridge Loans provided the company with the liquidity it needed to salvage operations, maintain over 2,800 jobs for the Debtors' employees and, ultimately, continue a fulsome marketing and sale process for the benefit of all stakeholders.
- 13. As of the Petition Date, the Prepetition Bridge Lender is owed approximately \$4,032,000 (inclusive of fees and expenses) under the Senior Secured Note.

c. Miscellaneous Secured Claims

14. U.S. Foods, the Debtors' largest food vendor, has a claim that is secured by substantially all of the assets of OTB Acquisition LLC ("<u>U.S. Foods Secured Claim</u>"). The Debtors' estimate that the outstanding amount of the U.S. Secured Claim is approximately \$3,820,000 as of the Petition Date. Pursuant to various motions filed on or about the Petition Date

(including a "Critical Vendor" motion), the Debtors are seeking authority to satisfy the U.S. Foods Secured Claim in full.

The Negotiation of the Proposed DIP Facility

- Debtors commenced these Chapter 11 Cases to implement a strategic asset sales strategy (the "Sale Transaction") under section 363 of the Bankruptcy Code. First Day Decl. ¶ 37. Pursuant to the Milestones, the Debtors seek entry of an order approving such Sale Transaction no later than 66 days after the Petition Date, and to close the Sale Transaction shortly thereafter. *Id.* at ¶ 39. However, the Debtors' businesses will not generate sufficient levels of operating cash flow in the ordinary course of business to cover the projected costs of the Chapter 11 Cases absent the DIP Facility. Tibus Decl. ¶ 8.
- 16. Following extensive, good faith, arm's-length negotiations with the Debtors and other parties-in-interest, the DIP Lender agreed to provide the Debtors with the DIP Facility. *Id.* at ¶ 12. The Debtors also anticipate that the DIP Lender will serve as the stalking horse bidder (in such capacity, the "Stalking Horse Bidder") and credit bid the full amount of its claim pursuant to the Sale Transaction.
- 17. Notably, the DIP Lender (in its capacity as the Prepetition Bridge Lender) has consented to the priming of its Prepetition Secured Note Liens. The Debtors anticipate that CrossFirst Bank will consent to the priming of the CrossFirst Liens. And the DIP Lender has agreed that its DIP Liens shall not prime any valid, perfected, and unavoidable liens of U.S. Foods.
- 18. The Debtors have determined that the DIP financing proposal from the DIP Lender is the most attractive and executable financing proposal for these Chapter 11 Cases. *Id.* at ¶ 15.

19. For these reasons, and for the reasons set forth below, in the Tibus Declaration, and in the First Day Declaration, the Debtors believe that entering into the DIP Facility is the best option available to the Debtors and will maximize the value of the Debtors' estates by allowing the Debtors to operate in the ordinary course while pursuing the Sale Transaction during the pendency of the Chapter 11 Cases, and therefore is a sound exercise of the Debtors' business judgment. Accordingly, the Debtors respectfully request that the Court enter the Proposed Orders.

BASIS FOR RELIEF

- I. The Debtors Should Be Authorized To Obtain Post-Petition Financing on a Senior Secured and Superpriority Basis
- 20. The Debtors meet the requirements for relief under section 364 of the Bankruptcy Code, which permit a debtor to obtain post-petition financing and, in return, to grant superpriority administrative status and liens on its property. Specifically, section 364(c) of the Bankruptcy Code provides as follows:

If the trustee is unable to obtain unsecured credit allowable under section 503(b)(1) of this title as an administrative expense, the court, after notice and a hearing, may authorize the obtaining of credit or the incurring of debt:

- (1) with priority over any or all administrative expenses of the kind specified in section 503(b) or 507(b) of this title;
- (2) secured by a lien on property of the estate that is not otherwise subject to a lien; or
- (3) secured by a junior lien on property of the estate that is subject to a lien.

11 U.S.C. § 364(c).

21. Further, section 364(d) of the Bankruptcy Code provides:

- (1) The court, after notice and a hearing, may authorize the obtaining of credit or the incurring of debt secured by a senior or equal lien on property of the estate that is subject to a lien only if:
 - (A) the trustee is unable to obtain such credit otherwise; and
 - (B) there is adequate protection of the interest of the holder of the lien on the property of the estate on which such senior or equal lien is proposed to be granted.
- (2) In any hearing under this subsection, the trustee has the burden of proof on the issue of adequate protection.

11 U.S.C. § 364(d).

22. Provided that an agreement to obtain secured credit is consistent with the provisions of, and policies underlying, the Bankruptcy Code, courts grant a debtor considerable deference in exercising its sound business judgment in obtaining such credit. See, e.g., In re L.A. Dodgers LLC, 457 B.R. 308, 313 (Bankr. D. Del. 2011) ("[C]ourts will almost always defer to the business judgment of a debtor in the selection of the lender."); In re Barbara K. Enters., Inc., Case No. 08-11474, 2008 WL 2439649, at *14 (Bankr. S.D.N.Y. June 16, 2008) (explaining that courts defer to a debtor's business judgment "so long as a request for financing does not 'leverage the bankruptcy process' and unfairly cede control of the reorganization to one party in interest"); In re Ames Dep't Stores, Inc., 115 B.R. 34, 40 (Bankr. S.D.N.Y. 1990) ("[C]ases consistently reflect that the court's discretion under section 364 [of the Bankruptcy Code] is to be utilized on grounds that permit [a debtor's reasonable business judgment to be exercised so long as the financing agreement does not contain terms that leverage the bankruptcy process and powers or its purpose is not so much to benefit the estate as it is to benefit a party-in-interest."); In re Farmland Indus., Inc., 294 B.R. 855, 881 (Bankr. W.D. Mo. 2003) (noting that approval of post-petition financing requires, inter alia, an exercise of "sound and reasonable business judgment").

23. In determining whether the Debtors have exercised sound business judgment in deciding to enter into the DIP Loan Documents, the Court may appropriately take into consideration non-economic benefits to the Debtors offered by a proposed post-petition facility. For example, in *In re ION Media Networks, Inc.*, the Bankruptcy Court for the Southern District of New York held that:

Although all parties, including the Debtors and the Committee, are naturally motivated to obtain financing on the best possible terms, a business decision to obtain credit from a particular lender is almost never based purely on economic terms. Relevant features of the financing must be evaluated, including noneconomic elements such as the timing and certainty of closing, the impact on creditor constituencies and the likelihood of a successful reorganization. This is particularly true in a bankruptcy setting where cooperation and established allegiances with creditor groups can be a vital part of building support for a restructuring that ultimately may lead to a confirmable reorganization plan. That which helps foster consensus may be preferable to a notionally better transaction that carries the risk of promoting unwanted conflict.

Case No. 09-13125, 2009 WL 2902568, at *4 (Bankr. S.D.N.Y. July 6, 2009).

24. Here, given all the facts and circumstances present in the Chapter 11 Cases, the Debtors have amply satisfied the necessary conditions under sections 364(c) and (d) of the Bankruptcy Code for authority to enter into the DIP Facility. The Debtors exercised proper business judgment in securing the DIP Facility on terms that are fair and reasonable and the best available to them under the circumstances and in the current market. Moreover, given the circumstances described above, the Debtors could not obtain credit on an unsecured or administrative expense basis or without the limited priming contemplated by the Interim Order. For all the reasons discussed further below, the Debtors respectfully submit that the Court should

grant the Debtors' request to enter into the DIP Financing pursuant to sections 364(c) and (d) of the Bankruptcy Code.

A. The Debtors Exercised Sound and Reasonable Business Judgment in Deciding to Enter into the DIP Financing

- 25. Based on the facts of the Chapter 11 Cases, the DIP Financing represents a proper exercise of the Debtors' business judgment. As noted above, bankruptcy courts routinely defer to a debtor's business judgment on most business decisions, including decisions about whether and how to borrow money. See, e.g., In re L.A. Dodgers LLC, 457 B.R. at 313; In re Trans World Airlines, Inc., 163 B.R. 964, 974 (Bankr. D. Del. 1994) (approving post-petition loan and receivables facility because such facility "reflect[ed] sound and prudent business judgment"); In re Ames Dep't Stores, Inc., 115 B.R. at 40; In re Curlew Valley Assocs., 14 B.R. 506, 511-13 (Bankr. D. Utah 1981); Grp. of Institutional Investors v. Chi., Milwaukee, St. Paul & Pac. R.R., 318 U.S. 523, 550 (1943); In re Farmland Indus., Inc., 294 B.R. at 882 ("Business judgments should be left to the board room and not to this Court.") (quoting *In re Simasko Prod. Co.*, 47 B.R. 444, 449 (Bankr. D. Colo. 1985)); In re Lifeguard Indus., Inc., 37 B.R. 3, 17 (Bankr. S.D. Ohio 1983). "More exacting scrutiny would slow the administration of the debtor's estate and increase its cost, interfere with the Bankruptcy Code's provision for private control of administration of the estate, and threaten the court's ability to control a case impartially." Richmond Leasing Co. v. Capital Bank N.A., 762 F.2d 1303, 1311 (5th Cir. 1985).
- 26. Specifically, to determine whether a debtor has met this business judgment standard, a court need only "examine whether a reasonable businessperson would make a similar decision under similar circumstances." *In re Dura Auto. Sys., Inc.*, No. 06-11202, 2007 WL 7728109, at *97 (Bankr. D. Del. Aug. 15, 2007) (quoting *In re Exide Techs.*, 340 B.R. 222, 239)

(Bankr. D. Del. 2006)); see also In re Stanziale v. Nachtomi (In re Tower Air, Inc.), 416 F.3d 229, 239 (3d Cir. 2005); In re AbitibiBowater, 418 B.R. 815, 831 (Bankr. D. Del. 2009) (the business judgment standard is "not a difficult standard to satisfy"). Under the business judgment rule, "management of a corporation's affairs is placed in the hands of its board of directors and officers, and the Court should interfere with their decisions only if it is made clear that those decisions are, inter alia, clearly erroneous, made arbitrarily, are in breach of the officers' and directors' fiduciary duty to the corporation, are made on the basis of inadequate information or study, are made in bad faith, or are in violation of the Bankruptcy Code." In re Farmland Indus., Inc., 294 B.R. at 881 (citing In re United Artists Theatre Co., 315 F.3d 217, 233 (3d Cir. 2003), Richmond Leasing Co., 762 F.2d at 1309, and *In re Defender Drug Stores, Inc.*, 145 B.R. 312, 317 (B.A.P. 9th Cir. 1992)); see also In re Food Barn Stores, Inc., 107 F.3d 558, 567 n. 16 (8th Cir. 1997) ("Where the [debtor's] request is not manifestly unreasonable or made in bad faith, the court should normally grant approval as long as the proposed action appears to enhance the debtor's estate" (citing Richmond Leasing Co., 762 F.2d at 1309)); In re Farmland Indus. Inc., 294 B.R. at 913 (approving the rejection of employment agreements and noting that "[u]nder the business judgment standard, the question is whether the [proposed action] is in the Debtors' best economic interests, based on the Debtors' best business judgment in those circumstances." (citations omitted)).

27. Courts defer to a debtor's decision to borrow money unless such decision is arbitrary or capricious. *See In re Trans World Airlines, Inc.*, 163 B.R. at 974. Furthermore, in considering whether the terms of postpetition financing are fair and reasonable, courts consider the terms in light of the relative circumstances of both the debtor and the potential lender. *See In re Farmland Indus., Inc.*, 294 B.R. at 886; *see also Unsecured Creditors' Comm. Mobil Oil Corp.*

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 27 of 171

v. First Nat'l Bank & Trust Co. (In re Ellingsen McLean Oil Co., Inc.), 65 B.R. 358, 365 n.7 (W.D. Mich. 1986) (recognizing a debtor may have to enter into "hard bargains" to acquire funds for its reorganization).

28. The Debtors' management team negotiated the terms and conditions of the DIP Loan Documents with the DIP Lender in good faith, at arms' length, and with the active involvement and assistance of their experienced legal and financial advisors. Tibus Decl. at ¶ 15. The Debtors believe that they have obtained the best financing available to them under the circumstances. Moreover, access to the DIP Financing is necessary to allow the Debtors to administer these Chapter 11 Cases and pursue the Sale Transaction. See Tibus Decl. ¶ 16. Consequentially, the Debtors' determination to move forward with the DIP Facility is a sound exercise of their business judgment following a careful evaluation of alternatives. Indeed, the Debtors believe that the terms of the DIP Facility are favorable in the context of the market for financings of this type, as the DIP Lender has offered attractive financing terms overall. *Id.* The pricing, fees and covenants provided for in the proposed DIP Facility are favorable, particularly for a severely distressed borrower with an urgent need for liquidity. *Id.* at ¶ 18. The Milestones in the DIP Facility are tailored to comport with the timeline of the Debtors' sale of substantially all of their assets. Further, entry into postpetition financing arrangements with one of the Debtors' prepetition secured lenders will allow the Debtors to avoid a costly, distracting and destructive adequate protection fight at the outset of these Chapter 11 Cases that would jeopardize the Debtors' ability to consummate the Sale Transaction and would be extremely difficult to win. Accordingly, the Court should authorize the Debtors' entry into the DIP Loan Documents and the borrowings contemplated therein as it is a reasonable exercise of the Debtors' business judgment.

- B. The Debtors Meet the Conditions Necessary Under Section 364(c) and (d) to Obtain Postpetition Financing on a Senior Secured and Superpriority Basis
- 29. The Debtors propose to obtain financing under the DIP Facility by providing superpriority claims and liens pursuant to sections 364(c) and 364(d)(1) of the Bankruptcy Code. The Debtors propose to provide the DIP Lender with liens on and security interests in the DIP Collateral, including the Prepetition Collateral and the CrossFirst Collateral (whether existing on or arising after the Petition Date), and any other assets of the Debtors that were not subject to any validly perfected liens or security interest as of the Petition Date. The DIP Liens shall not extend to, and the DIP Collateral shall not consist of, avoidance actions brought pursuant to Chapter 5 of the Bankruptcy Code or applicable state law equivalents but they shall extend to proceeds thereof.
- 30. In evaluating proposed postpetition financing under sections 364(c) and 364(d)(1) of the Bankruptcy Code, courts perform a qualitative analysis and consider whether (a) unencumbered credit or alternative financing without superpriority status is available to the debtor, (b) the credit transactions are necessary to preserve assets of the estate, and (c) the terms of the credit agreement are fair, reasonable, and adequate. *See, e.g., In re Los Angeles Dodgers LLC*, 457 B.R. at 312; *In re Aqua Assoc.*, 123 B.R. 192, 195-99 (Bankr. E.D. Pa. 1991); *In re St. Mary Hosp.*, 86 B.R. 393, 401-02 (Bankr. E.D. Pa. 1988); *In re Crouse Group, Inc.*, 71 B.R. 544, 549-551 (Bankr. E.D. Pa. 1987); *see also Bland v. Farmworker Creditors*, 308 B.R. 109, 113-14 (S.D. Ga. 2003); *In re Ames Dep't Stores*, 115 B.R. at 37-40.
 - (i) The Debtors Are Unable To Obtain Financing on More Favorable Terms Than the DIP Facility
- 31. In order to satisfy this test, a debtor need only demonstrate "by a good faith effort that credit was not available without" the protections afforded to potential lenders by sections

364(c) or 364(d) of the Bankruptcy Code. *In re Snowshoe Co., Inc.*, 789 F.2d 1085, 1088 (4th Cir. 1986) ("The statute imposes no duty to seek credit from every possible lender before concluding that such credit is unavailable"); *In re Pearl-Phil GMT (Far East) Ltd. v. Caldor Corp.*, 266 B.R. 575, 584-85 (S.D.N.Y. 2001) (superpriority administrative expenses authorized where debtor could not obtain credit as an administrative expense); *In re Ames Dep't Stores, Inc.*, 115 B.R. at 40 (approving financing facility and holding that debtor made reasonable efforts to satisfy the standards of section 364(c) to obtain less onerous terms where debtor approached four lending institutions, was rejected by two and selected the least onerous financing option from the remaining two lenders); *see also In re Plabell Rubber Prods., Inc.*, 137 B.R. 897, 900 (Bankr. N.D. Ohio 1992). This is especially true where time is of the essence. *See In re Reading Tube Indus.*, 72 B.R. 329, 332 (Bankr. E.D. Pa. 1987).

32. Moreover, in circumstances where only a few lenders likely can or will extend the necessary credit to a debtor, "it would be unrealistic and unnecessary to require [the debtor] to conduct such an exhaustive search for financing." *In re Sky Valley, Inc.*, 100 B.R. 107, 113 (Bankr. N.D. Ga. 1988), *aff'd sub nom. Anchor Sav. Bank FSB v. Sky Valley, Inc.*, 99 B.R. 117, 120 n.4 (N.D. Ga. 1989); *see also In re Snowshoe Co.*, 789 F.2d at 1088 (demonstrating that credit was unavailable absent the senior lien by establishment of unsuccessful contact with other financial institutions in the geographic area); *In re Stanley Hotel, Inc.*, 15 B.R. 660, 663 (D. Colo. 1981) (bankruptcy court's finding that two national banks refused to grant unsecured loans was sufficient to support conclusion that the requirements of section 364 of the Bankruptcy Code were met); *In re Ames Dep't Stores*, 115 B.R. at 37-39 (debtor must show that it made reasonable efforts to seek other sources of financing under section 364(a) and (b) of the Bankruptcy Code).

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 30 of 171

- 33. As noted above and discussed in greater detail in the Declarations, the Debtors and the DIP Lender negotiated the principal terms of the DIP Facility in connection with broader negotiations regarding the Chapter 11 Cases and the Sale Transaction. *See generally* Tibus Decl. The Debtors, together with their advisors, also engaged with other potential sources of postpetition financing to determine whether the Debtors could obtain debtor-in-possession financing on better terms. *Id.* at ¶¶ 10, 16. However, no other third-party lender was willing to provide financing on better terms. *Id.*
- 34. Notwithstanding the Debtors' lack of alternative financing options, the terms of the DIP Facility are the product of extensive and good faith negotiations between the Debtors, and the DIP Lender, among others, each of whom was represented by experienced counsel and financial advisors. Id. at ¶ 15. Through these vigorous negotiations, the Debtors were able to secure the most favorable terms possible for the DIP Facility. Id. The Debtors were unable to obtain financing on equal or better terms from the DIP Lender, or any other source, without granting the liens securing the DIP Loans and providing the DIP Lender with superpriority claims. Simply put, the DIP Facility provides the Debtors with the liquidity they need, at the lowest cost available. Id. at ¶ 19.
 - (ii) The DIP Facility Is Necessary to Preserve the Value of the Debtors' Estates
- 35. The Debtors, as debtors in possession, have a fiduciary duty to protect and maximize their estates' assets. *See In re Mushroom Transp. Co.*, 382 F.3d 325, 339 (3d Cir. 2004). The Debtors immediate access to postpetition financing is essential to the Debtors' ability to effectively carry out that duty. Without the proceeds of the DIP Facility, the Debtors would be

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 31 of 171

unable to fund critical payments, including obligations such as employee payroll and vendor payments in the ordinary course of business that are essential to the Debtors' operational viability.

- (iii) The Terms of the DIP Facility Are Fair, Reasonable, and Adequate under the Circumstances
- 36. In considering whether the terms of postpetition financing are fair and reasonable, courts consider the terms in light of the relative circumstances of both the debtor and the potential lender. See In re Farmland Indus., Inc., 294 B.R. at 886; see also In re Ellingsen MacLean Oil Co., 65 B.R. at 365 (a debtor may have to enter into hard bargains to acquire funds). The appropriateness of a proposed financing facility should also be considered in light of current market conditions. See In re Lyondell Chem. Co., No. 09-10023 (REG) (Bankr. S.D.N.Y. Feb. 27, 2009), Hr'g Tr. 740:4-6 ("[B]y reason of present market conditions, as disappointing as the [DIP] pricing terms are, I find the provisions [of the DIP] reasonable here and now.").
- 37. Here, the terms of the DIP Facility are fair, appropriate, reasonable, adequate under the circumstances, and in the best interests of the Debtors, their estates, and their creditors. Moreover, the DIP Lender agreed to provide the DIP Facility on favorable terms as part of an overall agreement to support the Debtors' Chapter 11 Cases and the Sale Transaction. *Id.* at ¶ 15.
- 38. All of the terms of the DIP Facility were strenuously negotiated by the Debtors and their advisors to ensure that the terms were as favorable to the Debtors as possible under the circumstances, and the Debtors believe that such terms represent fair consideration for the critical financing being provided by the DIP Lender. *See id.* at ¶ 18. In particular, the fees and rates provided for under the DIP Facility and described above are reasonable and within market practice for debtor-in-possession financings of this size and type, and therefore the Court should authorize the Debtors to pay the fees, interest, and other amounts provided under and in accordance with the

terms of the DIP Loan Documents. *Id.* As is also customary, the DIP Facility contains certain case controls (*e.g.*, milestones) and other bankruptcy-related terms. The Debtors believe that these terms provide sufficient flexibility for the Debtors to maximize the value of their estates. *Id.* at 19.

C. The Adequate Protection Is Appropriate

- 39. Although the Bankruptcy Code does not define "adequate protection," section 361 of the Bankruptcy Code delineates a non-exhaustive list of the available types of adequate protection, which include periodic cash payments, additional liens, replacement liens and the "indubitable equivalent of such entity's interest in such property." 11 U.S.C. § 361. The focus of the requirement is to protect a secured creditor from the diminution in the value of its interest in the particular collateral during the period of use. *See In re Swedeland Dev. Group, Inc.*, 16 F.3d 552, 564 (3d Cir. 1994) ("The whole purpose of adequate protection for a creditor is to insure that the creditor receives the value for which he bargained prebankruptcy.") (internal citations omitted). When priming of liens is sought under section 364(d), courts also examine whether the prepetition secured creditors are being provided adequate protection for the value of their liens. *See In re Utah* 7000, *LLC*, No. 08-21869, 2008 WL 2654919, at *3 (Bankr. D. Utah July 3, 2008); *In re Beker Indus. Corp.*, 58 B.R. 725, 737 (Bankr. S.D.N.Y. 1986).
- 40. CrossFirst is entitled to adequate protection of its interests in the applicable CrossFirst Collateral, including Cash Collateral, solely to the extent of any Diminution in Value of its interests in the CrossFirst Collateral, from and after the Petition Date ("Adequate Protection"). Such Adequate Protection shall include a valid, perfected replacement security interest in and lien on the CrossFirst Collateral (the "Adequate Protection Liens"), which security interests and liens

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 33 of 171

shall be subject to and subordinate in all respects to the Carve-Out. These terms are fair and customary for debtor-in-possession financings of this type.

D. The Refinancing of the Prepetition Secured Note Is Appropriate

- 41. The proposed refinancing of the Prepetition Secured Note is also justified under the circumstances. *First*, the DIP Lender specifically negotiated for the refinancing of the Prepetition Secured Note in the context of their commitment to provide the DIP Facility on the terms described herein and in the Interim Order. Tibus Decl. ¶ 17. The Debtors, on the one hand, and the DIP Lender, on the other, engaged in arm's-length negotiations and ultimately agreed to the refinancing of the Prepetition Secured Note as consideration for, among other things, the DIP Lender consenting to make millions of dollars available to the Debtors to fund these Chapter 11 Cases. *Id.* Indeed, such refinancing is a condition of the DIP, without which the DIP Lender would not have agreed to provide the DIP Facility or to allow the Debtors to use Cash Collateral. *Id. Second*, the refinancing of the Prepetition Secured Note will not prejudice the Debtors' stakeholders. *Third*, the refinancing of the Prepetition Secured Note through the DIP Facility is effectively a continuation of the prepetition status quo and will not layer any additional debt onto the top of the Debtors' capital structure. Further, it will not reshuffle the relative priorities of creditors in any way. *Id.*
- 42. Finally, roll-ups and repayments of prepetition debt are a common feature of debtor-in-possession financings and have been approved in a variety of cases, including pursuant to interim financing orders, both in this and other districts. *See In re Franchise Group*, Case No. 24-12480 (JTD) (Bankr. D. Del. Nov. 7, 2024) (authorizing debtor-in-possession financing that included roll-up under the interim order); *In re American Tire Distributors, Inc.*,

Case No. 24-12391 (CTG) (Bankr. D. Del. Oct. 25, 2024) (same); *Red Lobster Management, LLC*, Case No. 24-02486 (GER) (Bankr. M.D. Fl. May 21, 2024) (same); *In re Sientra, Inc.*, Case No. 24-10245 (JTD) (Bankr. D. Del. Feb. 14, 2024) (same); *In re NanoString Technologies, Inc.*, Case No. 24-10160 (CTG) (Bankr. D. Del. Feb. 6, 2024) (same); *In re Virgin Orbit Holdings, Inc.*, Case No. 23-10405 (KBO) (Bankr. D. Del. Apr. 5, 2023); *In re GNC Holdings, Inc.*, Case No. 20-11662 (KBO) (Bankr. D. Del. June 26, 2020) (same); *In re Jack Cooper Ventures, Inc.*, Case No. 19-62393 (PWB) (Bankr. N.D. Ga. Aug. 8, 2019) (same); *In re Scovill Fasteners Inc.*, Case No. 11-21650 (JRS) (Bankr. N.D. Ga. Apr. 25, 2011) (same); *In re Allied Holdings, Inc.*, Case No. 05-12515 (CRM) (Bankr. N.D. Ga. Aug. 1, 2005).

II. The Debtors Should Be Authorized to Use Cash Collateral

- 43. Section 363(c)(2)(A) of the Bankruptcy Code permits a debtor in possession to use Cash Collateral with the consent of the secured party. Section 363(e) of the Bankruptcy Code provides for adequate protection of interests in property when a debtor uses Cash Collateral. Further, section 362(d)(1) of the Bankruptcy Code provides for adequate protection of interests in property due to the imposition of the automatic stay. *See In re Cont'l Airlines*, 91 F.3d 553, 556 (3d Cir. 1996) (en banc). While section 361 of the Bankruptcy Code provides examples of forms of adequate protection, such as granting replacement liens and administrative claims, courts decide what constitutes sufficient adequate protection on a case-by-case basis. *See, e.g., In re Swedeland Dev. Grp., Inc.*, 16 F.3d 552, 564 (3d Cir. 1994) (explaining that the "determination of whether there is adequate protection is made on a case by case basis").
- 44. Here, the DIP Lender and Crossfirst have consented to the Debtors' use of the Cash Collateral, subject to the terms and limitations set forth in the Interim Order. Further, as set forth

above, the proposed Adequate Protection to CrossFirst is appropriate, fair and customary for debtor-in-possession financings of this type.

III. The DIP Lender Should Be Afforded Good-Faith Protection Under Section 364(e) of the Bankruptcy Code

45. Section 364(e) of the Bankruptcy Code protects a good faith lender's right to collect on loans extended to a debtor, and its right in any lien securing those loans, even if the authority of the debtor to obtain such loans or grant such liens is later reversed or modified on appeal. Section 364(e) of the Bankruptcy Code provides as follows:

The reversal or modification on appeal of an authorization under this section [364 of the Bankruptcy Code] to obtain credit or incur debt, or of a grant under this section of a priority or a lien, does not affect the validity of any debt so incurred, or any priority or lien so granted, to an entity that extended such credit in good faith, whether or not such entity knew of the pendency of the appeal, unless such authorization and the incurring of such debt, or the granting of such priority or lien, were stayed pending appeal.

11 U.S.C. § 364(e).

46. The DIP Facility is the result of (a) the Debtors' reasonable and informed determination that the DIP Lender offered the most favorable terms on which to obtain vital postpetition financing and (b) extensive arm's-length, good-faith negotiations between the Debtors and the DIP Lender. *Id.* at ¶¶ 17, 18. The Debtors submit that the terms and conditions of the DIP Facility are reasonable and appropriate under the circumstances, and the proceeds of the DIP Facility will be used only for purposes that are permissible under the Bankruptcy Code. Further, no consideration is being provided to the DIP Lender, or any other party to the DIP Loan Documents other than as described herein. Accordingly, the Court should find that the obligations arising under the DIP Facility and other financial accommodations made to the Debtors have been

extended by the DIP Lender in "good faith" within the meaning of section 364(e) of the Bankruptcy Code and, therefore, the DIP Lender is entitled to all of the protections afforded thereby.

IV. The Automatic Stay Should Be Modified on a Limited Basis

- 47. The Interim Order provides that the automatic stay provisions of section 362 of the Bankruptcy Code will be modified to: (a) permit the Debtors to grant the DIP Liens, Adequate Protection Liens, and DIP Superpriority Claims; (b) permit the Debtors to perform such acts as the DIP Lender may reasonably request to assure the perfection and priority of the liens granted herein; (c) permit the Debtors to incur all liabilities and obligations to the DIP Lender; and (d) authorize the Debtors to pay, and the DIP Lender to retain and apply, payments made in accordance with the terms of the Interim DIP Order.
- 48. Stay modifications of this kind are ordinary features of debtor-in-possession financing arrangements, and, in the Debtors' business judgment, are reasonable and fair under the circumstances of the Chapter 11 Cases. *See, e.g., In re Eastern Outfitters, LLC*, Case No. 17-10243 (LSS) Bankr. D. Del. Feb. 8, 2017); *In re Basic Energy Servs., Inc.*, Case No. 16-12320 (KJC) (Bankr. D. Del. Oct. 26, 2016); *In re Verso Corp.*, Case No. 16-10163 (Bankr. D. Del. Jan. 27, 2016); *In re the Standard Register Co.*, No. 15-10541 (BLS). (Bankr. D. Del. Apr. 16, 2015) (modifying automatic stay as necessary to effectuate the terms of the order); *In re Cache, Inc.*, No. 15-10172 (MFW) (Bankr. D. Del. May 20, 2015) (same); *In re S. Reg'l Health Sys. Inc.*, No. 1564266 (WLH) (Bankr. N.D. Ga. Aug. 26, 2015) (modifying the automatic stay to permit exercise of remedies by DIP lender upon event of default); *In re TMP Directional Mktg., LLC*, No. 11-13835 (MFW) (Bankr. D. Del. Jan. 17, 2012) (modifying automatic stay as necessary to effectuate the terms of the order); *In re Cagle's, Inc.*, No. 11-80202 (JB) (Bankr. N.D. Ga. Oct. 21, 2011)

(modifying the automatic stay to permit exercise of remedies by DIP lender upon event of default); *In re Broadway 401 LLC*, No. 10-10070 (KJC) (Bankr. D. Del. Feb. 16, 2010) (modifying automatic stay as necessary to effectuate the terms of the order); *In re Haights Cross Commc'ns*, *Inc.*, No. 10-10062 (BLS) (Bankr. D. Del. Feb. 8, 2010) (same).

V. The Scope of the Carve-Out Is Appropriate

- 49. The Interim Order subjects the security interests and administrative expense claims of the DIP Lender to the Carve-Out. Such carve-outs for professional fees have been found to be reasonable and necessary to ensure that a debtor's estate and any statutory committee appointed can reimburse their professionals in certain circumstances following an event of default under the terms of the debtor's postpetition financing. See In re Ames Dep't Stores, Inc., 115 B.R. at 40. Neither the Interim Order nor the DIP Financing directly or indirectly deprives the Debtors' estates or other parties in interest of possible rights and powers by restricting the services for which professionals may be paid in these Chapter 11 Cases. See id. at 38 (observing that courts insist on carve-outs for professionals representing parties-in-interest because "[a]bsent such protection, the collective rights and expectations of all parties-in-interest are sorely prejudiced"). Additionally, the Carve-Out protects against administrative insolvency during the course of the Chapter 11 Cases by ensuring that assets remain for the payment of the U.S. Trustee's fees and professional fees of the Debtors and any statutory committee notwithstanding the grant of priming liens, superpriority claims, and adequate protection liens and claims.
- 50. Courts routinely approve carve-outs agreed to by the debtors and their DIP financing lenders. *See, e.g., In re LaVie Care Centers, LLC*, Case No. 24-55507 (PMB) (June 28, 2024) *In re Jack Cooper Ventures, Inc.*, Case No. 19-62393 (PWB) (Bankr. N.D. Ga. Sept. 13,

2019); In re Eastern Outfitters, LLC, Case No. 17-10243 (LSS) (Bankr. D. Del. Feb. 8, 2017); In re Modular Space Holdings, Inc., Case No. 1612825 (KJC) (Bankr. D. Del. Dec. 22, 2016); In re Basic Energy Servs., Inc., Case No. 16-12320 (KJC) (Bankr. D. Del. Oct. 26, 2016); In re Halcón Resources Corp., Case No. 16-11724 (BLS) (Bankr. D. Del. July 29, 2016); In re RCS Capital Corp., Case No. 16-10223 (MFW) (Bankr. D. Del. Feb. 3, 2016) In re Verso Corp., Case No. 16-10163 (KG) (Bankr. D. Del. Jan. 27, 2016); In re Swift Energy Co., Case No. 15-12670 (Bankr. D. Del. Jan. 5, 2016).

VI. Failure to Obtain Immediate Interim Access to the DIP Facility Would Cause Immediate and Irreparable Harm

- 51. Bankruptcy Rules 4001(b) and 4001(c) provide that a final hearing on a motion to obtain credit pursuant to section 364 of the Bankruptcy Code may not be commenced earlier than 14 days after the service of such motion. Upon request, however, the Court may conduct a preliminary, expedited hearing on the motion and authorize the obtaining of credit to the extent necessary to avoid immediate and irreparable harm to a debtor's estate.
- 52. The Debtors request that the Court hold and conduct a hearing to consider entry of the Interim Order authorizing the Debtors, from and after entry of the Interim Order until the Final Hearing, to withdraw and borrow funds under the DIP Facility to the extent permitted in the Interim Order. The Debtors require access to the DIP Facility prior to the Final Hearing and entry of the Final Order to continue operating in the ordinary course by paying their administrative expenses and implementing the relief requested in the Debtors' other "first day" motions. This and such other relief is necessary for the Debtors to avoid immediate and irreparable harm to the Debtors' estates and to preserve and maximize value for the benefit of all parties in interest.

Emergency Consideration

53. The Debtors respectfully request emergency consideration of this Motion pursuant to Bankruptcy Rule 6003, which empowers a court to grant relief within the first twenty-one days after the commencement of a chapter 11 case "to the extent that relief is necessary to avoid immediate and irreparable harm." Here, the Debtors believe an immediate and orderly transition into chapter 11 is critical to the viability of their operations and that any delay in granting the relief requested could hinder the Debtors' operations and cause irreparable harm. Furthermore, the failure to receive the requested relief during the first twenty-one days of these chapter 11 cases would severely disrupt the Debtors' operations at this critical juncture. Accordingly, the Debtors submit that they have satisfied the "immediate and irreparable harm" standard of Bankruptcy Rule 6003 and, therefore, respectfully request that the Court approve the relief requested in this Motion on an emergency basis.

Request for Final Hearing

54. Pursuant to Bankruptcy Rules 4001(b)(2) and 4001(c)(2), the Debtors request that the Court set a date for the Final Hearing that is as soon as practicable, and fix the time and date prior to the Final Hearing for parties to file objections to this Motion.

Waiver of Bankruptcy Rule 6004(a) and 6004(h)

55. To implement the foregoing successfully, the Debtors request that the Court enter an order providing that notice of the relief requested herein satisfies Bankruptcy Rule 6004(a) and that the Debtors have established cause to exclude such relief from the 14-day stay period under Bankruptcy Rule 6004(h).

Notice

Trustee for the Northern District of Georgia; (b) the Debtors' thirty (30) largest unsecured creditors; (c) counsel to the Debtors' prepetition lenders; (d) counsel to the Debtors' debtor-in-possession lender; (e) the Internal Revenue Service; (f) the Georgia Department of Revenue; (g) the Attorney General for the State of Georgia; (h) the United States Attorney for the Northern District of Georgia; (i) the state attorneys general for states in which the Debtors conduct business; and (j) any party that has requested notice pursuant to Bankruptcy Rule 2002. In light of the nature of the relief requested, the Debtors respectfully submit that no further notice is necessary.

No Prior Request

57. No prior request for the relief sought in the motion has been made to this or any other court.

[Remainder of Page Intentionally Left Blank]

WHEREFORE, the Debtors respectfully request (a) entry of the Interim Order, substantially in the form attached hereto as **Exhibit A**, (b) granting the relief requested herein, (c) scheduling the Final Hearing to consider entry of the Final Order, and (d) granting such other relief as is just and proper.

Dated: March 5, 2025 Atlanta, Georgia

/s/ Jeffrey R. Dutson

Jeffrey R. Dutson Georgia Bar No. 637106 Brooke L. Bean Georgia Bar No. 764552 Alice Kyung Won Song Georgia Bar No. 692753

KING & SPALDING LLP

1180 Peachtree Street NE Atlanta, Georgia 30309 Telephone: (404) 572-4600 Email: jdutson@kslaw.com

Email: bbean@kslaw.com Email: asong@kslaw.com

Proposed Counsel for the Debtors in Possession

Exhibit A

The Interim Order

UNITED STATES BANKRUPTCY COURT NORTHERN DISTRICT OF GEORGIA ATLANTA DIVISION

1)	Chapter 11
In re)	Case No. 25-52415 (SMS)
OTB HOLDING LLC, et al., ¹)	,
)	(Joint Administration Requested)
Debtors.)	
	_)	

INTERIM ORDER (I) AUTHORIZING THE DEBTORS TO OBTAIN SENIOR SECURED SUPERPRIORITY POSTPETITION FINANCING; (II) GRANTING (A) LIENS AND SUPERPRIORITY ADMINISTRATIVE EXPENSE CLAIMS AND (B) ADEQUATE PROTECTION; (III) AUTHORIZING USE OF CASH COLLATERAL; (IV) MODIFYING THE AUTOMATIC STAY; (V) SCHEDULING A FINAL HEARING; AND (VI) GRANTING RELATED RELIEF

Upon the motion (the "Motion")² of OTB Acquisition LLC ("OTB") and its affiliated debtors in the above-captioned chapter 11 cases (collectively, the "Chapter 11 Cases"), as debtors

The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number, include: OTB Holding LLC (3213), OTB Acquisition LLC (8500), OTB Acquisition of New Jersey LLC (1506), OTB Acquisition of Howard County LLC (9865), Mt. Laurel Restaurant Operations LLC (5100), OTB Acquisition of Kansas LLC (9014), OTB Acquisition of Baltimore County, LLC (6963). OTB Holding LLC's service address is One Buckhead Plaza, 3060 Peachtree Road, NW, Atlanta, GA 30305.

² Capitalized terms used but not otherwise defined herein have the meanings ascribed to them in the Motion.

and debtors in possession (collectively, the "<u>Debtors</u>") seeking entry of an interim order (this "<u>Interim DIP Order</u>") and a final order (the "<u>Final DIP Order</u>", and together with the Interim DIP Order, the "<u>DIP Orders</u>") pursuant to sections 105, 361, 362, 363, 364(c), 364(d), 364(e), 503, and 507 of title 11 of the United States Code (the "<u>Bankruptcy Code</u>"), rules 2002, 4001, 6003, 6004, and 9014 of the Federal Rules of Bankruptcy Procedure (the "<u>Bankruptcy Rules</u>"), and rules 2002-1, 4001-1(b), 4002-1, and 9013-1(i) of the Local rules of the United States Bankruptcy Court for the Northern District of Georgia (the "Local Bankruptcy Rules"), seeking, among other things:

- (i) authorizing the Debtors to incur senior secured postpetition obligations on a superpriority basis in respect of a senior secured superpriority delayed multipledraw term loan facility in the aggregate principal amount of \$14 million (the "DIP Facility") (provided that such \$14 million commitment amount excludes all interest, fees and expenses that will be capitalized as part of the aggregate outstanding principal amount under the DIP Facility, and, all amounts extended, deemed extended or capitalized under the DIP Facility, the "DIP Loans"), comprised of: (a) upon entry of this Interim DIP Order, \$11.5 million, including \$7.5 million of new money DIP Loans (the "Interim Advance"), plus a roll-up and conversion into DIP Loans of \$4 million of the outstanding principal balance under the Prepetition Secured Note (as defined below) (the "Roll-Up"), plus capitalization of the DIP Facility Fee (defined below) and (b) upon entry of a Final DIP Order, an additional \$2.5 million of new money DIP Loans (for a total of \$10 million of new money DIP Loans (including amounts deemed to have been borrowed to pay the DIP Facility Fee)), pursuant to the terms and conditions of this Interim DIP Order, the Final DIP Order, and a Secured Super-priority Debtorin-Possession Credit Agreement, a copy of which is attached hereto as Exhibit 2 (as the same may be amended, restated, supplemented, waived, or otherwise modified from time to time, consistent with the terms of this Interim DIP Order and any Final DIP Order, the "DIP Credit Agreement"), by and among OTB, as borrower (the "DIP Facility Borrower"), and its affiliates who are Debtors in these Chapter 11 Cases, as guarantors (such guarantors together with the DIP Facility Borrower, the "DIP Facility Loan Parties" and, each, a "DIP Facility Loan Party"), and OTB Lender, LLC (the "DIP Lender");
- (ii) authorizing the Debtors to execute and deliver the DIP Credit Agreement in form and substance consistent with the terms of this Interim DIP Order, and any other agreements, instruments, pledge agreements, guarantees, fee letters, control agreements, and other ancillary documents related thereto (including any security agreements, intellectual property security agreements, or notes) (as amended,

- restated, supplemented, waived, and/or modified from time to time, collectively with the DIP Credit Agreement, the "<u>DIP Loan Documents</u>") and to perform such other acts as may be necessary or desirable in connection with the DIP Loan Documents;
- (iii) authorizing the Debtors to borrow the Interim Advance, which shall be available upon entry of the Interim DIP Order to avoid immediate and irreparable harm;
- (iv) approving an upfront commitment fee in an amount equal to 0.50% of the \$14 million DIP Facility, which such amount shall be deemed to have been borrowed concurrently with the Interim Advance and such amount will be capitalized and added to the principal amount outstanding under the DIP Facility (the "DIP Facility Fee");
- (v) subject to and subordinate in all respects to the Carve-Out (as defined below), granting the DIP Facility and all obligations owing thereunder and under, or secured by, the DIP Loan Documents, to the DIP Lender (collectively, and including all "Obligations" to be defined in the DIP Credit Agreement, the "DIP Obligations") allowed superpriority administrative expense claim status in each of the Chapter 11 Cases and any Successor Cases (as defined below);
- (vi) subject to the Carve-Out, granting to the DIP Lender automatically and validly perfected security interests in and liens on all of the DIP Collateral (as defined below), including all property constituting Cash Collateral (as defined below);
- (vii) authorizing and directing the Debtors to pay the principal, interest, premiums, fees, expenses, and other amounts payable under the DIP Loan Documents as such become earned, due and payable, including the DIP Facility Fee, any administrative agent's fees, the reasonable and documented fees and disbursements of the DIP Lender's attorneys (including Porter Hedges LLP), advisors, accountants, and other consultants, all to the extent provided in, and in accordance with, the DIP Loan Documents;
- (viii) authorizing the Debtors to use the Prepetition Collateral, including the Cash Collateral of the Prepetition Bridge Lender (as defined below) under (a) each of the agreements, instruments, and other documents executed in connection with the Prepetition Secured Note; and (b) the Prepetition Secured Note;
- (ix) authorizing the Debtors to use the assets ("<u>CrossFirst Collateral</u>") subject to junior liens held by CrossFirst Bank ("<u>CrossFirst</u>"), including the cash collateral of CrossFirst, and providing adequate protection to CrossFirst for any diminution in value of the CrossFirst Collateral from and after the Petition Date to the extent such diminution in value occurs on account of the Debtors' sale, lease or use of the CrossFirst Collateral, the priming of the liens securing the obligations to CrossFirst

- (the "CrossFirst Liens"), and the imposition of the automatic stay pursuant to section 362 of the Bankruptcy Code ("Diminution in Value");
- subject to and effective upon entry of the Final DIP Order, a waiver of (a) the Debtors' right to surcharge any collateral pursuant to sections 105(a) and 506(c) of the Bankruptcy Code or otherwise and (b) the equitable doctrine of marshaling and other similar doctrines;
- (xi) modifying the automatic stay imposed by section 362 of the Bankruptcy Code to the extent necessary to implement and effectuate the terms and provisions of the DIP Loan Documents and this Interim DIP Order; and
- (xii) scheduling a final hearing (the "<u>Final Hearing</u>") to consider the relief requested in the Motion and approving the form of notice with respect to the Final Hearing.

This Court having considered the Motion, the exhibits attached thereto, the Declaration of Jonathan Tibus in Support of Debtors' Motion for Interim and Final Orders (I) Authorizing the Debtors to Obtain Senior Secured Superpriority Postpetition Financing; (II) Granting (A) Liens and Superpriority Administrative Expense Claims and (B) Adequate Protection; (III) Authorizing Use of Cash Collateral; (IV) Modifying the Automatic Stay; (V) Scheduling a Final Hearing; and (VI) Granting Related Relief, the First Day Declaration, and the evidence submitted and arguments made at the interim hearing held on [], 2025 (the "Interim Hearing"); and it appearing that proper and adequate notice of the Motion has been given under the circumstances and no further notice need be provided; and all objections, if any, to the relief requested in the Motion having been withdrawn, resolved, or overruled by the Court; and it appearing that approval of the interim relief requested in the Motion is necessary to avoid immediate and irreparable harm to the Debtors pending the Final Hearing and is otherwise fair and reasonable and in the best interests of the Debtors, their creditors, and their estates, and essential for the continued operation of the Debtors' businesses and maximization of the value of the Debtors' assets; and it appearing that the Debtors' entry into the DIP Credit Agreement and the DIP Loan Documents is a sound and prudent exercise of the Debtors' business judgment; and after due deliberation and consideration; and for good and sufficient cause appearing therefor;

BASED UPON THE RECORD ESTABLISHED AT THE INTERIM HEARING, THE COURT MAKES THE FOLLOWING FINDINGS OF FACT AND CONCLUSIONS OF LAW:³

- A. <u>Petition Date</u>. On March 4, 2025 (the "<u>Petition Date</u>"), each of the Debtors filed a voluntary petition for relief under chapter 11 of the Bankruptcy Code in this Bankruptcy Court for the Northern District of Georgia (the "<u>Court</u>," and the cases initiated by the voluntary petitions, the "<u>Bankruptcy Cases</u>").
- B. <u>Debtors in Possession</u>. The Debtors have continued in the management and operation of their businesses and properties as debtors in possession pursuant to sections 1107 and 1108 of the Bankruptcy Code.
- C. <u>Jurisdiction and Venue</u>. This Court has jurisdiction over the Chapter 11 Cases, the Motion, and the parties and property affected hereby pursuant to 28 U.S.C. § 1334. Consideration of the Motion constitutes a core proceeding pursuant to 28 U.S.C. § 157(b)(2). Venue for the Chapter 11 Cases and proceedings with respect to the Motion is proper before this Court pursuant to 28 U.S.C. § 1408.
- D. <u>Committee Formation</u>. As of the date hereof, the United States Trustee for the Northern District of Georgia (the "<u>U.S. Trustee</u>") has not yet appointed an official

The findings and conclusions set forth herein constitute the Court's findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052, made applicable to this proceeding pursuant to Bankruptcy Rule 9014. To the extent that any of the following findings of fact constitute conclusions of law, they are adopted as such. To the extent any of the following conclusions of law constitute findings of fact, they are adopted as such.

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 48 of 171

committee of unsecured creditors in the Chapter 11 Cases (a "<u>Committee</u>") pursuant to section 1102 of the Bankruptcy Code.

- E. <u>Notice</u>. Notice of the Motion and the Interim Hearing has been provided in accordance with the Bankruptcy Code, the Bankruptcy Rules, and the Local Bankruptcy Rules, and no other or further notice of the Motion with respect to the relief requested at the Interim Hearing or the entry of this Interim DIP Order shall be required.
- F. Debtors' Stipulations. After consultation with their attorneys and financial advisors, and without prejudice to the rights of other parties in interest, including any Committee, the Debtors admit, stipulate, acknowledge, and agree to the statements set forth in this paragraph (collectively, the "Debtors' Stipulations"): (i) on February 14, 2025, OTB Lender, LLC, a Delaware corporation (the "Prepetition Bridge Lender"), entered into that certain Secured Promissory Note dated as of February 14, 2025 (such agreement, as amended, the "Prepetition Secured Note"); (ii) as of the Petition Date, the Debtors were Borrowers (as defined in the Prepetition Secured Note) and thus were justly and lawfully indebted and liable to the Prepetition Bridge Lender, without defense, counterclaim, or offset of any kind, in an aggregate principal amount of not less than \$4 million plus accrued and unpaid interest and all other obligations of whatever nature incurred in connection therewith which are chargeable or otherwise reimbursable under the applicable agreements or applicable law (collectively, the "Prepetition Secured Note Obligations"); (iii) as more fully set forth in the Prepetition Secured Note, prior to the Petition Date, the Debtors granted legal, valid, perfected, binding, enforceable, and nonavoidable liens and security interests on all of the Pledged Collateral (as defined in the Prepetition Secured Note) (the "Prepetition Collateral") to secure the Prepetition Secured Note Obligations (collectively, the

"Prepetition Liens"); (iv) the Debtors acknowledge and agree that as of the Petition Date (a) the Prepetition Liens on the Prepetition Collateral were valid, binding, enforceable, non-avoidable, and properly perfected and were granted to, or for the benefit of, the Prepetition Bridge Lender for fair consideration and reasonably equivalent value; (b) the Prepetition Liens were senior in priority over any and all other liens on the Prepetition Collateral, subject only to certain liens senior by operation of law, solely to the extent any such liens were, as of the Petition Date, valid, properly perfected (or are perfected subsequent to the Petition Date as permitted by section 546(b)), non-avoidable, and senior in priority to the Prepetition Liens; and (v) no claims or causes of action exist that are held by the Debtors or their estates against, or with respect to, the Prepetition Bridge Lender under the Prepetition Secured Note or any other applicable law. The Debtors and their estates have no valid claims, objections, challenges, causes of action, and/or choses in action against any of the Prepetition Bridge Lender or any of its respective affiliates. agents, attorneys, advisors, professionals, officers, directors, and employees with respect to the Prepetition Secured Note, the Prepetition Secured Note Obligations, or the Prepetition Liens whether arising at law or at equity, including, without limitation, any challenge, recharacterization, subordination, avoidance, recovery, disallowance, reduction, or other claims arising under or pursuant to sections 105, 502, 510, 541, 542 through 553, inclusive, or 558 of the Bankruptcy Code or applicable state law equivalents.

G. <u>Cash Collateral</u>. All of the Debtors' cash, including any cash in their deposit accounts, wherever located, whether as original collateral or proceeds of other Prepetition Collateral, constitutes or will constitute cash collateral ("<u>Cash Collateral</u>") of the Prepetition Bridge Lender, the DIP Lender, and CrossFirst.

H. Findings Regarding Postpetition Financing and Use of Cash

Collateral.

- (i) Request for Postpetition Financing and Use of Cash Collateral. The Debtors seek authority to (a) enter into the DIP Facility and incur the DIP Obligations on the terms described herein and (b) use Cash Collateral on the terms described herein, in each case, to administer their Chapter 11 Cases and fund their operations. At the Final Hearing, the Debtors will seek final approval of the DIP Facility and use of Cash Collateral pursuant to the Final DIP Order and the DIP Loan Documents, which both shall be in form and substance acceptable to the DIP Lender and not inconsistent with the terms of this Interim DIP Order. Notice of the Final Hearing, the DIP Loan Documents, and the proposed Final DIP Order will be provided in accordance with this Interim DIP Order.
- (ii) Priming of the Prepetition Liens. The consensual priming of the CrossFirst Liens under section 364(d) of the Bankruptcy Code, as provided herein, will enable the Debtors to obtain the financing needed to continue to operate their business during the pendency of the Chapter 11 Cases, to the benefit of their estates and creditors. CrossFirst is entitled to receive adequate protection as set forth in this Interim DIP Order pursuant to sections 361, 363, and 364 of the Bankruptcy Code, solely to the extent of any Diminution in Value of its interests in the CrossFirst Collateral (including Cash Collateral) as a result of the priming of the CrossFirst Liens. CrossFirst has consented to the priming of the CrossFirst Liens.
- (iii) Need for Postpetition Financing and Use of Cash Collateral. The Debtors have a need to use Cash Collateral on an interim basis and obtain credit in an amount equal to the Interim Advance pursuant to the DIP Facility in order to, among other things, enable the orderly continuation of their operations and to administer and preserve the value of their estates. The ability of the Debtors to maintain business relationships with their vendors, suppliers, and customers, to pay their employees, and otherwise finance their operations requires the availability of working capital from the DIP Facility and the use of Cash Collateral, the absence of either of which, on an interim basis as contemplated hereunder, would immediately and irreparably harm the Debtors, their estates, and parties in interest. The Debtors do not have sufficient available sources of working capital and financing to operate their businesses, maintain their properties in the ordinary course of business and fund the Chapter 11 Cases without the authorization to use Cash Collateral and to borrow the Interim Advance.
- (iv) No Credit Available on More Favorable Terms. The DIP Facility is the best source of debtor in possession financing available to the Debtors. Given their current financial condition, financing arrangements, and capital structure, the Debtors have been and continue to be unable to obtain financing from sources other than the DIP Lender on terms more favorable than the DIP Facility. The Debtors are unable to obtain unsecured credit allowable under section 503(b)(1) of the Bankruptcy Code as an administrative expense. The Debtors have also been unable to obtain (a) unsecured credit having priority over that of administrative expenses of the kind specified in sections

- 503(b), 507(a), and 507(b) of the Bankruptcy Code; (b) credit secured solely by a lien on property of the Debtors and their estates that is not otherwise subject to a lien; or (c) credit secured solely by a junior lien on property of the Debtors and their estates that is subject to a lien. Financing on a postpetition basis on better terms is not available without granting the DIP Lender, (1) perfected security interests in and liens on (each as provided herein) the DIP Collateral, with the priorities set forth herein; (2) superpriority administrative claims; and (3) the other protections set forth in this Interim DIP Order.
- Use of Cash Collateral and Proceeds of the DIP Facility. As a condition to entry into the DIP Credit Agreement, the extension of credit under the DIP Facility, and the authorization to use the Prepetition Collateral, including Cash Collateral, the DIP Lender (in its capacity as the Prepetition Bridge Lender) requires, and the Debtors have agreed, that proceeds of the DIP Facility and the DIP Lender's Cash Collateral shall be used in a manner consistent with the terms and conditions of this Interim DIP Order, and in a manner consistent with the budget (as the same may be modified from time to time with the written consent of the DIP Lender in its sole discretion and consistent with the terms of this Interim DIP Order, the "Budget," a copy of which is attached hereto as Exhibit 1), solely for the purposes set forth in this Interim DIP Order, including (a) ongoing working capital and other general corporate purposes of the Debtors; (b) permitted payment of costs of administration of the Chapter 11 Cases, including restructuring charges arising on account of the Chapter 11 Cases, including statutory fees of the U.S. Trustee (the "Statutory Fees") and allowed professional fees and expenses of the Professionals (as defined below), subject to the Investigation Budget (as defined below); (c) payment of such prepetition expenses as consented to by the DIP Lender or otherwise permitted under this Interim DIP Order; (d) payment of interest, premiums, fees, expenses, and other amounts (including, without limitation, legal and other professionals' fees and expenses of the DIP Lender) owed under this Interim DIP Order, including those incurred in connection with the preparation, negotiation, documentation, and Court approval of the DIP Facility, whether incurred before, on, or after the Petition Date; (e) funding the Roll-Up; and (f) payment of obligations arising from or related to the Carve-Out.
- (vi) Roll-Up of Prepetition Secured Note Obligations. Upon the entry of the Interim DIP Order, \$4 million of the outstanding principal amount of the Prepetition Secured Note Obligations (and all of the accrued and unpaid interest) shall be converted into DIP Loans in accordance with this Interim DIP Order. The DIP Lender would not otherwise provide the DIP Facility without the inclusion of the Roll-Up.
- (vii) Application of Proceeds of DIP Collateral. As a condition to entry into a DIP Credit Agreement, the extension of credit under the DIP Facility and authorization to use Cash Collateral, the Debtors, and the DIP Lender have agreed that, as of and commencing on the date of the Interim Hearing, the Debtors shall utilize the proceeds of the DIP Collateral in accordance with this Interim DIP Order.

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 52 of 171

- I. <u>Adequate Protection</u>. CrossFirst is entitled to receive adequate protection of its interests in the CrossFirst Collateral, including, without limitation, the Cash Collateral.
- J. <u>Sections 506(c) and Marshaling.</u> In light of (i) the DIP Lender's agreement that its DIP Liens (as defined below) and superpriority claims shall be subject and subordinate in all respects to the Carve-Out; and (ii) the DIP Lender's agreement to the payment (in a manner consistent with the Budget, and subject to the terms and conditions of this Interim DIP Order) of certain expenses of administration of these Chapter 11 Cases, upon entry of a Final DIP Order, the Debtors shall waive the provisions of section 506(c) of the Bankruptcy Code and the equitable doctrine of marshaling and other similar doctrines.
- Based upon the pleadings and proceedings of record in the Chapter 11 Cases, (a) the extensions of credit under the DIP Facility are fair and reasonable, are appropriate for secured financing to debtors in possession, are the best available to the Debtors under the circumstances, reflect the Debtors' exercise of prudent business judgment consistent with their fiduciary duties, and are supported by reasonably equivalent value and fair consideration; (b) the terms and conditions of the DIP Facility and the use of the Cash Collateral have been negotiated in good faith and at arm's-length among the Debtors, the DIP Lender, and the Prepetition Bridge Lender with the assistance and counsel of their respective advisors; (c) the use of Cash Collateral, including, without limitation, pursuant to this Interim DIP Order, has been allowed in "good faith" within the meaning of section 364(e) of the Bankruptcy Code; (d) any credit to be extended, loans to be made, and other financial accommodations to be extended to the Debtors by the DIP Lender or the Prepetition Bridge Lender, including, without limitation, pursuant to this Interim DIP Order,

have been allowed, advanced, extended, issued, or made, as the case may be, in "good faith" within the meaning of section 364(e) of the Bankruptcy Code by the DIP Lender and the Prepetition Bridge Lender in express reliance upon the protections offered by section 364(e) of the Bankruptcy Code; and (e) the DIP Facility, the DIP Liens, and the DIP Superpriority Claims (as defined below) shall be entitled to the full protection of section 364(e) of the Bankruptcy Code in the event that this Interim DIP Order or any provision hereof is vacated, reversed, or modified, on appeal or otherwise.

- L. <u>Immediate Entry</u>. Sufficient cause exists for immediate entry of this Interim DIP Order pursuant to Bankruptcy Rules 4001(b)(2) and 4001(c)(2) and the Local Bankruptcy Rules.
- M. <u>Interim Hearing</u>. Notice of the Interim Hearing and the relief requested in the Motion has been provided by the Debtors, whether by electronic mail, overnight courier, or first class mail to certain parties in interest, including the Notice Parties (as defined in the Motion). The Debtors have made reasonable efforts to afford the best notice possible under the circumstances, and no other notice is required in connection with the relief set forth in this Interim DIP Order.

Based upon the foregoing findings and conclusions, the Motion, and the record before the Court with respect to the Motion, and after due consideration and good and sufficient cause appearing therefor, **IT IS HEREBY ORDERED THAT**:

1. <u>DIP Facility Approved on an Interim Basis</u>. On an interim basis, the DIP Facility is hereby authorized and approved to the extent set forth herein, and the use of Cash Collateral on an interim basis is authorized, in each case subject to the terms and conditions set forth in this

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 54 of 171

Interim DIP Order. All objections to entry of this Interim DIP Order, to the extent not withdrawn, waived, settled, or resolved, are hereby denied and overruled. This Interim DIP Order shall become effective immediately upon its entry.

2. Authorization of the DIP Facility. The Debtors are expressly and immediately authorized and empowered to execute and deliver the DIP Loan Documents and to incur and to perform the DIP Obligations in accordance with, and subject to, the terms of this Interim DIP Order and the DIP Loan Documents, and to deliver all instruments, certificates, agreements, and documents that may be required or necessary for the performance by the Debtors under the DIP Facility and the creation and perfection of the DIP Liens. The Debtors shall pay, in accordance with this Interim DIP Order, the principal, interest, premiums, fees, payments, expenses, and other amounts described in the DIP Loan Documents as such amounts become due and payable, without the need to obtain further Court approval, whether or not such fees arose before, on, or after the Petition Date, and whether or not the transactions contemplated hereby are consummated, and to take any other actions that may be necessary or appropriate, all as provided in this Interim DIP Order or the DIP Loan Documents; provided that the payment of legal and other professionals' fees and expenses of the DIP Lender (other than legal and other professionals' fees and expenses incurred prior to the Petition Date) shall be subject to the requirements of paragraph 26 hereof. All collections and proceeds, whether from ordinary course collections, asset sales, debt or equity issuances, insurance recoveries (subject to paragraph 35 herein), condemnations, or otherwise, will be deposited and applied as required by this Interim DIP Order and the DIP Loan Documents. Upon execution and delivery, the DIP Loan Documents shall represent valid and binding obligations of the Debtors, enforceable against each of the Debtors and their estates in accordance Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 55 of 171

with their terms. Upon the entry of this Interim DIP Order, the DIP Facility Fee shall be fully earned and non-refundable and, concurrently with the Interim Advance under the DIP Facility, the full amount of the DIP Facility Fee shall be capitalized and added to the aggregate principal amount outstanding under the DIP Loans.

- 3. Authorization to Borrow and Use Cash Collateral. To prevent immediate and irreparable harm to the Debtors' estates, from the entry of this Interim DIP Order through and including the entry of the Final DIP Order and subject to the terms, conditions, and limitations on availability set forth in this Interim DIP Order, the Debtors are hereby authorized to (i) borrow under the DIP Facility in an aggregate outstanding principal amount equal to the Interim Advance, plus an amount equal to the Facility Fee, plus an amount equal to the aggregate amount of the Roll-Up, and the Debtors shall be authorized to consummate the Roll-Up, and (ii) use the Cash Collateral for the purposes described in this Interim DIP Order.
- 4. <u>DIP Obligations</u>. This Interim DIP Order shall constitute and evidence the validity and binding effect of the DIP Obligations, which shall be enforceable against each of the Debtors, their estates, and any successors thereto, including, without limitation, any trustee appointed in the Chapter 11 Cases or in any case under chapter 7 of the Bankruptcy Code upon the conversion of any of the Chapter 11 Cases, or in any other proceedings superseding or related to any of the foregoing (collectively, the "<u>Successor Cases</u>"). Upon entry of this Interim DIP Order, the DIP Obligations will include all loans and any other indebtedness or obligations, contingent or absolute, which may now or from time to time be owing by any of the Debtors to the DIP Lender, in each case, under this Interim DIP Order or secured by the DIP Liens, including, without limitation, all principal, accrued and unpaid interest, costs, fees, expenses, and other amounts

owing under this Interim DIP Order. The Debtors shall be jointly and severally liable for the DIP Obligations. No obligation, payment, transfer, or grant of collateral security hereunder (including any DIP Obligations or DIP Liens) shall be stayed, restrained, voidable, avoidable, or recoverable under the Bankruptcy Code or under any applicable law (including, without limitation, under chapter 5 of the Bankruptcy Code, section 724(a) of the Bankruptcy Code, or any other provision with respect to avoidance actions under the Bankruptcy Code or applicable state or foreign law equivalents ("Avoidance Actions") or subject to any avoidance, reduction, setoff, recoupment, offset, recharacterization, subordination (whether equitable, contractual, or otherwise, but other than to the Carve-Out), counterclaim, cross-claim, defense, or any other challenge under the Bankruptcy Code or any applicable law or regulation by any person or entity.

5. <u>DIP Liens.</u> All DIP Obligations shall be secured by the following liens and security interests (the "<u>DIP Liens</u>") on all assets and properties of the Debtors (whether tangible, intangible, real, personal or mixed), whether now owned or hereafter acquired and wherever located, before or after the Petition Date, including, without limitation, all cash, cash equivalents, accounts, inventory, equipment, equity interests or capital stock in subsidiaries, investment property, instruments, chattel paper, real property, leasehold interests, contracts, patents, copyrights, trademarks and other general intangibles, receivables, all claims or causes of action (excluding avoidance actions arising under Chapter 5 of the Bankruptcy Code but including all proceeds thereof) and all products, offspring, profits and proceeds of each of the foregoing (collectively, the "<u>DIP Collateral</u>"; provided that notwithstanding the definition of DIP Collateral, (i) DIP Liens shall only attach to the Debtors' liquor licenses to the extent permitted by applicable non-bankruptcy law, (ii) to the extent applicable non-bankruptcy law does not permit the DIP

Liens to attach directly to the Debtors' liquor licenses, the DIP Liens shall be valid, binding, continuing, enforceable, non-avoidable, and automatically and properly perfected security interests and liens in any proceeds from any sale or other disposition of the Debtors' liquor licenses and such proceeds shall constitute DIP Collateral and (iii) in no event shall DIP Collateral include any leasehold interest in non-residential real property that prohibits or restricts the granting of liens thereon (except as permitted pursuant to applicable non-bankruptcy law), but DIP Collateral shall include the proceeds of the sale or other disposition of such leases) and the DIP Liens shall be:

- (a) secured pursuant to section 364(c)(2) of the Bankruptcy Code, by a first-priority perfected lien on, and security interest in, all present and after-acquired property of the Debtors, wherever located, that is not subject to a perfected lien or security interest on the Petition Date;
- (b) secured pursuant to section 364(c)(3) of the Bankruptcy Code, by a junior perfected lien on, and security interest in, all present and after-acquired property of the Debtors, wherever located, that is subject to a valid, perfected, enforceable and unavoidable lien or security interest (excluding the CrossFirst Liens) on the Petition Date or subject to a lien or security interest (excluding the CrossFirst Liens) in existence on the Petition Date that is perfected subsequent thereto as permitted by section 546(b) of the Bankruptcy Code (including, without limitation, the Carve-Out (as defined below)), in each case, that is expressly permitted to be senior to the DIP Liens pursuant to the Interim DIP Order or the DIP Credit Agreement;
- (c) secured pursuant to section 364(d)(1) of the Bankruptcy Code, a first-priority, perfected senior priming lien on, and security interest in, all present and after-acquired property of the Debtors, wherever located, that is or was subject to a perfected lien or security interest on the Petition Date (including the CrossFirst Liens but excluding the liens and security interests set forth in clause (b) of this subparagraph); and
- (d) other than as set forth herein (including with respect to the Carve-Out), the DIP Liens shall not be made subject to or *pari passu* with any lien or security interest heretofore or hereinafter granted in the Chapter 11 Cases or any Successor Cases and shall be valid and enforceable against any trustee appointed in the Chapter 11 Cases or any Successor Cases, upon the conversion of any of the Chapter 11 Cases to any Successor Case, and/or upon the dismissal or conversion of any of the

Chapter 11 Cases or Successor Cases. The DIP Liens shall not be subject to any of sections 510, 549, or 550 of the Bankruptcy Code. No lien or interest avoided and preserved for the benefit of the estate pursuant to section 551 of the Bankruptcy Code shall be *pari passu* with or senior to the DIP Liens.

For the avoidance of doubt, the DIP Liens (i) shall be subject only to (x) the Carve-Out and (y) validly perfected and non-avoidable liens (excluding the CrossFirst Liens) existing as of the Petition Date to which the liens securing the obligations under the Prepetition Secured Note were subject, and (ii) shall not extend to, and the DIP Collateral shall not consist of, Avoidance Actions but they shall extend to the proceeds thereof. The DIP Liens and security interests of the DIP Lender shall not prime any Permitted Liens (as defined in the DIP Credit Agreement).

- 6. <u>Superpriority Claims</u>. Subject to and subordinate in all respects to the Carve-Out and in the priority set forth herein, upon entry of this Interim DIP Order, the DIP Lender is hereby granted, pursuant to section 364(c)(1) of the Bankruptcy Code, allowed superpriority administrative expense claims against each of the Debtors in each of the Chapter 11 Cases and any Successor Cases (collectively, the "<u>DIP Superpriority Claims</u>") for all DIP Obligations with priority over any and all administrative expense claims and unsecured claims against the Debtors or their estates in any of the Chapter 11 Cases or any Successor Cases, at any time existing or arising, of any kind or nature whatsoever, including, without limitation, administrative expenses of the kinds specified in or ordered pursuant to sections 105, 326, 327, 328, 330, 331, 361, 362, 363, 364, 503(a), 503(b), 507(a), 507(b), 546(c), 546(d), 726, 1113, or 1114 of the Bankruptcy Code or any other provision of the Bankruptcy Code.
- 7. <u>Interest</u>. Interest shall be computed and accrue monthly on the outstanding principal amount of all draws under the DIP Facility at a rate of 12% per annum. All accrued interest will be paid-in-kind and capitalized on the first business day of each month (and thereby

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 59 of 171

added to principal, which shall thereafter accrue interest). Effective immediately upon the occurrence of an Event of Default (as defined below), interest on the DIP Loans under the DIP Facility shall accrue at a rate that is 15% per annum.

- 8. No Obligation to Extend Credit. The DIP Lender shall have no obligation to make any loan under this Interim DIP Order and/or the Final DIP Order, as applicable, unless all conditions set forth in the DIP Loan Documents, have been satisfied in full or waived by the DIP Lender in accordance with the terms of this Interim DIP Order and the DIP Facility.
- 9. <u>Use of Proceeds of DIP Facilities</u>. From and after the Petition Date, the Debtors shall use proceeds of borrowings under the DIP Facility only for the purposes specifically set forth in this Interim DIP Order, and, in each case, in a manner consistent with the Budget and the terms and conditions in this Interim DIP Order, including for (i) the Roll-Up, (ii) to fund postpetition operating expenses and working capital needs of the Debtors; (iii) to pay interest, fees, and expenses (including attorneys' fees) to the DIP Lender pursuant to this Interim DIP Order (whether or not such amounts are reflected in the Budget); (d) to fund fees and expenses incurred in connection with the Sale Transaction (as defined below); (e) to pay the fees and expenses incurred by, and/or payable to, any Professional (the "<u>Professional Fees</u>") and expenses solely as provided for in the Budget; and (f) to pay certain other costs and expenses of administration of the Chapter 11 Cases.
- 10. <u>Authorization to Use Cash Collateral</u>. Subject to the terms and conditions of this Interim DIP Order, and in a manner consistent with the Budget, the Debtors are authorized to use Cash Collateral. Nothing in this Interim DIP Order shall authorize the disposition of any assets of the Debtors or their estates outside the ordinary course of business, or any Debtor's use of any

Cash Collateral or other proceeds resulting therefrom, except as permitted in this Interim DIP Order (including with respect to the Carve-Out).

11. Adequate Protection for CrossFirst. CrossFirst is entitled, pursuant to sections 361, 362, 363(c)(2), 363(e), and 507 of the Bankruptcy Code, to adequate protection of its interests in the applicable Prepetition Collateral, including Cash Collateral, solely to the extent of any Diminution in Value of its interests in the Prepetition Collateral. As adequate protection, CrossFrist is hereby granted (the "Adequate Protection"), effective and perfected upon the Petition Date and without the necessity of the execution by the Debtors of security agreements, pledge agreements, mortgages, financing statements, or other agreements, a valid, perfected replacement security interest in and lien on the CrossFirst Collateral (the "Adequate Protection Liens"), which security interests and liens shall be subject to and subordinate in all respects to the Carve-Out, the DIP Liens, and any liens or security interests that are valid, perfected, enforceable and unavoidable as of the Petition Date. The Adequate Protection Liens shall be in addition to all valid and enforceable liens and security interests now existing in favor of CrossFirst and not in substitution therefor. The Adequate Protection Liens shall not attach to Avoidance Actions or the proceeds thereof. The Adequate Protection Liens and the CrossFirst Liens shall be subject to and subordinate to the Carve-Out.

12. Budget Maintenance.

(a) The Debtors shall use the proceeds of all borrowings under the DIP Facility and Cash Collateral in a manner consistent with the Budget and exclusions set forth herein. The Budget annexed hereto as **Exhibit 1** shall constitute the initial Budget. The Debtors shall update the Budget on or before the entry of the Final DIP Order and, thereafter, every four (4) weeks

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 61 of 171

(with such Budget update (along with a cash flow forecast (in Microsoft Excel) through the anticipated consummation of the Sale Transaction for all of Debtors' assets to the DIP Lender in the same format as the cash flow forecast delivered at or before the Petition Date) being delivered to the DIP Lender no later than the Thursday of the week before entry of the Final DIP Order (or, to the extent applicable, the week before the conclusion of such four-week period) (or delivered more frequently and on such dates as consented to exclusively by the DIP Lender), which shall be subject to the DIP Lender's written approval in its sole discretion. During any period a proposed Budget has not been approved, the previously approved Budget shall be the Budget. Each Budget delivered to the DIP Lender shall be accompanied by such supporting documentation as reasonably requested by the DIP Lender and its advisors, and each Budget shall be prepared in good faith based upon assumptions the Debtors believe to be reasonable.

- (b) The Debtors' budgeted disbursements shall be tested on a rolling four-week basis, and the aggregate amount of such disbursements may not vary unfavorably from the applicable Budget more than ten percent (10%) for each four-week Budget period (the "Budget Variances"); provided, however, that: (i) such testing shall not commence until after the second week after the Petition Date; and (ii) in determining compliance with the Budget, all disbursements to the DIP Lender's professionals shall be excluded from the calculation.
- (c) For the avoidance of doubt, the Budget Variances shall not include or otherwise require a test of payment on Professional Fees; *provided*, *however*, that payment on Professional Fees shall not vary unfavorably from the applicable Budget on a cumulative fourweek basis per Professional Fee line item for that portion of the Budget period then ending (and

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 62 of 171

each professional that may receive Professional Fees shall be reflected on its own Budget line item).

- (d) On or before Thursday of each week, commencing with the first week following the Petition Date, the Debtors shall deliver to the DIP Lender a current report that:

 (i) details the actual amount of cash receipts and disbursements for the prior week for each line item included in the Budget (on a weekly and cumulative basis); (ii) compares such actual cash receipts and disbursements (on a line item by line item basis) with the weekly and cumulative budgeted amounts for each line item set forth in the Budget for such period; and (iii) provides an explanation for all material variances between budgeted amounts and actual amounts (an "Approved Budget Variance Report"). Each Approved Budget Variance Report will be certified as true and correct by the Debtors' chief restructuring officer.
- Modification of Automatic Stay. The automatic stay imposed under section 362(a)(2) of the Bankruptcy Code is hereby modified as necessary to effectuate all of the terms and provisions of this Interim DIP Order, including, without limitation, to: (a) permit the Debtors to grant the DIP Liens, Adequate Protection Liens, and DIP Superpriority Claims; (b) permit the Debtors to perform such acts as the DIP Lender may reasonably request to assure the perfection and priority of the liens granted herein; (c) permit the Debtors to incur all liabilities and obligations to the DIP Lender; and (d) authorize the Debtors to pay, and the DIP Lender to retain and apply, payments made in accordance with the terms of this Interim DIP Order.
- 14. <u>Cash Management</u>. The Debtors shall use a cash management system that is acceptable to the DIP Lender and the motion and orders to approve the use of the Debtors' cash

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 63 of 171

management system shall be acceptable to the DIP Lender. Any material changes from such cash management system must be acceptable to the DIP Lender.

15. Perfection of DIP Liens and Adequate Protection Liens. This Interim DIP Order shall be sufficient and conclusive evidence of the creation, validity, perfection, and priority of all liens granted herein, including the DIP Liens and the Adequate Protection Liens, without the necessity of filing or recording any financing statement, mortgage, notice, or other instrument or document which may otherwise be required under the law or regulation of any jurisdiction or the taking of any other action (including, for the avoidance of doubt, entering into any deposit account control agreement) to validate or perfect (in accordance with applicable non-bankruptcy law) the DIP Liens or the Adequate Protection Liens or to entitle the DIP Lender to the priorities granted herein. This Interim DIP Order shall be sufficient to grant the DIP Lender control of each deposit account subject to a DIP Lien. Notwithstanding the foregoing, the DIP Lender is authorized to file or record, in its sole discretion, as applicable, as each may deem necessary or advisable, such financing statements, security agreements, mortgages, notices of liens, and other similar documents to perfect its respective liens in accordance with applicable non-bankruptcy law, and all such financing statements, mortgages, notices, and other documents shall be deemed to have been filed, sent, or recorded as of the Petition Date. The Debtors shall execute and deliver upon reasonable request to the DIP Lender, all such financing statements, mortgages, notices, and other documents as the DIP Lender may reasonably request. The DIP Lender may file a photocopy of this Interim DIP Order as a financing statement with any filing or recording office or with any registry of deeds or similar office, in addition to or in lieu of such financing statements, notices of lien, or similar instrument, and all applicable officials shall accept a photocopy of this Interim DIP Order for filing or recordation for such purpose.

- 16. Events of Default. The following shall constitute "Events of Default:"
- (a) termination of the Debtors' Chief Restructuring Officer ("<u>CRO</u>") without the appointment of a successor that is acceptable to the DIP Lender;
- (b) resignation of the CRO without the appointment of a successor that is acceptable to the DIP Lender;
- (c) conversion of any of the Chapter 11 Cases to a case under Chapter 7 of the Bankruptcy Code or dismissal of any of the Chapter 11 Cases;
- (d) filing or support of a proposed plan of reorganization or liquidation by any Debtor that does not provide for the indefeasible payment in full and in cash of the Debtors' obligations outstanding under the DIP Facility;
- (e) appointment of a trustee for any of the Debtors under section 1104 of the Bankruptcy Code, or the filing of any motion or other pleading requesting such relief that is not timely opposed by the Debtors;
- (f) appointment of an examiner with enlarged powers (powers beyond those set forth in section 1106(a)(3) and (4) of the Bankruptcy Code) under section 1106(b) of the Bankruptcy Code for any of the Debtors, or the filing of a motion or other pleading requesting such relief that is not timely opposed by the Debtors;
- (g) entry of an order by the Court amending, supplementing, staying, vacating, or otherwise modifying the DIP Facility, the Interim DIP Order or the Final DIP Order approving

the DIP Facility, or the filing of a motion or other pleading requesting such relief that is not timely opposed by the Debtors;

- (h) any attempt by any Debtor to obtain, or if any other party in interest obtains, an order of the Court or other judgment, and the effect of such order or judgment is to, invalidate, reduce, or otherwise impair the DIP Lender's claims or to subject any of the Prepetition Collateral or the DIP Collateral to a surcharge pursuant to section 506(c) of the Bankruptcy Code;
- (i) request by any Debtor for approval of any postpetition financing other than the DIP Facility, except with the consent of the DIP Lender;
- (j) application by any Debtor for an order substituting any assets for all or any portion of the DIP Collateral;
- (k) entry of an order granting liens or claims that are senior to or *pari passu* with the liens granted in favor of the DIP Lender under the DIP Loan Documents;
- (l) any payment of, or application for authority to pay, any prepetition claim in whole or in part not included in an approved Budget;
 - (m) the sale process is terminated;
- (n) entry of an order granting any creditor (other than as provided under paragraph 13 of this Interim DIP Order) with a claim in excess of \$50,000 relief from the automatic stay;
 - (o) failure to make all payments under the DIP Facility when due;
 - (p) failure to pay any material postpetition indebtedness;
- (q) breach of any term or covenant of the DIP Facility or in any DIP Loan Document, including, without limitation, compliance with any Milestone (as defined below);

- (r) any representation or warranty by any Debtor is incorrect in any material respect when made;
- (s) exclusivity shall have been terminated or any Debtor shall have agreed to any such termination;
- (t) after entry thereof, either of (i) an order in form and substance satisfactory to the DIP Lender approving the bidding procedures to be applicable to the Sale Transaction, or (ii) an order entered by the Court in form and substance satisfactory to the DIP Lender that, among other things, approves the Sale Transaction, shall cease to be in full force and effect, shall have been reversed, stayed, vacated or subject to stay pending appeal or shall have been modified or amended;
- (u) action taken by any Debtor (or any Debtor's support for any other person in taking such action) to restrict or prohibit the DIP Lender from submitting a "credit bid" for any assets of the Debtors in accordance with Bankruptcy Code section 363(k);
- (v) failure by the Debtors to disburse the proceeds of the Sale Transaction to the DIP Lender contemporaneously with the closing of the Sale Transaction;
- (w) commencement of any action or the filing of any motion by the Debtors that is inconsistent with the terms of this Interim DIP Order; and
- (x) failure to pay the DIP Facility in full prior to May 30, 2025 (the "Maturity Date").
- 17. <u>Milestones</u>. The Debtors shall be required to comply with the following milestones (the "<u>Milestones</u>"), which may only be modified by the written consent of the DIP Lender in its sole discretion:

- (a) No later than five (5) days after the Petition Date, the Court shall have entered the Interim DIP Order in form and substance satisfactory to the DIP Lender;
- (b) No later than 30 days after the Petition Date, the Court shall have entered the Final DIP Order in form and substance satisfactory to the DIP Lender;
- (c) No later than two (2) days after the Petition Date, the Debtors shall have filed a motion to sell, and to approve bidding procedures for the sale of substantially all of the Debtors' assets (the "Assets") (the "Sale and Bidding Procedures Motion");
- (d) No later than 30 days after the Petition Date, the Court shall have entered an order in form and substance acceptable to the DIP Lender, approving the bidding procedures set forth in the Sale and Bidding Procedures Motion;
- (e) No later than 66 days after the Petition Date, the Court shall have entered an order (the "Sale Order") in form and substance acceptable to the DIP Lender, approving the sale of all or substantially all of the Assets; and
- (f) No later than 15 days after the entry of the Sale Order, the Debtors shall have consummated the Sale Transaction for all of Debtors' assets to the DIP Lender in a manner consistent with the Sale Order.
- 18. <u>Credit Bidding</u>. In connection with any sale process authorized by the Court, whether effectuated through sections 363, 725, or 1123 of the Bankruptcy Code, the DIP Lender shall have the right to "credit bid" pursuant to section 363(k) of the Bankruptcy Code all or any portion of the prepetition and postpetition secured obligations owed to it or any of its affiliated entities in the sale of the Debtors' assets, and shall not be prohibited or limited from making such credit bid "for cause" under section 363(k) of the Bankruptcy Code. Without limiting the

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 68 of 171

foregoing, the DIP Lender may assign all or any portion of the DIP Obligations owed to it to an acquisition vehicle, affiliate, or any other Person (any such Person, the "Agent Assignee"). The Agent Assignee shall have the right to "credit bid" any prepetition or postpetition secured obligations so assigned to it or assigned to or owned by any of its affiliated entities in any sale of the Debtors' assets.

- 19. Application of Sale Proceeds. Unless the DIP Lender or any Agent Assignee is the bidder selected by the Debtors as the highest and best bid for the assets included in the sale of all or substantially all of the assets of the Debtors under section 363 of the Bankruptcy Code (the "Sale Transaction"), subject and subordinate in all respects to the Carve-Out, proceeds from the Sale Transaction shall be paid pursuant to a Sale Order, through which the Debtors shall seek to be made directly to the DIP Lender upon the closing of the Sale Transaction and shall be applied: (i) first, to reduce outstanding DIP Facility obligations (including all fees and professional fees payable pursuant to the DIP Loan Documents); (ii) second, to outstanding Prepetition Secured Note Obligations, in the manner set forth in the Prepetition Secured Note, with the surplus remaining (if any) following satisfaction of the Prepetition Secured Note Obligations to be disbursed in the manner set forth in the Prepetition Secured Note, to the extent not rolled up; and (iii) with the surplus remaining (if any) to be disbursed in accordance with the Bankruptcy Code and any plan of liquidation.
- 20. <u>Proceeds of Subsequent Financing</u>. If the Debtors, any trustee, any examiner with expanded powers, or any responsible officer subsequently appointed in these Chapter 11 Cases or any Successor Cases shall obtain credit or incur debt pursuant to sections 364(b), 364(c), 364(d) of the Bankruptcy Code in violation of this Interim DIP Order at any time prior to repayment in

full of the DIP Facility (the "<u>DIP Repayment</u>"), including subsequent to the confirmation of any chapter 11 plan with respect to any or all of the Debtors (if applicable), then all the cash proceeds derived from such credit or debt shall immediately be applied in accordance with this Interim DIP Order.

- 21. <u>Maintenance of DIP Collateral</u>. Until the DIP Repayment has occurred, the Debtors shall (a) insure the DIP Collateral as required under the DIP Credit Agreement; and (b)(i) maintain accurate records of all transfers (including intercompany transactions) within the cash management system so that all postpetition transfers and transactions shall be adequately and promptly documented in, and readily ascertainable from, their books and records, to the same extent maintained by the Debtors before the Petition Date, and (ii) provide reasonable access to such records to the DIP Lender.
- 22. <u>Disposition of DIP Collateral</u>. Except as otherwise provided for in this Interim DIP Order, the Debtors shall not sell, transfer, lease, encumber, or otherwise dispose of any portion of the DIP Collateral other than in the ordinary course of business without the prior written consent of the DIP Lender and no such consent shall be implied from any other action, inaction, or acquiescence by the DIP Lender.
 - 23. Rights and Remedies Upon Event of Default.
- (a) Upon the occurrence of an Event of Default and after the transmission of written notice thereof to counsel for the Debtors, any Committee, and the U.S. Trustee stating an occurrence of an Event of Default, the DIP Lender, in its sole discretion, may: (i) cease making DIP Facility advances to the Debtors; (ii) terminate the Debtors' use of the DIP Collateral (including, without limitation, Cash Collateral), and/or (iii) immediately terminate the DIP

Facility and demand immediate repayment, in cash, of the DIP Facility obligations then outstanding.

(b) Upon the occurrence of an Event of Default, but prior to exercising any additional remedies not listed in paragraph 23(a) but later approved by the Court as set forth in this paragraph 23(b), the DIP Lender shall be required to file a motion with the Court seeking emergency relief on not less than five (5) days' notice, which may be by email, to the Debtors, the U.S. Trustee, and counsel to the Committee (if any) for a further order of the Court, who may fashion any appropriate remedy at the hearing, including modifying the automatic stay in the chapter 11 cases to allow the DIP Lender to exercise customary remedies, including, without limitation, the ability to foreclose upon and/or sell the DIP Collateral and the right to exercise any remedy available under applicable law.

24. <u>Carve-Out</u>.

(a) The "Carve-Out" shall be, collectively, (i) all fees required to be paid to the Clerk of the Court and to the U.S. Trustee pursuant to 28 U.S.C. § 1930(a) *plus* interest at the statutory rate, if any, pursuant to 31 U.S.C. § 3717 (without regard to the Carve-Out Trigger Notice (as defined below)), (ii) reasonable fees and expenses incurred by a trustee and payable under section 726(b) of the Bankruptcy Code in an aggregate amount not to exceed \$100,000 (without regard to the Carve-Out Trigger Notice), (iii) to the extent allowed at any time and only as specifically permitted under the Budget, all unpaid fees and expenses of the professionals retained by the Debtors and any Committee appointed in the Bankruptcy Cases, (A) that are incurred on or prior to the third business day succeeding the date of delivery of the Carve-Out Trigger Notice, or (B) subject to an aggregate cap of \$100,000 for the Debtors' professionals and

a separate aggregate cap of \$25,000 for the Committee's professionals that are incurred after the third business day succeeding the date of delivery of a Carve-Out Trigger Notice (the caps set forth in this clause (B), the "Post Carve-Out Trigger Notice Cap), (iv) to the extent allowed by order of the Court, the fees and expenses payable to Hilco Corporate Finance in connection with the consummation of a sale of substantially all of the Debtors' assets; and (v) to the extent allowed by order of the Court, the completion fee payable by the Debtors to Alvarez & Marsal.

(b) No proceeds of the DIP Facility, the Carve-Out or any Cash Collateral may be used for any fees and expenses, if any, of the Professionals incurred directly or indirectly, in respect of, arising from or relating to: (i) the initiation, joinder, support, or prosecution of any action contesting the indebtedness owed to the Prepetition Bridge Lender or the DIP Lender or the validity of any liens granted to the Prepetition Bridge Lender or the DIP Lender; (ii) preventing, hindering or otherwise delaying, whether directly or indirectly, the exercise by the Prepetition Bridge Lender or the DIP Lender of any of its rights and remedies under the Interim DIP Order, the Final DIP Order, or the Prepetition Secured Note, (iii) the commencement, support, or prosecution of any action or proceeding of any claims, causes of action or defenses against the DIP Lender, the Prepetition Bridge Lender, or any of their respective affiliates, officers, directors, employees, representatives, and agents (including all of their professionals), including, without limitation, any attempt to recover or avoid any claim or interest held by the Prepetition Bridge Lender or the DIP Lender; (iv) any request to borrow money other than pursuant to the terms of the Interim DIP Order, the Final DIP Order, or the documents comprising the DIP Facility; or (v) with respect to the Debtors, any of the Debtors' Professionals, or any of their successors or assigns (including, without limitation, any trustee, responsible officer, examiner, estate administrator or representative or similar person appointed in a case for any of the Debtors under any chapter of the Bankruptcy Code) performing or commencing any investigation or litigation (whether threatened or pending) by the Debtors with respect to any matter released or to be released, waived, or to be waived, or specified as not subject to challenge by the Debtors pursuant to the Interim DIP Order or Final DIP Order.

Contemporaneously with the Interim Advance, the Debtors will transfer (c) cash proceeds from the DIP Facility in an amount equal to the total budgeted weekly fees and expenses for the Debtors' and Committee's (if appointed and as applicable) retained professionals (the "Professional Persons") for the period beginning on the Petition Date through the first four weeks of the Chapter 11 Cases under the approved Budget and after such four-week period on a weekly basis, in each case, excluding any success or other transaction fees of any investment banker or financial advisor of the Debtors or Committee, into one or more escrow accounts at Truist Bank (the "Escrow Agent") for the benefit of the Professional Persons (the "Professional Fee Reserve"). Upon the delivery of a Carve-Out Trigger Notice, the Carve-Out Trigger Notice shall be deemed a request by the Debtors for, and the DIP Lender shall fund DIP Loans under the DIP Facility, in an amount equal to (A) the aggregate amount of budgeted accrued and unpaid Professional Fees incurred before or on the third business day following delivery of a Carve-Out Trigger Notice (to the extent not previously funded) and (B) the Post Carve-Out Trigger Notice Cap (less any amounts already funded in respect of such amounts) (any such amounts actually advanced shall constitute DIP Loans). Amounts funded into the Professional Fee Reserve shall be considered used by the Debtors at such time as they are deposited into the Professional Fee Reserve for distribution to Professional Persons in accordance with orders of the Court. The Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 73 of 171

Debtors are hereby permitted to instruct the Escrow Agent to disburse funds to Professional Persons consistent with orders of the Court (including the Second Amended and Restated General Order 26-2019, Procedures for Complex Chapter 11 Cases). Any amounts remaining in the Professional Fee Reserve after payment of allowed fees and expenses pursuant to the Budget shall be subject to the DIP Liens. The Professional Fee Reserve shall not constitute a cap on the allowed amount of professional fees. Pursuant to the Budget, the Debtors shall also deposit in such escrow account the anticipated fees required to be paid to the Clerk of the Court and to the U.S. Trustee pursuant to 28 U.S.C. §1930(a) for the period from the Petition Date through March 31, 2025.

- exclusively to pay Professional Fees within the Carve-Out as they become allowed and payable pursuant to the Bankruptcy Code, the Bankruptcy Rules, the Local Bankruptcy Rules, and any interim or final orders of the Court; provided that when all Professional Fees and the other obligations that are a part of the Carve-Out have been paid in full (regardless of when such Professional Fees are allowed by the Court), any funds remaining in the Professional Fee Reserve shall revert to the DIP Lender.
- (e) Notwithstanding anything to the contrary in the DIP Loan Documents, this Interim DIP Order, or any other Court order, the Professional Fee Reserve and the amounts on deposit in the Professional Fee Reserve shall be available and used only to satisfy Professional Fees benefitting from the Carve-Out, and the other obligations that are a part of the Carve-Out. The failure of the Professional Fee Reserve to satisfy Professional Fees in full shall not affect the priority of the Carve-Out; provided, that, to the extent that the Professional Fee Reserve is actually funded, the Carve-Out shall be reduced by such funded amount dollar-for-dollar. In no way shall

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 74 of 171

the Carve-Out, Professional Fee Reserve, or the approved Budget or any of the foregoing be construed as a cap or limitation on the amount of the allowed Debtor Professional Fees or Statutory Fees due and payable by the Debtors or that may be allowed by the Court at any time (whether by interim order, final order, or otherwise).

- (f) "<u>Carve-Out Trigger Notice</u>" shall mean a written notice delivered by the DIP Lender to the Debtors' lead counsel, the U.S. Trustee, and lead counsel to the Committee, which notice may only be delivered following the occurrence and during the continuation of an Event of Default under the DIP Facility.
- Lender nor the Prepetition Bridge Lender shall be responsible for the payment or reimbursement of any fees or disbursements of any Professional incurred in connection with the Chapter 11 Cases or any Successor Cases. Nothing in this Interim DIP Order or otherwise shall be construed to obligate the DIP Lender or the Prepetition Bridge Lender, in any way, to pay compensation to, or to reimburse expenses of, any Professional or to guarantee that the Debtors have sufficient funds to pay such compensation or reimbursement. As used herein, the term "Professional" means any professional retained by the Debtors or retained by any Committee appointed in the Bankruptcy Cases.
- 25. Good Faith Under Section 364(e) of the Bankruptcy Code; No Modification or Stay of this Interim DIP Order. Based on the findings set forth in this Interim DIP Order and the record made during the Interim Hearing, and in accordance with section 364(e) of the Bankruptcy Code, in the event any or all of the provisions of this Interim DIP Order are hereafter modified, amended, waived, or vacated by a subsequent order of this Court or any other court of competent

jurisdiction, each of the DIP Lender and the Prepetition Bridge Lender is entitled to the protections provided in section 364(e) of the Bankruptcy Code. Any such modification, amendment, waiver or vacatur shall not affect the validity and enforceability of any advances previously made, including advances made hereunder, or any lien, claim, or priority authorized or created hereby, unless such authorization and the incurring of such debt, or the granting of such priority or lien, is stayed pending appeal.

26. Payment of Fees and Expenses. The Debtors shall pay all reasonable and documented prepetition and postpetition fees and out-of-pocket expenses of the DIP Lender in connection with the DIP Facility and the Chapter 11 Cases, as provided in this Interim DIP Order. Any time that professionals of the DIP Lender, including, without limitation, Porter Hedges LLP, as counsel to the DIP Lender, seek payment of fees and expenses from the Debtors, such professional shall provide summary copies of its fee and expense statements or invoices (which shall not be required to contain time entries and which may be redacted or modified to the extent necessary to delete any information subject to the attorney-client privilege, any information constituting attorney work product, or any other confidential information, and the provision of such invoices shall not constitute any waiver of the attorney-client privilege or of any benefits of the attorney work product doctrine) with: (i) a summary of the work performed during the relevant compensation period; (ii) the name of, hourly rate (if applicable) of, and number of hours worked by each professional and paraprofessional who worked on the matter during the relevant compensation period; and (iii) the total fee amount being requested by electronic mail to the U.S. Trustee and counsel to the Committee (if appointed) contemporaneously with the delivery of such fee and expense statements to the Debtors. The Debtors, any Committee, or the U.S. Trustee may dispute the payment of any portion of such invoiced fees and expenses (the "Disputed Invoiced <u>Fees</u>") if a Debtor, any Committee that may be appointed in these Chapter 11 Cases, or the U.S. Trustee notifies the submitting party in writing, within ten (10) days of the receipt of such fee and expense statement or invoice, setting forth the specific objections to the Disputed Invoiced Fees (to be followed by the filing with the Court, if necessary, of an objection, with at least ten (10) days' prior written notice to the submitting party of any hearing on such objection). The Debtors shall promptly pay in full all such invoiced fees and expenses other than the Disputed Invoiced Fees. Notwithstanding the foregoing, the Debtors shall pay on or about the date of entry of the Interim DIP Order all reasonable and documented fees, costs, and out-of-pocket expenses of the DIP Lender incurred on or prior to such date without the need for any professional engaged by the DIP Lender to first deliver a copy of its invoice as provided for herein (other than to the Debtors). No attorney or advisor to the DIP Lender shall be required to file an application seeking compensation for services or reimbursement of expenses with the Court. Any and all fees, costs, and expenses paid prior to the Petition Date by any of the Debtors to the DIP Lender (including the fees and expenses of Porter Hedges LLP, as counsel to the DIP Lender) shall be approved in full in the Interim DIP Order and the Final DIP Order. The DIP Lender shall have the right to charge the DIP Facility for any such fees and costs (whether such costs were incurred prior to the Petition Date or after the Petition Date). To the extent the DIP Lender charges the DIP Facility for such fees and costs, they will become part of the DIP Facility.

27. <u>Proofs of Claim</u>. The DIP Lender and the Prepetition Bridge Lender will not be required to file proofs of claim in any of the Chapter 11 Cases or Successor Cases for any claim allowed herein, including any claims arising under the Prepetition Secured Note. The Debtors'

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 77 of 171

Stipulations shall be deemed to constitute a timely filed proof of claim for the DIP Lender and the Prepetition Bridge Lender upon approval of this Interim DIP Order, and the DIP Lender and the Prepetition Bridge Lender shall be treated under section 502(a) of the Bankruptcy Code as if they filed a proof of claim. Any order entered by the Court in relation to the establishment of a bar date for any claim (including without limitation administrative claims) in any of the Chapter 11 Cases or any Successor Cases shall not apply to (i) the DIP Lender, or (ii) the Prepetition Bridge Lender with respect to any claims arising under the Prepetition Secured Note.

- 28. <u>Effect of Stipulations and Order</u>. The Debtors' Stipulations and the releases set forth in paragraph 32 of this Interim DIP Order (the "<u>Releases</u>"), shall be binding on the Debtors in all circumstances and for all purposes, including with respect to any Prepetition Secured Note Obligations.
- 29. <u>No Third-Party Rights</u>. Except as explicitly provided for herein, this Interim DIP Order does not create any rights for the benefit of any third party, creditor, equity holder, or any direct, indirect, or incidental beneficiary.
- 30. Section 506(c) Claims. Subject to and upon entry of the Final DIP Order, except to the extent of the Carve-Out, no costs or expenses of administration that have been or may be incurred in the Chapter 11 Cases at any time shall be charged against the DIP Lender, the DIP Collateral, the Prepetition Bridge Lender, or the Prepetition Collateral pursuant to sections 105 or 506(c) of the Bankruptcy Code, or otherwise, without the prior written consent of the DIP Lender or the Prepetition Bridge Lender, as may be applicable, and no such consent shall be implied from any action, inaction, or acquiescence by any party.

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 78 of 171

- 31. <u>No Marshaling/Applications of Proceeds</u>. Subject to and upon entry of the Final DIP Order, in no event shall the DIP Lender be subject to the equitable doctrine of "marshaling" or any other similar doctrine with respect to any of the DIP Collateral or the Prepetition Collateral, as applicable.
- 32. Releases. In exchange for good and valuable consideration, the adequacy of which is hereby confirmed, each of the Debtors, (in their own right, on behalf of their estates) (collectively, the "Releasing Parties") hereby unconditionally and irrevocably releases, acquits, absolves, forever discharges and covenants not to sue the DIP Lender, the Prepetition Bridge Lender, and each such entities' current and former affiliates, and each such entity's current and former directors, officers, managers and equityholders (regardless of whether such interests are held directly or indirectly), predecessors, successors and assigns, and direct and indirect subsidiaries, and each of such entity's current and former officers, members, managers, directors, equityholders (regardless of whether such interests are held directly or indirectly), principals, members, employees, agents, independent contractors, representatives, managed accounts or funds, management companies, fund advisors, investment advisors, financial advisors, and partners (including both general and limited partners) (the "Released Parties") and their respective property and assets from any and all acts and omissions of the Released Parties, and from any and all claims, interests, causes of action, avoidance actions, counterclaims, defenses, setoffs, demands, controversies, suits, judgments, costs, debts, sums of money, accounts, reckonings, bonds, bills, damages, obligations, objections, legal proceedings, equitable proceedings, executions of any nature, type, or description and liabilities whatsoever (including any derivative claims asserted or assertable on behalf of the Debtors, their estates, or such entities' successors or

assigns, whether individually or collectively), which the Releasing Parties now have, may claim to have or may come to have against the Released Parties through the date of the Final DIP Order, at law or in equity, by statute or common law, in contract or in tort, including, without limitation, (a) any so-called "lender liability" or equitable subordination claims or defenses, (b) any and all "claims" (as defined in the Bankruptcy Code) and causes of action arising under the Bankruptcy Code and (c) any and all offsets, defenses, claims, counterclaims, set off rights, objections, challenges, causes of action, and/or choses in action of any kind or nature whatsoever, whether liquidated or unliquidated, fixed or contingent, known or unknown, suspected or unsuspected, disputed or undisputed, whether arising at law or in equity, including any recharacterization, recoupment, subordination, disallowance, avoidance, challenge, or other claim or cause of action arising under or pursuant to section 105, chapter 5, or section 724(a) of the Bankruptcy Code or under other similar provisions of applicable state, federal, or foreign laws, including without limitation, any right to assert any disgorgement, recovery, and further waives and releases any defense, right of counterclaim, right of setoff, or deduction on the payment of the Prepetition Secured Note Obligations, but excluding obligations of the DIP Lender under the DIP Facility arising after the date of the Final DIP Order. This paragraph is in addition to and shall not in any way limit any other release, covenant not to sue, or waiver by the Releasing Parties in favor of the Released Parties.

33. <u>Indemnification</u>. The Debtors shall indemnify and hold the DIP Lender and each such entities' current and former affiliates, and each such entity's current and former directors, officers, managers and equityholders (regardless of whether such interests are held directly or indirectly), predecessors, successors and assigns, and direct and indirect subsidiaries, and each of

such entity's current and former officers, members, managers, directors, equityholders (regardless of whether such interests are held directly or indirectly), principals, members, employees, agents, independent contractors, representatives, managed accounts or funds, management companies, fund advisors, investment advisors, financial advisors, and partners (including both general and limited partners) (each an "Indemnified Party") harmless from and against any and all claims, damages, losses, liabilities and expenses (including, without limitation, all fees and disbursements of attorneys and other professionals) to which any Indemnified Party may become liable or which may be incurred by or asserted against any Indemnified Party, in each case in connection with or arising out of or by reason of any investigation, litigation or proceeding arising out of or relating to or in connection with the DIP Facility, any obligation, or any act, event or transaction related or attendant thereto or any use or intended use of the proceeds of the DIP Facility, except to the extent the same is found in a final, nonappealable judgment by a court of competent jurisdiction to have resulted from such Indemnified Party's own gross negligence or willful misconduct. The Debtors shall file a notice with the Court if any Indemnified Party asserts a claim against the Debtors under this paragraph.

24. <u>Limits on Lender Liability</u>. Nothing in this Interim DIP Order, any of the DIP Loan Documents, any of the Prepetition Secured Note, or any other documents related thereto, shall in any way be construed or interpreted to impose or allow the imposition upon the DIP Lender, or the Prepetition Bridge Lender of any liability for any claims arising from any activities by the Debtors in the operation of their businesses or in connection with the administration of these Chapter 11 Cases or any Successor Cases. The DIP Lender and the Prepetition Bridge Lender shall not, solely by reason of having made loans under the DIP Facility or authorizing the use of

Cash Collateral, be deemed in control of the operations of the Debtors or to be acting as a "responsible person" or "owner or operator" with respect to the operation or management of the Debtors (as such terms, or any similar terms, are used in the United States Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. §§ 9601 *et seq.*, as amended, or any similar federal or state statute). Nothing in this Interim DIP Order shall in any way be construed or interpreted to impose or allow the imposition upon the DIP Lender or the Prepetition Bridge Lender of any liability for any claims arising from the prepetition or postpetition activities of any of the Debtors.

- 35. <u>Insurance Proceeds and Policies</u>. Upon entry of this Interim DIP Order and to the fullest extent provided by applicable law, the DIP Lender shall be deemed to be, without any further action or notice, named as additional insured and loss payee on each insurance policy maintained by the Debtors that in any way relates to the DIP Collateral or any collateral subject to Adequate Protection Liens; *provided* that the rights of a landlord under any non-residential real property lease to any such insurance proceeds are hereby preserved.
- 36. <u>Joint and Several Liability</u>. Nothing in this Interim DIP Order shall be construed to constitute a substantive consolidation of any of the Debtors' estates, it being understood, however, that the Debtors shall be jointly and severally liable for the obligations hereunder and all DIP Obligations. All unpaid principal, interest, fees, costs, and expenses accruing under the DIP Facility shall be due and payable in full on the Maturity Date to the extent not due earlier, and if such amounts are not paid in full in cash, interest, fees, costs, and expenses in respect of the DIP Facility shall continue to accrue until paid in full.

37. [Reserved].

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 82 of 171

- Rights Preserved. Notwithstanding anything herein to the contrary, the entry of this Interim DIP Order is without prejudice to, and does not constitute a waiver of, expressly or implicitly: (a) the DIP Lender's and the Prepetition Bridge Lender's rights to seek any other or supplemental relief; (b) any of the rights of any of the DIP Lender and/or the Prepetition Bridge Lender under the Bankruptcy Code or applicable non-bankruptcy law, including, without limitation, the right to request modification of the automatic stay imposed by section 362 of the Bankruptcy Code, request dismissal of any of the Chapter 11 Cases or Successor Cases, conversion of any of the Chapter 11 Cases to cases under chapter 7, or appointment of a chapter 11 trustee or examiner with expanded powers, to enforce any guarantee against a non-Debtor; or (iii) propose, subject to the provisions of section 1121 of the Bankruptcy Code, a chapter 11 plan or plans; or (c) any other rights, claims, or privileges (whether legal, equitable, or otherwise) of the DIP Lender or the Prepetition Bridge Lender.
- 39. <u>No Waiver by Failure to Seek Relief.</u> The failure of the DIP Lender or the Prepetition Bridge Lender to seek relief or otherwise exercise their rights and remedies under this Interim DIP Order, the Prepetition Secured Note, or applicable law, as the case may be, shall not constitute a waiver of any of the rights hereunder, thereunder, or otherwise of the DIP Lender or the Prepetition Bridge Lender.
- 40. <u>Binding Effect of Interim DIP Order</u>. The provisions of this Interim DIP Order shall be binding upon and inure to the benefit of the Debtors, the DIP Lender, the Prepetition Bridge Lender, any Committee appointed in these Chapter 11 Cases, all other creditors of any of the Debtors and all other parties in interest and, in each case, their respective successors and assigns (including any chapter 7 or chapter 11 trustee hereinafter appointed or elected for the

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 83 of 171

estate of the Debtors, an examiner appointed pursuant to section 1104 of the Bankruptcy Code or any other fiduciary appointed as a legal representative of any of the Debtors or with respect to the property of the estate of any of the Debtors). This Interim DIP Order shall bind any trustee hereafter appointed for the estate of any of the Debtors, whether in these Chapter 11 Cases or in the event of the conversion of any of the Chapter 11 Cases, any Successor Cases, or upon dismissal of any Chapter 11 Case or Successor Case, to a liquidation under chapter 7 of the Bankruptcy Code. Such binding effect is an integral part of this Interim DIP Order.

- 41. <u>No Modification of Interim DIP Order</u>. Until the DIP Repayment has occurred, the Debtors shall be prohibited from seeking or consenting to, directly or indirectly, any modification, stay, vacatur, waiver, or amendment to this Interim DIP Order or any provision hereof without the prior written consent of the DIP Lender, and no such consent shall be implied by any action or inaction of the DIP Lender.
- 42. <u>Modifications to DIP Loan Documents.</u> Upon entry of the Final DIP Order, the Debtors and the DIP Lender are authorized to implement, in accordance with the terms of the DIP Loan Documents, any non-material modifications or amendments of the DIP Loan Documents without further order of this Court; *provided, however*, that notice of any non-material modification or amendment to the DIP Loan Documents shall be provided to lead counsel to the Committee (if appointed) and the U.S. Trustee, each of whom shall have five (5) business days from the date of such notice within which to object in writing (email to suffice) to such modification or amendment. If any such party timely objects to any non-material modification or amendment to the DIP Loan Documents, such modification or amendment shall only be permitted pursuant to an order of this Court. For the avoidance of doubt, the extension of a Milestone shall

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 84 of 171

not constitute a material amendment, modification, waiver, or supplement to the DIP Loan Documents.

- 43. <u>Interim DIP Order Controls</u>. In the event of any inconsistency between the terms and conditions of any (as yet finalized) DIP Loan Documents and this Interim DIP Order, the provisions of this Interim DIP Order shall control.
- 44. <u>Discharge</u>. Subject to the terms and conditions of this Interim DIP Order, the DIP Obligations shall not be discharged by the entry of an order confirming any plan of reorganization in any of the Chapter 11 Cases, notwithstanding the provisions of section 1141(d) of the Bankruptcy Code, unless the DIP Repayment has occurred on or before the effective date of such plan of reorganization or the DIP Lender has otherwise agreed in writing.
- 45. <u>Survival</u>. The provisions of this Interim DIP Order, any actions taken pursuant hereto or thereto, and all of the protections, rights, remedies, liens, priorities, privileges, and benefits granted to the DIP Lender shall survive, and shall not be modified, impaired, or discharged by, the entry of any order confirming any plan of reorganization or liquidation in any of the Chapter 11 Cases, converting any of the Chapter 11 Cases to a chapter 7 case, dismissing any of the Chapter 11 Cases or any Successor Cases, withdrawing of the reference of any of these Chapter 11 Cases, any Successor Cases, or providing for abstention from handling or retaining of jurisdiction of any of these Chapter 11 Cases in this Court, or terminating the joint administration of these Chapter 11 Cases or by any other act or omission. The terms and provisions of this Interim DIP Order shall continue in the Chapter 11 Cases, in any Successor Cases, or following the dismissal of the Chapter 11 Cases or any Successor Cases, notwithstanding the entry of any such order, and such protections, rights, remedies, liens, priorities, privileges, and benefits shall

continue in full force and effect in these proceedings and after dismissal of any thereof, and shall maintain their respective priorities as provided by this Interim DIP Order, and to the maximum extent permitted by law, until all of the DIP Obligations are indefeasibly paid in full in cash and discharged.

- 46. <u>Headings</u>. Section headings used herein are for convenience only and are not to affect the construction of or to be taken into consideration in interpreting this Interim DIP Order.
- 47. <u>Retention of Jurisdiction</u>. The Court retains exclusive jurisdiction to resolve any dispute arising from or related to the interpretation or enforcement of the DIP Facility and/or this Interim DIP Order.
- 48. <u>Final Hearing</u>. The Final Hearing is scheduled for [______], 2025 at [__] a.m./p.m. (prevailing Central Time) before this Court. Any objections or responses to entry of the Final DIP Order on the Motion shall be filed on or before 4:00 p.m. (prevailing Central time), on [_____], 2025. In the event no objections to entry of the Final DIP Order on the Motion are timely received, the Court may enter such Final DIP Order without need for the Final Hearing.

[END OF ORDER]

Exhibit 1

Budget

Exhibit 2

DIP Credit Agreement

EXHIBIT 1

DIP Term Loan Agreement

EXECUTION VERSION

SECURED SUPER-PRIORITY DEBTOR-IN-POSSESSION CREDIT AGREEMENT

by and between

OTB ACQUISITION LLC, Debtor and Debtor-In-Possession

as Borrower,

and

OTB LENDER, LLC

as DIP Lender

Dated as of March 4, 2025

TABLE OF CONTENTS

1.	DEFI	NITIONS AND CONSTRUCTION.	2
	1.1.	Definitions	2
	1.2.	Accounting Terms	3
	1.3.	UCC	3
	1.4.	Construction	3
	1.5.	Schedules and Exhibits	3
2.	LOAN	N AND TERMS OF PAYMENT	3
	2.1.	Advances; Termination of Loan Commitment.	3
	2.2.	Borrowing Procedures.	4
	2.3.	Payments.	5
	2.4.	Interest Rate: Rates, Payments, and Calculations	6
	2.5.	Intent to Limit Charges to Maximum Lawful Rate	7
	2.6.	Borrower Designated Account	8
	2.7.	Facility Fee	8
	2.8.	Security Interest; Superpriority Claim and Liens; Collateral	8
	2.9.	Continuing and Unconditional Guaranty	10
	2.10.	Joint and Several	12
	2.11.	Waiver of any Priming Rights	12
3.	CONI	DITIONS PRECEDENT; TERMINATION OF AGREEMENT	12
	3.1.	Conditions Precedent to Effectiveness of this Agreement	12
	3.2.	Conditions Precedent to all Advances	13
	3.3.	Term	14
	3.4.	Effect of Termination	15
4.	REPRESENTATIONS AND WARRANTIES		
	4.1.	Due Organization and Qualification	15
	4.2.	Due Authorization; Enforceability	15
	4.3.	No Conflict	15
	4.4.	No Encumbrances	16
	4.5.	(Reserved)	16
	4.6.	(Reserved)	16

	4.7.	Location of Offices	16
	4.8.	Inventory Records	16
	4.9.	State of Incorporation; Location of Chief Executive Office; Organizational Identification Number; Commercial Tort Claims; Books and Records Pertaining to Accounts	16
	4.10.	Capitalization; Subsidiaries.	17
	4.11.	(Reserved)	17
	4.12.	(Reserved)	17
	4.13.	(Reserved)	17
	4.14.	(Reserved)	17
	4.15.	Intellectual Property	17
	4.16.	Leased and Owned Real Property	17
	4.17.	Deposit Accounts and Securities Accounts	17
	4.18.	Complete Disclosure	17
	4.19.	(Reserved)	18
	4.20.	(Reserved)	18
	4.21.	(Reserved)	18
	4.22.	(Reserved)	18
	4.23.	Labor Matters	18
	4.24.	Security Documents; Priority of Liens	18
	4.25.	Insurance	18
	4.26.	Compliance with Statutes, etc	18
	4.27.	Patriot Act	18
5.	AFFII	RMATIVE COVENANTS	19
	5.1.	Reports and Information	19
	5.2.	Accounting System	20
	5.3.	Notices of Material Events	20
	5.4.	Material Contracts	21
	5.5.	Inspection	21
	5.6.	Maintenance of Properties and Books and Records	21
	5.7.	Payment of Obligations; Taxes	21
	5.8.	Insurance	22
	5.9.	Location of Offices	22

	5.10.	Compliance with Laws	. 22
	5.11.	Leases	. 23
	5.12.	Existence	. 23
	5.13.	Use of Proceeds	. 23
	5.14.	Environmental	. 24
	5.15.	Disclosure Updates	. 24
	5.16.	Control of Collateral and Control Agreements	. 24
	5.17.	Further Assurances	. 24
	5.18.	Bankruptcy Related Affirmative Covenants; Milestones	. 25
	5.19.	Approved Budget; Variance Reports	. 25
	5.20.	Professional Fee Reserve	. 26
	5.21.	Post-Closing Requirements	. 26
5.	NEGA	ATIVE COVENANTS.	. 26
	6.1.	Indebtedness	. 26
	6.2.	Liens	. 26
	6.3.	Restrictions on Fundamental Changes	. 26
	6.4.	Disposal of Assets	. 27
	6.5.	Change Name	. 27
	6.6.	Nature of Business	. 27
	6.7.	Restricted Payments; Certain Payments of Indebtedness	. 27
	6.8.	Change of Control	. 27
	6.9.	Consignments	. 27
	6.10.	Accounting Methods	. 27
	6.11.	Investments	. 27
	6.12.	Transactions with Affiliates	. 28
	6.13.	Inventory and Equipment with Bailees	. 28
	6.14.	Compliance with the Approved Budget	. 28
	6.15.	Restrictive Agreements	. 28
	6.16.	Amendment of Material Documents	. 28
	6.17.	Bankruptcy Related Negative Covenants	. 28
7.	EVEN	TS OF DEFAULT	. 29
3.	RIGH'	TS AND REMEDIES ON DEFAULT.	. 34
	8.1.	Rights and Remedies	. 34

	8.2.	Power of Attorney	36
	8.3.	Remedies Cumulative	37
9.	APPL	LICATION OF PROCEEDS ON DEFAULT	37
10.	TAXI	ES AND EXPENSES	37
	10.1.	Taxes	37
	10.2.	Reserved	38
	10.3.	Payment of other Amounts	38
11.	WAIV	VERS; INDEMNIFICATION	38
	11.1.	Demand; Protest, etc	38
	11.2.	DIP Lender's Liability for Collateral	38
	11.3.	Indemnification	39
12.	NOTI	ICES	40
13.	CHO	ICE OF LAW AND VENUE; JURY TRIAL WAIVER	41
14.	ASSI	GNMENTS; SUCCESSORS	42
	14.1.	Successors and Assigns Generally	42
	14.2.	Assignments by DIP Lender	42
	14.3.	Participations	42
15.	AME	NDMENTS; WAIVERS	42
	15.1.	Amendments and Waivers	42
	15.2.	No Waivers; Cumulative Remedies	42
16.	GENI	ERAL PROVISIONS.	43
	16.1.	Section Headings	43
	16.2.	Interpretation	43
	16.3.	Severability of Provisions	43
	16.4.	Counterparts; Electronic Execution	43
	16.5.	Revival and Reinstatement of Obligations	43
	16.6.	Reserved	43
	16.7.	Integration	43
	16.8.	Marshaling	43
	16.9.	Press Releases and Related Matters	44
	16.10	. USA PATRIOT Act Notice	44
	16.11	. No Advisory or Fiduciary Responsibility	44
	16.12	. Relationship with DIP Orders	45

EXHIBITS AND SCHEDULES

Schedule 1.1	Definitions
Schedule 4.3	Consents

Schedule 4.7 Locations of Offices
Schedule 4.9(a) States of Organization
Schedule 4.9(b) Chief Executive Offices

Schedule 4.9(c) Organizational Identification Numbers

Schedule 4.10(a) Capitalization of Borrower

Schedule 4.10(b) Capitalization of Borrower's Subsidiaries

Schedule 4.16 Leases

Schedule 4.17 Collateral Accounts

Schedule 5.21 Post-Closing Requirements
Schedule 6.12 Transactions with Affiliates

Exhibit A Reserved

Exhibit B Form of Notice of Borrowing

SECURED SUPER-PRIORITY DEBTOR-IN-POSSESSION CREDIT AGREEMENT

THIS SECURED SUPER-PRIORITY DEBTOR IN POSSESSION CREDIT AGREEMENT (this "Agreement"), dated as of March 4, 2025, between OTB ACQUISITION LLC, a Delaware limited liability company, as debtor and debtor in possession under Chapter 11 of the Bankruptcy Code (as defined below) ("Borrower"), OTB HOLDING LLC, a Delaware limited liability company ("Holding"), OTB ACQUISITION OF NEW JERSEY LLC, a New Jersey limited liability company ("OTBNJ"), OTB ACQUISITION OF HOWARD COUNTY LLC, a Maryland limited liability company ("OTB Howard"), MT. LAUREL RESTAURANT OPERATIONS LLC, a New Jersey limited liability company ("OTB Operations"), OTB ACQUISITION OF KANSAS LLC, a Kansas limited liability company ("OTB Kansas"), and OTB ACQUISITION OF BALTIMORE COUNTY, LLC, a Maryland limited liability company ("OTB Baltimore" and OTB Baltimore together with Holding, OTBNJ, OTB Howard, OTB Operations and OTB Kansas, collectively, the "Guarantors" and the Guarantors herein collectively with the Borrower, the "Debtors") and OTB LENDER, LLC, a Delaware limited liability company, as lender ("DIP Lender").

RECITALS:

- A. DIP Lender extended a term loan in an aggregate principal amount of \$4,000,000 (the "<u>Prepetition Bridge Loan Amount</u>") to the Debtors pursuant to that certain Secured Promissory Note, dated as of February 14, 2025 (as amended or otherwise modified prior to the Petition Date, the "<u>Prepetition Secured Note</u>" and the loans thereunder, the "<u>Prepetition Bridge Loans</u>").
- B. Reference is made to (i) that certain Credit Agreement, dated as of September 9, 2022 (as amended or otherwise modified prior to the Petition Date, the "CrossFirst Credit Agreement"), by and among Borrower, Holding and CrossFirst Bank, (ii) the obligations owed to CrossFirst Bank and the other lenders named in the CrossFirst Credit Agreement, (iii) the Liens against certain assets of Debtors (which such Liens were perfected prior to the Liens granted in connection with the Prepetition Secured Note) held by CrossFirst Bank (the "CrossFirst Liens").
- C. Concurrently with entering into the Prepetition Secured Note, the DIP Lender entered into that certain Subordination and Intercreditor Agreement, dated as of February 14, 2025 (as amended or otherwise modified, the "Subordination Agreement") pursuant to which CrossFirst Bank has consented to the priming of the CrossFirst Liens and the subordination of the obligations arising under the CrossFirst Credit Agreement.
- D. On March 4, 2025 (the "<u>Petition Date</u>"), each Debtor filed a voluntary petition for relief (each, a "<u>Chapter 11 Case</u>" and, collectively, the "<u>Chapter 11 Cases</u>") under Chapter 11 of the Bankruptcy Code with the United States Bankruptcy Court for the Northern District of Georgia (the "Bankruptcy Court").
- E. Each Debtor is continuing to operate its business and manage its properties as a debtor in possession under Sections 1107 and 1108 of the Bankruptcy Code.

- F. Borrower has requested that DIP Lender provide a secured super-priority term loan credit facility providing for an aggregate principal amount not to exceed the sum of \$14,000,000 plus all interest, fees and expenses that will be capitalized as part of the Principal Amount under this Agreement to Borrower on the terms and conditions set forth herein. The Loans hereunder shall consist of:
 - (a) up to \$10,000,000 of new money loans, (i) with the portion that is sufficient to fund the estimated professional fees for the entire Availability Period plus all other amounts required to fund the Approved Budget for the first twenty-eight (28) days after the Petition Date (but in any event not in excess of \$7,500,000) is to be available to Borrower upon entry of the Interim DIP Order (such amount that is actually extended on such date, the "Interim Advance") and (ii) the remainder of which is to be available to Borrower upon entry of the Final DIP Order;
 - (b) An Advance in an amount equal to the Prepetition Bridge Loan Amount, together with all accrued and unpaid interest and fees under the Prepetition Secured Note, shall be deemed to be made hereunder concurrently with the Interim Advance; and
 - (c) An amount equal to the Facility Fee shall be capitalized and added to the Principal Amount outstanding hereunder concurrently with the Interim Advance.
- G. Each Guarantor will materially benefit from the extensions of credit made hereunder and the DIP Lender will not make the extensions of credit hereunder without the guarantee of all of Borrower's obligations hereunder by each Guarantor.
- H. DIP Lender is willing to make available to Borrower such post-petition loans and other extensions of credit upon the terms and subject to the conditions set forth herein.
- I. Such post-petition loans and other extensions of credit will be used to repay the Prepetition Bridge Loans of Debtors, together with all accrued and unpaid interest and fees under the Prepetition Secured Note, to pay DIP Lender's related fees and expenses associated with the negotiation, execution and delivery of this Agreement and for working capital and other general corporate purposes in accordance with Section 6.14 to the extent permitted hereunder.
- J. Each Debtor has agreed to secure its obligations to DIP Lender hereunder with, *inter alia*, security interests in, and liens on, all of its property and assets, whether real or personal, tangible or intangible, now existing or hereafter acquired or arising, all as more fully provided herein and in the other Loan Documents.

NOW, THEREFORE, in consideration of the premises and the covenants and agreements contained herein, the parties hereto hereby agree as follows:

1. DEFINITIONS AND CONSTRUCTION.

1.1. <u>Definitions</u>. Capitalized terms used in this Agreement shall have the meanings specified therefor on <u>Schedule 1.1</u>.

- 1.2. <u>Accounting Terms</u>. All accounting terms not specifically defined herein shall be construed in accordance with GAAP. When used herein, the term "financial statements" shall include the notes and schedules thereto.
- 1.3. <u>UCC</u>. Any terms used in this Agreement that are defined in the UCC shall be construed and defined as set forth in the UCC unless otherwise defined herein; <u>provided</u>, <u>however</u>, that to the extent that the UCC is used to define any term herein and such term is defined differently in different Articles of the UCC, the definition of such term contained in Article 9 of the UCC shall govern.
- 1.4. **Construction**. Unless the context of this Agreement or any other Loan Document clearly requires otherwise, references to the plural include the singular, references to the singular include the plural, the terms "includes" and "including" are not limiting, and the term "or" has, except where otherwise indicated, the inclusive meaning represented by the phrase "and/or." The words "hereof," "herein," "hereby," "hereunder," and similar terms in this Agreement or any other Loan Document refer to this Agreement or such other Loan Document, as the case may be, as a whole and not to any particular provision of this Agreement or such other Loan Document, as the case may be. Section, subsection, clause, schedule, and exhibit references herein are to this Agreement unless otherwise specified. Any reference in this Agreement or in the other Loan Documents to any agreement, instrument, or document shall include all alterations, amendments, changes, extensions, modifications, renewals, replacements, substitutions, joinders, and supplements, thereto and thereof, as applicable (subject to any restrictions on such alterations, amendments, changes, extensions, modifications, renewals, replacements, substitutions, joinders, and supplements set forth herein). Any reference herein to the satisfaction or repayment in full of the Obligations shall mean the repayment in full in cash of all Obligations other than contingent indemnification Obligations. Any reference herein to any Person shall be construed to include such Person's successors and assigns. Any requirement of a writing contained herein or in the other Loan Documents shall be satisfied by the transmission of a Record and any Record transmitted shall constitute a representation and warranty as to the accuracy and completeness of the information contained therein.
- 1.5. <u>Schedules and Exhibits</u>. All of the schedules and exhibits attached to this Agreement shall be deemed incorporated herein by reference.

2. LOAN AND TERMS OF PAYMENT.

2.1. Advances; Termination of Loan Commitment.

- (a) <u>Term Loans.</u> During the Availability Period, subject to the terms and conditions of this Agreement and the Budget, DIP Lender agrees to make the Interim Advance and one or more additional Advances in Dollars on a non-revolving basis to Borrower from time to time in an aggregate amount not to exceed the lesser of the Available Commitment Amount and the Loan Commitment Amount.
- (b) <u>Deemed Advances</u>. (i) An Advance in an aggregate principal amount equal to the Prepetition Bridge Loan Amount shall be deemed to have been made hereunder concurrently

with the Interim Advance (the "<u>Prepetition Bridge Loan Roll-Up</u>"). (ii) An amount equal to the Facility Fee shall be added to the Principal Amount as further described in <u>Section 2.7</u>.

- (c) <u>General Terms</u>. Amounts borrowed pursuant to this <u>Section 2.1</u> may be prepaid or repaid but may not be reborrowed. Upon each extension (in cash, by net funding or by deemed issuance) of an Advance hereunder, the Available Commitment Amount and the Loan Commitment shall be reduced by an amount equal to the principal amount of such Advance. The Advances shall, from the Interim DIP Order Date until the Effective Date, be subject to all conditions set forth in the Interim DIP Order and, after the Effective Date, be subject and conditioned upon satisfaction of the DIP Orders and the conditions in <u>Section 3</u>.
- (d) <u>Limitations on Advances</u>. Notwithstanding anything to the contrary contained in this <u>Section 2</u>, DIP Lender will not be required and shall have no obligation to make any Advance requested hereunder if (i) the amount of such requested Advance exceeds the Available Commitment Amount in effect on such date, (ii) the amount of such requested Advance is not supported by or consistent with the Approved Budget for the applicable Budget Period or (iii) the conditions in <u>Section 3</u> have not been satisfied. The portion of the Loan Commitment Amount equal to the Prepetition Bridge Loan Amount shall only be available hereunder in the form of a deemed Advance and in no event shall be available as a new money Advance.
- (e) <u>Termination</u>. The Loan Commitment shall automatically terminate on the Termination Date.

2.2. **Borrowing Procedures**.

- Borrower may request by an irrevocable written request substantially in the form attached hereto as Exhibit B or such other form approved by DIP Lender (each, a "Notice of Borrowing") by the chief restructuring officer of the Debtors delivered to the DIP Lender. Such Notice of Borrowing must be received by the DIP Lender no later than 12:00 noon (Central time) at least one (1) Business Day prior to the proposed funding date, which such funding date shall be a Business Day. Each Notice of Borrowing shall specify (i) the amount of the proposed Advance, (ii) the requested funding date, and (iii) Budget Period covered by such Notice of Borrowing and a reconciliation of the intended use of proceeds of such Advance against the items budgeted for during such Budget Period. The Prepetition Bridge Loan Roll-Up shall be made automatically on the applicable date specified in Section 2.1 without delivery of any Notice of Borrowing.
- Borrower in immediately available funds, to the Borrower Designated Account, not later than 1:00 p.m. (Eastern time) on the funding date applicable thereto; provided, however, that each Advance shall be net of all amounts deemed to have been advanced hereunder and other amounts reserved pursuant to Section 10.3(a); provided, however, that DIP Lender shall not have the obligation to make any Advance if DIP Lender shall have knowledge that (i) one or more of the applicable conditions precedent set forth in Section 3 will not be satisfied on the requested funding date for such Advance, unless such condition has been waived inwriting by the DIP Lender or (ii) the amount of the requested Borrowing would exceed the total projected expenditures of Borrower set forth in the Approved Budget for the applicable Budget Period.

(c) **Notation**. The Advances made by DIP Lender shall be evidenced by one or more accounts or records maintained by DIP Lender in the ordinary course of business. The accounts or records maintained by the DIP Lender shall be conclusive absent manifest error of the amount of the Advances made by the DIP Lender to Borrower and the interest and payments thereon. Any failure to so record or any error in doing so shall not, however, limit or otherwise affect the obligation of Borrower hereunder to pay any amount owing with respect to the Obligations.

2.3. Payments.

(a) **Payments by Borrower**. Except as otherwise expressly provided herein, all payments by Borrower shall be made to the DIP Lender Designated Account for the account of DIP Lender and shall be made in immediately available funds, no later than 2:00 p.m. (Eastern time) on the date specified herein. Any payment received by DIP Lender later than 2:00 p.m. (Eastern time) shall be deemed to have been received on the following Business Day and any applicable interest or fee shall continue to accrue until such following Business Day.

(b) **Application**.

- (i) So long as DIP Lender has not commenced an Enforcement Action, all payments and proceeds of Collateral shall be applied in accordance with the DIP Orders; provided, however, that, subject to the DIP Orders, if an Event of Default shall have occurred and be continuing and DIP Lender has commenced an Enforcement Action, any payments made by Debtors and all proceeds of Collateral received by DIP Lender, shall be applied in accordance with Section 9.
- (ii) In the event of a direct conflict between the DIP Orders, the priority provisions of this <u>Section 2.3</u> and other provisions contained in any other Loan Document, it is the intention of the parties hereto that such priority provisions in such documents shall be read together and construed, to the fullest extent possible, to be in concert with each other. In the event of any actual, irreconcilable conflict that cannot be resolved as aforesaid, the terms and provisions of the DIP Orders shall control and govern.

(c) **Prepayments and Repayment**.

(i) <u>Voluntary Prepayments</u>. Subject to the terms of this Agreement (including <u>Section 2.11</u>), Borrower may at any time and from time to time voluntarily prepay the Obligations, in whole or in part, upon not less than five (5) days prior irrevocable written notice to DIP Lender; <u>provided</u>, <u>however</u>, that each partial prepayment shall be in an amount that is an integral multiple of \$25,000 and not less than \$50,000.

(ii) Mandatory Prepayment.

(A) Immediately upon the receipt of Net Cash Proceeds in respect of any Disposition of assets (other than Permitted Dispositions) in excess of \$500,000, through one or a series of

related transactions or Casualty Event, Borrower shall apply 100% of the Net Cash Proceeds received with respect thereto to prepay the outstanding Obligations.

- (B) In the event that the Borrower shall receive Net Cash Proceeds from any issuance or incurrence of Indebtedness (other than Permitted Indebtedness) or issuance of shares of Capital Stock, Borrower shall immediately upon the receipt of such Net Cash Proceeds by Borrower, apply an amount equal to 100% of such Net Cash Proceeds to prepay outstanding Obligations.
- (C) All prepayments under <u>Section 2.3(c)(ii)</u> shall be accompanied by the concurrent payment in immediately available cash of all accrued and unpaid interest that has not been capitalized as part of the Principal Amount as of the date of such prepayment.
- (D) Borrower shall deliver to DIP Lender, (1) at the time of each prepayment required under this <u>Section 2.3(c)(ii)</u>, a certificate signed by a Responsible Officer of Borrower setting forth in reasonable detail the calculation of the amount of such prepayment and (2) at least five (5) days prior to such prepayment, written notice of such prepayment.
- (E) Without affecting any of DIP Lender's rights and remedies hereunder or in respect hereof, if Borrower fails to pay on the Termination Date any amounts required to be paid pursuant to this Section 2.3(c)(ii), then such amounts shall be payable on demand and shall bear interest at the Default Rate until Paid in Full.
- (iii) Repayment on the Termination Date. On the Termination Date, Borrower shall pay, in full and in cash, all outstanding Advances and all other outstanding Obligations owing by it (including all accrued and unpaid interest that has not been added to the Principal Amount as of the Termination Date). Without affecting any of DIP Lender's rights and remedies hereunder or in respect hereof, if Borrower fails to pay on the Termination Date any amounts required to be paid pursuant to this Section 2.3(c)(iii), then such amounts shall be payable on demand and shall bear interest at the Default Rate until Paid in Full.

2.4. Interest Rate: Rates, Payments, and Calculations.

(a) Interest Rate. Except as provided in <u>clause (b)</u> below, all Obligations shall bear interest at a fixed rate equal to 12.00% per annum. Interest in respect of the Obligations shall be due and payable in arrears and shall accrue and compound monthly beginning on the date of each Advance (in cash, by net funding or by deemed issuance) hereunder through the last day of the calendar month in which such Advance is made (or deemed made), and thereafter, beginning on the first day of each subsequent month through the last day of such calendar month, and shall continue to accrue until the Obligations are indefeasibly Paid in Full. Except as otherwise specified

herein, interest shall be paid-in-kind, automatically, and without any further action by Borrower or DIP Lender by capitalizing and adding such amount to the outstanding Principal Amount on the first Business Day of each month. The adjusted outstanding and unpaid Principal Amount together with all other outstanding Obligations shall accrue interest at the rate applicable under this <u>Section</u> 2.4.

- (b) **Default Rate**. Upon the occurrence of an Event of Default, all Obligations (including, but not limited to, the outstanding and unpaid Principal Amount) shall bear interest at the Default Rate.
- (c) **DIP Lender Expenses**. Subject to the DIP Orders, Borrower hereby agrees to pay all DIP Lender Expenses as and when permitted by the DIP Orders but in any event no less frequently than once per month (and, for the avoidance of doubt, all such DIP Lender Expenses shall be Obligations and accrue interest in accordance with Section 2.4(a) from the date of incurrence). All DIP Lender Expenses shall be paid-in-kind, automatically, and without any further action by Borrower or DIP Lender by capitalizing and adding such amount to the outstanding Principal Amount and so long as the timing of each such payment is approved by the DIP Orders, such capitalization shall occur on the first Business Day of each month occurring after the Petition Date. Notwithstanding the foregoing, all professional fees, including the fees of the DIP Lender's legal counsel shall be paid in cash on the Effective Date as required by Section 3 hereof and the Borrower shall remain responsible for paying all fees and expenses of professionals as and when due after the Petition Date to the extent permitted by the DIP Orders and the Bankruptcy Court.
- (d) **Payment**. Interest, fees and all DIP Lender Expenses payable hereunder shall be due and payable on the Termination Date (and on such other date as specified herein). DIP Lender shall be entitled, but not obligated to pay such interest, fees and DIP Lender Expenses under this Agreement and the other Loan Documents and fees and DIP Lender Expenses incurred in connection with the Chapter 11 Cases. Borrower hereby authorizes DIP Lender, from time to time, without prior notice to Borrower, to charge all interest and fees (when due and payable), all DIP Lender Expenses (as and when incurred), and all other payments as and when due and payable under any Loan Document or the DIP Orders to Borrower, which amounts thereafter shall accrue interest at the rate then applicable under this <u>Section 2.4</u>.
- (e) **Computation**. All interest, fees and DIP Lender Expenses chargeable under this Agreement and the other Loan Documents shall be computed on the basis of a 365 day year for the actual number of days elapsed.
- 2.5. <u>Intent to Limit Charges to Maximum Lawful Rate</u>. In no event shall the interest rate or rates payable under this Agreement, plus any other amounts paid in connection herewith, exceed the highest rate permissible under any law that a court of competent jurisdiction shall, in a final determination, deem applicable. Borrower and DIP Lender, in executing and delivering this Agreement, intend legally to agree upon the rate or rates of interest and manner of payment stated within it; <u>provided</u>, <u>however</u>, that, anything contained herein to the contrary notwithstanding, if said rate or rates of interest or manner of payment exceeds the maximum allowable under Applicable Law, then, *ipso facto*, as of the date of this Agreement, Borrower is and shall be liable only for the payment of such maximum as allowed by law, and payment received from Borrower

in excess of such legal maximum, whenever received, shall be applied to reduce the principal balance of the relevant Obligations to the extent of such excess.

- 2.6. <u>Borrower Designated Account</u>. Borrower agrees to establish and maintain the Borrower Designated Account with the Borrower Designated Account Bank for the purpose of receiving the proceeds of each Advance requested by Borrower and made by Lender hereunder. Unless otherwise agreed by Lender and Borrower, any Advance requested by Borrower and made by Lender hereunder shall be made to the Borrower Designated Account.
- 2.7. <u>Facility Fee</u>. On the date of the Interim Advance, Borrower shall owe to DIP Lender an amount equal to \$65,000 (the "<u>Facility Fee</u>") as commitment fee for entering into this facility. The Facility Fee shall be fully earned when due and shall be non-refundable for any reason whatsoever. Concurrently with the Interim Advance, the Facility Fee shall be paid-in-kind, automatically, and without any further action by Borrower or DIP Lender by capitalizing and adding such amount to the outstanding Principal Amount. The adjusted outstanding and unpaid Principal Amount together with all other outstanding Obligations shall accrue interest at the rate applicable under <u>Section 2.4</u>.

2.8. <u>Security Interest; Superpriority Claim and Liens; Collateral.</u>

- (a) As security for the payment and performance of the Obligations and without limiting any other grant of a Lien and security interest in any Security Document, each Debtor hereby assigns and grants to DIP Lender a continuing Lien on and security interest in, upon, and to all assets and properties of such Debtor (whether tangible, intangible, real, personal or mixed), whether now owned or hereafter acquired, and wherever located, before or after the Petition Date, including, without limitation, all cash, Deposit Accounts, Cash Equivalents, Accounts, Inventory, Equipment, equity interests or Capital Stock in domestic and foreign Subsidiaries, Investment Property, Instruments, Chattel Paper, Real Property, leasehold interests, contracts, trademarks, trade names, copyrights, patents, patent rights, licenses, liquor licenses (and the proceeds of any liquor licenses) and other General Intangibles, Receivables, all claims or causes of action (excluding avoidance actions arising under Chapter 5 of the Bankruptcy Code but including all proceeds thereof), and all products, offspring, profits and proceeds of each of the foregoing (the "Collateral"). For the avoidance of doubt, any amounts remaining in the Professional Fee Reserve (as defined in the DIP Orders) after payment of allowed fees and expenses shall be Collateral.
- (b) Subject to the Carve-Out and the terms and provisions of the DIP Orders, each Debtor hereby covenants, represents and warrants that, upon entry of the Interim DIP Order, the Obligations of the Debtors hereunder and under the other Loan Documents:
 - (i) pursuant to Section 364(c)(1) of the Bankruptcy Code, shall at all times constitute joint and several allowed administrative expense claims in the Chapter 11 Cases having priority over all administrative expenses of the kind specified in Sections 105, 326, 328, 330, 331, 503(b), 506(c), 507(a), 507(b), 546(c), 1113 and 1114 of the Bankruptcy Code;
 - (ii) pursuant to Section 364(c)(2) of the Bankruptcy Code, upon the entry of the DIP Orders, shall at all times be secured by a perfected first priority

Lien on and security interest in all of the Collateral that is not subject to a perfected lien or security interest as of the Petition Date;

- (iii) pursuant to Section 364(c)(3) of the Bankruptcy Code, shall at all times be secured by a perfected Lien on and security interest in all of the Collateral, subject only to (A) the validly perfected, enforceable and unavoidable Liens (excluding the CrossFirst Liens) existing on the Petition Date and (B) the Liens (excluding the CrossFirst Liens) existing on the Petition Date but that were perfected subsequent to the Petition Date as permitted by Bankruptcy Code §546(b) (including, without limitation, the Carve-Out), in each case, that is expressly permitted to be senior to the DIP Lender's Liens pursuant to the Interim DIP Order; and
- (iv) pursuant to Section 364(d)(1) of the Bankruptcy Code, shall at all times be secured by a perfected first-priority, senior priming Lien on and security interest in all of the Collateral that is or was subject to a perfected lien or security interest on the Petition Date (including the CrossFirst Liens); <u>provided, however</u> that the Liens and security interests of DIP Lender shall not prime (A) the Liens and security interests granted to the DIP Lender pursuant to Section 364(c)(3) of the Bankruptcy Code as described in Section 2.8(b)(iii) or (B) any Permitted Liens.

(c) (Reserved)

- (d) Except as set forth herein (including with respect to the Carve-Out and Permitted Liens) or in the DIP Orders, the Liens granted hereunder shall not be made subject to or *pari passu* with any Lien or security interest heretofore or hereinafter granted in the Chapter 11 Cases or any Successor Cases and shall be valid and enforceable against any trustee appointed in the Chapter 11 Cases or any Successor Cases, upon the conversion of any of the Chapter 11 Cases to any Successor Case, and/or upon the dismissal or conversion of any of the Chapter 11 Cases or Successor Cases. The Liens granted hereunder shall not be subject to any of sections 510, 549, or 550 of the Bankruptcy Code. No Lien or interest avoided and preserved for the benefit of the estate pursuant to section 551 of the Bankruptcy Code shall be *pari passu* with or senior to the Liens granted hereunder.
- (e) Subject to the priorities set forth in <u>subsection (a)</u> above and the Carve-Out and Permitted Liens, as to all Collateral, including, without limitation, all Real Property, the title to which is held by any Debtor, or the possession of which is held by a Debtor pursuant to a leasehold interest, such Debtor hereby assigns and conveys as security, grants a Lien and security interest in, hypothecates, mortgages, pledges and sets over unto DIP Lender, all of the right, title and interest of such Debtor in all of such Collateral, including without limitation, all owned Real Property and in all such leasehold interests, together in each case with all of the right, title and interest of such Debtor in and to all buildings, improvements, and fixtures related to such Real Property, any lease or sublease thereof, all general intangibles relating thereto and all proceeds thereof. Each Debtor acknowledges that, pursuant to the DIP Orders, the Liens granted in favor of DIP Lender in all of the Collateral shall be perfected by operation of law and without the recordation of any Uniform Commercial Code financing statements, notices of Lien, leases, or other instruments of mortgage or assignment. Notwithstanding this Section 2.8(e), or any failure

on the part of Debtors and DIP Lender to take any further act to perfect, maintain, protect or enforce the Liens and security interests in the Collateral granted hereunder, the DIP Orders shall automatically, and without further action by any Person, perfect such Liens and security interests against the Collateral. Each Debtor further agrees that (i) DIP Lender shall have rights and remedies set forth in Section 8 in respect of the Collateral and (ii) if requested by DIP Lender, each Debtor shall enter into separate Security Documents with respect to such Collateral on terms reasonably satisfactory to counsel to DIP Lender.

2.9. <u>Continuing and Unconditional Guaranty</u>.

- (a) <u>Guarantee</u>. Each Guarantor hereby irrevocably and unconditionally becomes surety for and guarantees to DIP Lender the full and prompt payment and performance of the Obligations.
- (b) Absolute and Irrevocable Guaranty. This Section 2.9 constitutes an absolute, continuing, irrevocable and unconditional guaranty of payment and performance, and not a guaranty of collection, and each Guarantor shall remain liable on its obligations hereunder until full satisfaction of the Obligations. No set-off, counterclaim, recoupment, reduction or diminution of any obligation, or any defense of any kind or nature that the Borrower may have against DIP Lender, any Participant or any subsequent holder of the Obligations or any other party shall be available to, or shall be asserted by, any Guarantor against DIP Lender, and Participant or any subsequent holder of the Obligations or any part thereof (other than Payment in Full).
- Agreement to Pay Obligations. In the event of default by the Borrower in (c) payment or performance of the Obligations, or any part thereof, when such Obligations become due, whether by its terms, by acceleration, or otherwise, each Guarantor shall promptly pay the full amount due thereon to DIP Lender, without notice or demand, in Dollars, and it shall not be necessary for DIP Lender, in order to enforce such payment by any Guarantor, first to institute suit or exhaust its remedies against the Borrower, any other Guarantor or others liable on such Obligations, or to enforce any rights against any Collateral. In the event such payment is made by a Guarantor, then such Guarantor shall be subrogated to the rights then held by DIP Lender with respect to the Obligations to the extent to which the Obligations were discharged by Guarantor. Notwithstanding the foregoing, upon payment by a Guarantor of any sums to DIP Lender hereunder, all rights of each Guarantor against the Borrower, any other guarantor or any collateral arising as a result therefrom by way of right of subrogation, reimbursement, contribution or otherwise shall in all respects be subordinate and junior in right of payment to the prior full satisfaction of the Obligations. All payments received by DIP Lender hereunder shall be applied pursuant to Section 2.3(b) or Section 9, as applicable.
- (d) <u>Stay of Acceleration</u>. If acceleration of the time for payment of any amount payable by Borrower under the Obligations is stayed upon the insolvency, bankruptcy, or reorganization of the Borrower, all such amounts otherwise subject to acceleration under the terms of the Obligations shall nonetheless be payable by Guarantors hereunder forthwith on demand by DIP Lender.

- Obligations Not Impaired. Each Guarantor hereby agrees that its (e) obligations under the Loan Documents shall not be released, discharged, diminished, impaired, reduced, or affected for any reason or by the occurrence of any event, including, one or more of the following events, whether or not with notice to or the consent of any Guarantor: (i) the taking or accepting of collateral as security for any or all of the Obligations or the release, surrender, exchange, or subordination of any Collateral; (ii) any partial release of the liability of any Guarantor hereunder, or the full or partial release of any other guarantor from liability for any or all of the Obligations; (iii) any disability of the Borrower, or the dissolution, insolvency, or bankruptcy of the Borrower, any Guarantor, or any other party at any time liable for the payment of any or all of the Obligations; (iv) any renewal, extension, modification, waiver, amendment, or rearrangement of any or all of the Obligations or any instrument, document, or agreement evidencing, securing, or otherwise relating to any or all of the Obligations; (v) any adjustment, indulgence, forbearance, waiver, or compromise that may be granted or given by DIP Lender to the Borrower, Guarantor, or any other party ever liable for any or all of the Obligations; (vi) any neglect, delay, omission, failure, or refusal of DIP Lender to take or prosecute any action for the collection of any of the Obligations or to foreclose or take or prosecute any action in connection with any instrument, document, or agreement evidencing, securing, or otherwise relating to any or all of the Obligations; (vii) the unenforceability or invalidity of any or all of the Obligations or of any instrument, document, or agreement evidencing, securing, or otherwise relating to any or all of the Obligations; (viii) any payment by the Borrower or any other party to DIP Lender is held to constitute a preference under applicable bankruptcy or insolvency law or if for any other reason DIP Lender is required to refund any payment or pay the amount thereof to someone else; (ix) the settlement or compromise of any of the Obligations; (x) the non-perfection of any Lien securing any or all of the Obligations; (xi) any impairment of any Collateral; (xii) the failure of DIP Lender to sell any Collateral in a commercially reasonable manner or as otherwise required by law; (xiii) any change in the corporate existence, structure, or ownership of the Borrower or any other Guarantor; or (xiv) any other circumstance that might otherwise constitute a defense available to, or discharge of, the Borrower or any Guarantor (other than the full satisfaction of the Obligations).
- (f) <u>Certain Waivers</u>. Each Guarantor hereby waives the benefits of any provision of law requiring that DIP Lender exhaust any right or remedy, or take any action, against the Borrower, any other Guarantor, any other person and/or property.
- (g) <u>Subrogation</u>. No Guarantor shall exercise any right of subrogation, contribution, indemnity, reimbursement or similar rights with respect to any payments it makes under this Agreement, and all such rights are hereby waived. If any amounts are paid to a Guarantor in violation of the foregoing limitation, then such amounts shall be held in trust for the benefit of DIP Lender and shall forthwith be paid to DIP Lender to reduce the amount of the Obligations, whether matured or unmatured.
- (h) <u>Borrower Information</u>. Each Guarantor has, independently and without reliance upon DIP Lender and based upon such documents and information as it has deemed appropriate, made its own analysis and decision to enter into the Loan Documents to which such Guarantor is a party. Each Guarantor has adequate means to obtain from the Borrower on a continuing basis information concerning the financial condition and assets of the Borrower and no Guarantor is relying upon DIP Lender to provide (and DIP Lender shall not have any duty to provide) any such information either now or in the future.

- (i) <u>Benefit of Guaranty</u>. The value of the consideration received and to be received by each Guarantor as a result of this Agreement, the transactions contemplated hereby and Guarantor's executing and delivering this Agreement and the other Loan Documents to which Guarantor is a party, is reasonably worth at least as much as the expected liability and obligation of Guarantor hereunder and thereunder, and such liability and obligation may reasonably be expected to benefit each Guarantor directly or indirectly.
- 2.10. **Joint and Several**. The obligations of the Borrower and each Guarantor hereunder constitute joint and several obligations.
- 2.11. Waiver of any Priming Rights. Upon the Effective Date, and on behalf of itself and its estate, and for so long as any Obligations shall be outstanding, each Debtor hereby irrevocably waives any right, pursuant to Sections 364(c) or 364(d) of the Bankruptcy Code or otherwise, to grant any Lien of equal or greater priority than the Liens securing the Obligations, or to approve a claim of equal or greater priority than the Obligations other than as expressly set forth in the DIP Orders.

3. CONDITIONS PRECEDENT; TERMINATION OF AGREEMENT.

- 3.1. <u>Conditions Precedent to Effectiveness of this Agreement</u>. The effectiveness of this Agreement and the obligation of DIP Lender to make each Advance on or after the Effective Date is subject to the following conditions precedent:
- (a) DIP Lender shall have received from Debtors a counterpart of this Agreement duly executed by each Debtor.
- (b) DIP Lender shall have received from Debtors a counterpart of each of the Security Documents, each duly executed by each Debtor party thereto.
- (c) DIP Lender shall have received the Governing Documents and such other documents and certificates as DIP Lender or its counsel may reasonably request relating to the organization, existence and good standing of each Debtor, the authorization of the transactions contemplated by the Loan Documents, the capacity of each Responsible Officer and any other legal matters relating to any Debtor, the Loan Documents or the transactions contemplated thereby, all in form and substance reasonably satisfactory to DIP Lender.
 - (d) The Debtors shall have initiated the Chapter 11 Cases on the Petition Date.
- (e) The Interim DIP Order shall have been entered, shall be in full force and effect, and shall not have been reversed, vacated or stayed, or modified without the prior written consent of DIP Lender, and all other necessary consents and approvals to the transactions contemplated hereby shall have been obtained and shall be satisfactory to DIP Lender.
- (f) DIP Lender shall have received the initial Approved Budget which shall be in form and substance satisfactory to DIP Lender.

- (g) Subject to the Effect of Bankruptcy, there shall not be pending any litigation or other proceeding, the result of which could reasonably be expected to have a Material Adverse Change.
 - (h) (Reserved)
 - (i) (Reserved)
- (j) The Bankruptcy Court, in the Interim DIP Order, must find and conclude that this Agreement, the other Loan Documents, and the transactions contemplated hereby, were negotiated at arms' length and in good faith, and that DIP Lender is entitled to the protections offered by Section 364(e) of the Bankruptcy Code.
- (k) Debtors shall have complied in full with the notice and other requirements of the Bankruptcy Code in a manner acceptable to DIP Lender and its counsel, with respect to the Interim DIP Order.
- (l) No trustee, examiner or receiver under chapter 7 or chapter 11 of the Bankruptcy Code shall have been appointed in any Bankruptcy Case (nor shall any similar appointment have been made) with respect to any Debtor or any Debtor's business, properties or assets, and no motion shall be pending seeking any such relief.
 - (m) (Reserved).
- (n) The representations and warranties contained in this Agreement or in the other Loan Documents shall be true, correct and complete in all material respects on and as of the date of such extension of credit, as though made on and as of such date (except to the extent that such representations and warranties relate solely to an earlier date in which case such representations and warranties shall be true and correct as of such earlier date).
- (o) No material changes in Applicable Law adversely affecting Debtors or DIP Lender shall have occurred prior to the Effective Date.
- (p) DIP Lender shall have received, in form and substance satisfactory to it, such other assurances, documents or consents related to the foregoing as DIP Lender may reasonably require.
- 3.2. <u>Conditions Precedent to all Advances</u>. The obligation of the DIP Lender to make each Advance shall be subject to the following conditions precedent:
- (a) The DIP Lender shall have received a Notice of Borrowing with respect to such Advance.
- (b) The representations and warranties contained in this Agreement or in the other Loan Documents shall be true, correct and complete in all material respects on and as of the date of such extension of credit, as though made on and as of such date (except to the extent that such representations and warranties relate solely to an earlier date in which case such representations and warranties shall be true and correct as of such earlier date);

- (c) No Default, Event of Default or Material Adverse Change shall have occurred and be continuing on the date of such Advance, nor shall any Default, Event of Default or Material Adverse Change result from the making thereof.
- (d) Such Advance shall not exceed the amounts permitted in accordance with Section 2.1(d) and the proceeds of such Advances shall be used in strict compliance with Section 6.14.
- (e) The DIP Lender shall be satisfied that the Debtors have complied with the Approved Budget and that the requested Advance is consistent with the Approved Budget.
- (f) With respect to the Interim Advance, the Interim DIP Order, and with respect to any other Advance, the Final DIP Order, shall have been entered, shall be in full force and effect, and shall not have been reversed, vacated or stayed, or modified without the prior written consent of the DIP Lender, and all other necessary consents and approvals to the transactions contemplated hereby shall have been obtained and shall be satisfactory to the DIP Lender.
- (g) All motions and other documents to be filed with and submitted to the Bankruptcy Court in connection with this Agreement on or prior to the requested date of such Advance (including, without limitation and as applicable, the Interim DIP Order, the Final DIP Order, the Sale and Bidding Procedures Motion, the Bid Procedures Order and the Sale Order, and each other motion and order contemplated by Section 5.18(c)) shall be in form and substance satisfactory to the DIP Lenders.
- (h) The DIP Lender shall be satisfied that, as of the date of such Advance, Debtors have satisfied the Milestones required to be achieved by such date and is continuing to take action and demonstrate progress towards all future Milestones.
- (i) The DIP Lender shall have received, in form and substance satisfactory to it, such other assurances, documents or consents related to the foregoing as such DIP Lender may reasonably require.
- (j) Borrower has paid (or concurrently with the applicable Advance, will pay) the DIP Lender Expenses to the extent that the DIP Lender has provided to Borrower, on or before the date that is one (1) Business Day prior to such Advance, an invoice reflecting all reasonable and documented fees, costs, disbursements and expenses for which reimbursement is being requested and that have been accrued or incurred by the DIP Lender through such date.

The request by Borrower for, and the acceptance by Borrower of, each Advance hereunder shall be deemed to be a representation and warranty by Debtors that the conditions specified in this Section 3.2 have been satisfied at that time and that after giving effect to such Advance, Borrower shall continue to be in compliance with Section 6.14.

3.3. <u>Term</u>. This Agreement shall continue in full force and effect until the Termination Date.

3.4. Effect of Termination. Upon the Termination Date, all Obligations immediately shall become due and payable without notice or demand. No termination of this Agreement, however, shall relieve or discharge Debtors of their duties, Obligations, or covenants hereunder or under any other Loan Document and DIP Lender's Liens in the Collateral shall remain in effect until all Obligations have been Paid in Full and DIP Lender's obligations to provide additional credit hereunder have been irrevocably terminated. When this Agreement has been terminated and all of the Obligations have been Paid in Full and DIP Lender's obligations to provide additional credit under the Loan Documents have been irrevocably terminated, DIP Lender will, at Borrower's sole expense, execute and deliver any termination statements, Lien releases, mortgage releases, re-assignments of trademarks, discharges of security interests, and other similar discharge or release documents (and, if applicable, in recordable form) as are reasonably necessary or requested to release, as of record, DIP Lender's Liens and all notices of security interests and liens previously filed by DIP Lender with respect to the Obligations.

4. REPRESENTATIONS AND WARRANTIES.

In order to induce DIP Lender to enter into this Agreement, each Debtor makes the following representations and warranties to DIP Lender which shall be true, correct, and complete, as of the Effective Date and as of the date of the making of each Advance (or other extension of credit) made hereunder, as though such representation and warranty were made on and as of the date of such Advance (or other extension of credit) (except to the extent that such representations and warranties relate solely to an earlier date, in which case such representations and warranties shall be true, correct and complete as of such earlier date) and such representations and warranties shall survive the execution and delivery of this Agreement:

- 4.1. <u>Due Organization and Qualification</u>. Each Debtor and each of its Subsidiaries (a) is duly organized and existing and in good standing under the laws of the jurisdiction of its organization and qualified to do business in any state where the failure to be so qualified reasonably could be expected to result in a Material Adverse Change, and (b) subject to Bankruptcy Court approval, has all requisite power and authority to own and operate its properties, to carry on its business as now conducted and as proposed to be conducted, to execute, deliver and perform the Loan Documents to which it is a party and to carry out the transactions contemplated thereby, except where failure to do so could not reasonably be expected to cause a Material Adverse Change.
- 4.2. <u>Due Authorization; Enforceability</u>. The execution, delivery, and performance by Debtors of this Agreement and the other Loan Documents to which it is a party have, upon the entry of the Interim DIP Order or the Final DIP Order, as applicable, been duly authorized by all necessary action on the part of each Debtor. This Agreement has been duly executed and delivered by each Debtor and upon the entry of the Interim DIP Order or the Final DIP Order, as applicable, constitutes, and each other Loan Document to which such Debtor is a party, when executed and delivered by such Debtor will constitute, a legal, valid and binding obligation of such Debtor, enforceable in accordance with its terms.
- 4.3. <u>No Conflict</u>. Subject to the entry of the Interim DIP Order or the Final DIP Order, as applicable, the execution, delivery, and performance by each Debtor of this Agreement and the other Loan Documents to which it is a party do not and will not (a) require any registration with,

consent, or approval of, or notice to, or other action with or by, any Governmental Authority, except for (i) such as have been obtained or made and are in full force and effect, or (ii) filings, the failure of which to obtain would not reasonably be expected to result in a Material Adverse Change, (b) violate any Applicable Law, (c) violate the Governing Documents of such Debtor, (d) except as set forth on Schedule 4.3, conflict with, result in a breach of, or constitute (with due notice or lapse of time or both) a default under any indenture or any other agreement, instrument or other evidence of Material Indebtedness or any other Material Contract, (e) result in or require the creation or imposition of any Lien of any nature whatsoever upon any properties or assets of such Debtor, other than Liens created under the Loan Documents and Permitted Liens and the Carve-Out, and (f) require any approval of such Debtor's equity holders or any approval or consent of any Person under any material contractual obligation of such Debtor, other than consents or approvals that have been obtained and that are still in force and effect.

- 4.4. **No Encumbrances**. Each Debtor has good and indefeasible title to, or a valid leasehold interest in, its personal property assets and good and marketable title to, or a valid leasehold interest in, its Real Property, in each case, free and clear of Liens except for Permitted Liens and the Carve-Out and the CrossFirst Liens.
 - 4.5. **(Reserved)**.
 - 4.6. **(Reserved)**.
- 4.7. <u>Location of Offices</u>. All offices of the Debtors are located on the locations identified on <u>Schedule 4.7</u> (as such Schedule may be updated pursuant to <u>Section 5.9</u>).
- 4.8. <u>Inventory Records</u>. Debtors keep correct and accurate (in all material respects) records itemizing and describing the type and quantity of their and their Subsidiaries' Inventory and the book value thereof.
- 4.9. <u>State of Incorporation; Location of Chief Executive Office; Organizational Identification Number; Commercial Tort Claims; Books and Records Pertaining to Accounts.</u>
- (a) The jurisdiction of organization of each Debtor and each of its Subsidiaries is set forth on Schedule 4.9(a).
- (b) The chief executive office of each Debtor and each of its Subsidiaries is located at the address indicated on <u>Schedule 4.9(b)</u> (as such Schedule may be updated pursuant to <u>Section 5.9</u>).
- (c) Each Debtor's and each of its Subsidiaries' organizational identification number, if any, are identified on Schedule 4.9(c).
 - (d) (Reserved).
- (e) Except to the extent otherwise agreed to by DIP Lender, the books and records pertaining to Accounts are located at those locations listed on <u>Schedule 4.9(b)</u>.

4.10. <u>Capitalization; Subsidiaries</u>.

- (a) Set forth on Schedule 4.10(a), is a complete and accurate description of the authorized Capital Stock of each Debtor, by class, and, as of the Effective Date, a description of the number of shares of each such class that are issued and outstanding. Other than as described on Schedule 4.10(a), there are no subscriptions, options, warrants, or calls granted by any Debtor relating to any shares of such Debtor's Capital Stock, including any right of conversion or exchange under any outstanding security or other instrument. No Debtor is subject to any obligation (contingent or otherwise) to repurchase or otherwise acquire or retire any shares of its Capital Stock or any security convertible into or exchangeable for any of its Capital Stock.
- (b) Set forth on Schedule 4.10(b), is a complete and accurate list of each Debtor's direct and indirect Subsidiaries, if any, showing for each such Subsidiary: (i) its jurisdiction of organization, (ii) the number of shares of each class of Capital Stock authorized, and (iii) the number and the percentage of the outstanding shares of each such class owned directly or indirectly by such Debtor. All of the outstanding Capital Stock of each such Subsidiary has been validly issued and is fully paid and non-assessable.
 - 4.11. **(Reserved)**.
 - 4.12. **(Reserved)**.
 - 4.13. **(Reserved)**.
 - 4.14. **(Reserved)**.
- 4.15. <u>Intellectual Property</u>. As of the Petition Date, each Debtor and each Subsidiary of each Debtor owns, or holds licenses in, all trademarks, trade names, copyrights, and licenses that are necessary to the conduct of its business as currently conducted and the use thereof by each such Debtor and each such Subsidiary does not infringe upon the rights of any other Person, except for any such infringements that, individually or in the aggregate, would not reasonably be expected to result in a Material Adverse Change.
- 4.16. <u>Leased and Owned Real Property</u>. Set forth on <u>Schedule 4.16</u> is a listing of all leased real property of the Debtors. The Debtors do not own any real property.
- 4.17. <u>Deposit Accounts and Securities Accounts</u>. Set forth on <u>Schedule 4.17</u> is a listing of all of each Debtor's Collateral Accounts.
- 4.18. <u>Complete Disclosure</u>. To the knowledge of the Debtors, all factual information furnished by or on behalf of the Debtors in writing to DIP Lender (including all information contained in the Schedules hereto or in the other Loan Documents) for purposes of or in connection with this Agreement, the other Loan Documents, or any transaction contemplated herein or therein is, and all other such factual information hereafter furnished by or on behalf of the Debtors in writing to DIP Lender will be, true and accurate in all material respects on the date as of which such information is dated or certified and not incomplete by omitting to state any fact necessary to make such information not misleading in any material respect at such time in light of the circumstances under which such information was provided. The Approved Budget is based upon

good faith estimates and assumptions believed by management of the Borrower to be reasonable at the time made..

- 4.19. **(Reserved)**.
- 4.20. (**Reserved**).
- 4.21. (**Reserved**).
- 4.22. (**Reserved**).
- 4.23. <u>Labor Matters</u>. There are no collective bargaining agreements covering the employees of Debtors. The Debtors have not suffered any strikes, walkouts, work stoppages or other material labor difficulty within the last five (5) years.
- 4.24. Security Documents; Priority of Liens. The Security Documents and the DIP Orders (a) will create in favor of DIP Lender, as security for all Obligations, legal, valid, continuing and enforceable Liens in the Collateral, and (b) constitute a fully perfected, enforceable and unavoidable first priority Lien on, and security interest in, and hypothecation of, all right, title and interest of the Borrower thereunder in all Collateral, except as otherwise provided in the DIP Orders, and no further recording, filing or other action of any kind will be required in connection with the creation, perfection or enforcement of such security interests and Liens. Subject to the Carve-Out and Permitted Liens, no claims having a priority superior or *pari passu* to the Liens granted to DIP Lender shall be granted or approved while any of the Obligations or the Loan Commitment remain outstanding.
- 4.25. <u>Insurance</u>. The Debtors have in place (as of the Petition Date and on each date of an Advance hereunder) insurance that complies with <u>Section 5.8</u>.
- 4.26. <u>Compliance with Statutes, etc.</u> Each Debtor and its Subsidiaries is in material compliance with all applicable statutes, regulations and orders of, and all applicable restrictions imposed by, all Governmental Authorities, in respect of the conduct of its business and the ownership of its property (including compliance with all applicable Environmental Laws with respect to any Real Property or governing its business and the requirements of any permits issued under such Environmental Laws with respect to any such Real Property or the operations of such Borrower and its Subsidiaries). Each Debtor and its Subsidiaries is in material compliance with, and has procured and is now in possession of, all material licenses or permits required by all Governmental Authorities for the conduct of its business and the ownership of its property.
- 4.27. Patriot Act. To the extent applicable, each Debtor and its Subsidiaries is in compliance with the (a) Trading with the Enemy Act, as amended, and each of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V, as amended) and any other enabling legislation or executive order relating thereto, and (b) the Patriot Act. No part of the proceeds of the Advances will be used, directly or indirectly, for any payments to any governmental official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended.

5. AFFIRMATIVE COVENANTS.

Each Debtor covenants and agrees that, until termination of the Loan Commitment and all of the Obligations are Paid in Full, such Debtor shall, and shall cause each of its Subsidiaries to, do all of the following:

5.1. **Reports and Information**. Furnish to DIP Lender:

- (a) At least one Business Day prior to the date of the Initial Advance, the initial Approved Budget, and prior to the adoption thereof, any proposed subsequent modifications or supplements to the Approved Budget;
- (b) On or before Thursday of each Fiscal Week, commencing with the first full Fiscal Week following the Petition Date, a Variance Report for the immediately preceding Fiscal Week:
 - (c) (Reserved);
 - (d) (Reserved);
- (e) Promptly upon filing its monthly operating report with the U.S. Trustee (unless otherwise agreed to in writing by DIP Lender), deliver or cause to be delivered a copy of such report to DIP Lender;
 - (f) (Reserved);
- (g) Within five (5) days after the end of each Fiscal Month, a summary of the amount of Professional Fees and Expenses that have been incurred or accrued (but remain unpaid) as of the end of such Fiscal Month;
- (h) Promptly upon receipt thereof, copies of all reports submitted to Debtors by independent certified public accountants in connection with each annual, interim, or special audit of the books of Debtors or any of their Subsidiaries made by such accountants, including any management letter commenting on Debtors' internal controls submitted by such accountants to management in connection with its annual audit;
- (i) Except as exigencies may dictate, all draft documents, pleadings and motions, in each case at least three (3) Business Days prior to filing with the Bankruptcy Court and copies of all pleadings, motions, applications, judicial information, financial information and other documents filed by any Debtor with the Bankruptcy Court or the Creditors' Committee (the form and substance of any pleadings or documents related to the sale of the Debtors' assets or the DIP Facility shall be satisfactory to the DIP Lender); and
- (j) Promptly following any request therefor, such other information regarding the Collateral, operations, business affairs and financial condition of Debtors, or compliance with the terms of any Loan Document, as DIP Lender may reasonably request.

- 5.2. <u>Accounting System</u>. Maintain a system of accounting that enables Debtors to produce financial statements in accordance with GAAP and maintain records pertaining to the Collateral that contain information as from time to time reasonably may be requested by DIP Lender.
- 5.3. <u>Notices of Material Events</u>. Furnish to DIP Lender prompt written notice of the occurrence of any of the following:
- (a) A Default or Event of Default, specifying the nature and extent thereof and the action (if any) which is proposed to be taken with respect thereto;
- (b) The filing or commencement of any action, suit or proceeding by or before any arbitrator or Governmental Authority (including, without limitation, any of the foregoing relating to any Environmental Liability or potential Environmental Liability) against or affecting any Debtor or any Subsidiary thereof;
- (c) Any development (including, without limitation, any Environmental Liability or potential Environmental Liability) that results in, or would reasonably be expected to result in, a Material Adverse Change;
 - (d) Any change in Borrower's chief restructuring officer;
- (e) The discharge by Debtors of their present independent accountants or any withdrawal or resignation by such independent accountants;
- (f) Any collective bargaining agreement or other contract with organized labor to which any Debtor becomes a party, or the application for the certification of a collective bargaining agent;
- (g) The filing of any Lien for unpaid Taxes incurred or arising Post-Petition against any Debtor, except those Taxes which are being contested in good faith by appropriate proceedings diligently conducted and for which adequate reserves have been provided in accordance with GAAP;
- (h) Any Casualty Event or other damage to any material portion of the Collateral or the commencement of any action or proceeding for the taking of any interest in a material portion of the Collateral or any part thereof or interest therein under power of eminent domain or by condemnation or similar proceeding; and
- (i) The issuance of any material license or permit by any Governmental Authority or any alleged non-compliance with, proposed modification to, or termination of any material license or permit issued by any Governmental Authority.

Each notice delivered under this Section shall be accompanied by a statement of a Responsible Officer of Borrower setting forth the details of the event or development requiring such notice and, if applicable, any action taken or proposed to be taken with respect thereto.

- 5.4. Material Contracts. Subject to the Bankruptcy Code, (a) observe and perform all of the terms, covenants, conditions and provisions of the Material Contracts to be observed and performed by it, except to the extent the failure to do so could not reasonably be expected to result in a Material Adverse Change, (b) not do, permit, suffer or refrain from doing anything that would reasonably be expected to result in a material default under or breach of any of the terms of any Material Contract, (c) not cancel, permit the termination of or surrender any Material Contract except as any Material Contract expires in accordance with its terms, and (d) give DIP Lender prompt written notice of any material breach of any Material Indebtedness, or any default, by any Debtor, or the knowledge of any Debtor of any other party, under any Material Contract, and deliver to DIP Lender a copy of each notice of default (other than any such breach or default caused by an Effect of Bankruptcy) or any failure to pay amounts due or the filing of Lien notices, affidavits or the like.
- 5.5. <u>Inspection</u>. Permit DIP Lender and each of its representatives or agents to visit any of its properties and inspect any of its assets or books and records, during normal business hours, to make physical verifications and other appraisals of the Collateral, and business valuations, to examine and make copies of its books and records, and to discuss its affairs, finances, and accounts with, and to be advised as to the same by, its officers and employees at such reasonable times and intervals as DIP Lender may designate and, so long as no Default or Event of Default exists, with reasonable prior notice to Borrower.

5.6. **Maintenance of Properties and Books and Records**. Maintain and preserve:

- (a) Subject to <u>Section 6.4</u>, all of its properties which are necessary or useful in the proper conduct of its business in good working order and condition, ordinary wear, tear, and casualty excepted, and comply at all times with the provisions of all material leases to which it is a party as lessee, so as to prevent any loss or forfeiture thereof or thereunder; and
- (b) Proper books and records, in which full, true and correct (in all material respects) entries are made of all financial transactions and matters involving their respective properties and businesses consistent with past practices; and in material conformity with all applicable requirements of any Governmental Authority having regulatory jurisdiction over them, as the case may be.
- 5.7. Payment of Obligations; Taxes. Subject to the Bankruptcy Code, pay, and cause its Subsidiaries to pay, its liabilities for Taxes, and claims for labor, materials, or supplies, in each case, to the extent incurred Post-Petition, before delinquency or default or before the expiration of any extension period, except to the extent that the validity of such assessment or Tax shall be the subject of a Permitted Protest. Subject to the Effect of Bankruptcy, each Debtor will, make, and cause its Subsidiaries to make, timely payment or deposit of all Tax payments and withholding Taxes required of them by Applicable Laws, including those laws concerning F.I.C.A., F.U.T.A., state disability, and local, state, and federal income Taxes, and will, upon request, furnish DIP Lender with proof satisfactory to DIP Lender indicating that Debtors or any Subsidiary of Borrower has made such payments or deposits.

5.8. **Insurance**.

- (a) At Debtors' expense, maintain insurance respecting Debtors' and their Subsidiaries' assets wherever located, covering loss or damage by fire, theft, explosion, and all other hazards and risks as ordinarily are insured against by other Persons engaged in the same or similar businesses. Each Debtor also shall maintain, and cause its Subsidiaries to maintain, business interruption, public liability, and product liability insurance. Each Debtor shall also maintain, and cause its Subsidiaries to maintain, all insurance coverage as required by any Material Contract or Applicable Law. All such policies of insurance shall be in such amounts and with such insurance companies as are reasonably satisfactory to DIP Lender. Upon the request of the DIP Lender, Borrower shall deliver copies of all such policies to DIP Lender with an endorsement naming DIP Lender as the sole loss payee (under a satisfactory DIP Lender's loss payable endorsement) or additional insured, as appropriate. Each policy of insurance or endorsement shall contain a clause requiring the insurer to give not less than 30 days prior written notice to DIP Lender in the event of cancellation of the policy for any reason (or 10 days prior written notice in the event of cancellation of the policy due to non-payment).
- (b) Debtors shall give DIP Lender prompt notice of any loss exceeding \$50,000 covered by such insurance. So long as no Event of Default has occurred and is continuing, Debtors shall have the exclusive right to adjust any losses payable under any such insurance policies which are less than \$25,000. Following the occurrence and during the continuation of an Event of Default, or in the case of any losses payable under such insurance exceeding \$25,000, DIP Lender shall have the exclusive right to adjust any losses payable under any such insurance policies, without any liability to Debtors whatsoever in respect of such adjustments. Any monies received as payment for any loss under any insurance policy mentioned above (other than liability insurance policies) or as payment of any award or compensation for condemnation or taking by eminent domain, shall be paid over to DIP Lender to be applied at the option of DIP Lender either to the prepayment of the Obligations in accordance with Section 2.3(c)(ii) or to be disbursed to Borrower under staged payment terms reasonably satisfactory to DIP Lender for application to the cost of repairs, replacements, or restorations.
- 5.9. Location of Offices. (a) Keep each Debtor's and its Subsidiaries' chief executive offices only at the locations identified on Schedule 4.9(b); provided, however, that Borrower may amend Schedule 4.9(b) so long as such amendment occurs by written notice to DIP Lender not less than 30 days prior to the date on which such chief executive office is relocated, so long as such new location is within the continental United States, and so long as, at the time of such written notification, the Borrower uses commercially reasonable efforts to provide the DIP Lender a Collateral Access Agreement with respect thereto), (b) keep each Debtor's and its Subsidiaries' books and records pertaining to Accounts at locations for which Collateral Access Agreements have been obtained (to the extent any such Collateral Access Agreement is requested by DIP Lender), (c) provide DIP Lender with an updated Schedule 4.7 every 90 days (or more frequently if requested by DIP Lender) and (d) use commercially reasonable efforts to provide DIP Lender with fully executed Collateral Access Agreements within 10 business days of DIP Lender requesting any such Collateral Access Agreement.
- 5.10. <u>Compliance with Laws</u>. Comply with the requirements of all Applicable Laws, rules, regulations, and orders of any Governmental Authority, and all rights, privileges, permits,

licenses and franchises necessary or desirable in the normal conduct of each Debtor's or any of its Subsidiaries' respective businesses.

- 5.11. <u>Leases</u>. Subject to the Bankruptcy Code, pay when due all rents and other amounts incurred Post-Petition under any material leases to which any Debtor or any Subsidiary of any Debtor is a party or by which any Debtor's or any of its Subsidiaries' properties and assets are bound, unless such payments are the subject of a Permitted Protest.
- 5.12. Existence. Subject to the Bankruptcy Code, at all times (a) preserve, renew and maintain in full force and effect each Debtor's and each of its Subsidiaries' respective legal existence and good standing under the Applicable Laws of the jurisdiction of their organization, (b) take all reasonable actions to maintain all rights, privileges, permits, licenses and franchises necessary or desirable in the normal conduct of their respective businesses, (c) use the standard of care typical in the industry in the operation and maintenance of its facilities, and (d) preserve or renew all of their respective registered Intellectual Property as it deems necessary or desirable in its business judgment (provided that the Debtors shall consult with the DIP Lender with respect to any Intellectual Property that any Debtor has determined it will cease to preserve or protect prior to taking or failing to take any relevant action).
- 5.13. **Use of Proceeds**. The proceeds of all Advances made hereunder will be used only in strict compliance with Section 6.14: (a) for general corporate purposes, in each case to the extent (i) not prohibited by Applicable Law and (ii) expressly permitted under the Loan Documents, the DIP Orders and in accordance with Section 6.14; (b) to refinance the Prepetition Bridge Loans, together with all accrued and unpaid interest, fees and other amounts under the Prepetition Secured Note; (c) to pay fees and expenses of Debtors' professionals in the Chapter 11 Cases, including those associated with the negotiation, execution and delivery of this Agreement and the other Loan Documents and in connection with the transactions contemplated hereby and thereby and, to the extent approved by the Bankruptcy Court and as set forth in the DIP Orders, in connection with the Chapter 11 Cases; and (d) to pay interest, fees and DIP Lender Expenses under this Agreement and the other Loan Documents. No part of the proceeds of any Advance will be used, whether directly or indirectly: (i) to finance in any way any action, suit, arbitration, proceeding, application, motion or other litigation of any type adverse to the interests of DIP Lender or its rights and remedies under this Agreement, the other Loan Documents or any of the DIP Order, including, without limitation, for the payment of any services rendered by the professionals retained by Debtors or an official committee of unsecured creditors in connection with the assertion of or joinder in any claim, counterclaim, action, proceeding, application, motion, objection, defense or other contested matter, the purpose of which is to seek, or the result of which would be to obtain, any order, judgment determination, declaration or similar relief (x) invalidating, setting aside, avoiding or subordinating, in whole or in part, the Obligations or the Prepetition Bridge Loans or the Liens securing the Obligations or the Prepetition Bridge Loans, (y) for monetary, injunctive or other affirmative relief against DIP Lender or the Collateral, or (z) preventing, hindering or otherwise delaying the exercise by DIP Lender of any rights and remedies under any of the DIP Orders, the Loan Documents or Applicable Law, or the enforcement or realization (whether by foreclosure, credit bid, further order of the court or otherwise) by DIP Lender upon any of the Collateral, (ii) to make any distribution under a Reorganization Plan, (iii) to make any payment in settlement of any claim, action or proceeding, before any court, arbitrator or other governmental body without the prior written consent of DIP Lender, (iv) to pay any fees or

similar amounts to any Person who has made an offer to Debtors or that has proposed or may propose to purchase shares of Capital Stock in any Debtor without the prior written consent of DIP Lender, or (v) to object to or challenge, or (vi) for any purpose that entails a violation of any of the regulations of the Board, including Regulations U and X.

- 5.14. **Environmental**. Except as otherwise required by the Bankruptcy Code or by a Final Order, will, and will cause its Subsidiaries to, (a) keep any property either owned or operated by any Debtor or any Subsidiary of any Debtor free of any Environmental Liens or post bonds or other financial assurances sufficient to satisfy the obligations or liability evidenced by such Environmental Liens, (b) comply with Environmental Laws in all material respects, and provide to DIP Lender documentation of such compliance which DIP Lender reasonably requests, (c) promptly notify DIP Lender of any release of a Hazardous Material in any reportable quantity from or onto property owned or operated by any Debtor or any Subsidiary of any Debtor and take any Remedial Actions required to abate said release or otherwise to come into compliance with applicable Environmental Law, and (d) promptly, but in any event within 5 days of its receipt thereof, provide DIP Lender with written notice of any of the following: (i) that an Environmental Lien has been filed against any of the real or personal property of any Debtor or any Subsidiary of any Debtor, (ii) of the commencement of any Environmental Action or notice that an Environmental Action will be filed against any Debtor or any Subsidiary of any Debtor, and (iii) of a violation, citation, or other administrative order which reasonably could be expected to result in a Material Adverse Change.
- 5.15. <u>Disclosure Updates</u>. Promptly and in no event later than five (5) Business Days after obtaining knowledge thereof, notify DIP Lender if any written information, exhibit, or report furnished to DIP Lender contained, at the time it was furnished, any untrue statement of a material fact or omitted to state any material fact necessary to make the statements contained therein not misleading in light of the circumstances in which made. The foregoing to the contrary notwithstanding, any notification pursuant to the foregoing provision will not cure or remedy the effect of the prior untrue statement of a material fact or omission of any material fact nor shall any such notification have the effect of amending or modifying this Agreement or any of the Schedules hereto.
- 5.16. Control of Collateral and Control Agreements. If requested by DIP Lender, take reasonable steps as requested by DIP Lender in order for DIP Lender to obtain control in accordance with the UCC with respect to all of its Collateral Accounts, electronic chattel paper, investment property, and letter of credit rights. For the avoidance of doubt, Debtors hereby acknowledge and agree that the Liens granted hereunder in all such Collateral has been perfected by the DIP Orders without any need of entering into a Control Agreement or taking any other steps to obtain control of such Collateral and any such agreement or such other action shall be in addition to the rights afforded DIP Lender pursuant to the DIP Orders.
- 5.17. **Further Assurances**. File and prosecute any and all motions, pleadings or other filings with the Bankruptcy Court (in each case in form and substance satisfactory to DIP Lender), execute any and all further documents, financing statements, agreements and instruments, and take all such further actions (including the filing and recording of financing statements and other documents), that may be required under this Agreement, any of the other Loan Documents or any Applicable Law, or which DIP Lender may reasonably request, to effectuate the transactions

contemplated by the Loan Documents or to grant, preserve, protect or perfect the Liens created or intended to be created by the Security Documents or the validity or priority of any such Lien, all at the expense of Debtors. To the extent that DIP Lender elects to file financing statements to further evidence the Liens created hereunder and perfected by the DIP Orders, each Debtor agrees that the Collateral may be described therein as "all assets" or words of similar import.

5.18. Bankruptcy Related Affirmative Covenants; Milestones.

- (a) Promptly provide DIP Lender with updates of any material developments in connection with each Debtor's sale under the Chapter 11 Cases.
- (b) Promptly, punctually, and faithfully perform all and singular the terms and conditions of the DIP Orders.
- (c) Comply with each of the following Bankruptcy milestones (the "Milestones"):
 - (i) No later than five (5) days after the Petition Date, the Bankruptcy Court shall have entered the Interim DIP Order in form and substance satisfactory to the DIP Lender.
 - (ii) No later than thirty (30) days after the Petition Date, the Bankruptcy Court shall have entered the Final DIP Order in form and substance satisfactory to the DIP Lender.
 - (iii) No later than two (2) days after the Petition Date, the Debtors shall have filed a motion to sell, and to approve bidding procedures for a 363 Sale of all of Debtors' assets to the DIP Lender, in manner, form and substance acceptable to DIP Lender (the "Sale and Bidding Procedures Motion").
 - (iv) No later than thirty (30) days after the Petition Date, the Bankruptcy Court shall have entered a Final Order (the "<u>Bid Procedures Order</u>") in form and substance acceptable to the DIP Lender, approving the bidding procedures set forth in the Sale and Bidding Procedures Motion.
 - (v) No later than sixty-six (66) days after the Petition Date, the Bankruptcy Court shall have entered a Final Order (the "Sale Order") in form and substance acceptable to the Lender, approving the 363 Sale of all of Debtors' assets to the DIP Lender.
 - (vi) No later than fifteen (15) days after the entry of the Sale Order, the Debtors shall have consummated the 363 Sale of all of Debtors' assets to the DIP Lender in a manner consistent with the Sale Order.

5.19. Approved Budget; Variance Reports.

(a) The initial Approved Budget shall be the budget attached to the Interim DIP Order.

- (b) The Approved Budget may be amended by Borrower at any time with the prior written consent of DIP Lender.
- (c) Any Advances under this Agreement shall be limited in accordance with the Approved Budget (subject to any variances permitted under this Agreement or the DIP Orders). The initial Approved Budget shall be attached to the Interim DIP Order.
- (d) The Approved Budget may be updated, modified, or supplemented from time to time, and each such updated, modified, or supplemented budget shall be subject to approval by, and in form and substance satisfactory to, DIP Lender in its sole discretion. Each proposed Approved Budget shall be prepared in good faith based upon the assumptions which Borrower believes to be reasonable and are satisfactory to DIP Lender.
- 5.20. Professional Fee Reserve. Subject in all respects to the DIP Order, on the date of the Interim Advance, the Borrower shall deposit into a segregated account funds in an amount equal to the aggregate total of (i) fees anticipated to be paid to the Clerk of the Court and to the U.S. Trustee pursuant to 28 U.S.C. §1930(a) *plus* interest at the statutory rate, if any, pursuant to 31 U.S.C § 3717, (ii) all anticipated professional fees of the Borrower and anticipated professional fees of the Creditors' Committee for the period beginning on the Petition Date through the consummation of the 363 Sale and (iii) any other amounts constituting part of the Professional Fee Reserve (as defined in the DIP Order), which such funds shall be held in such segregated account by Borrower for the benefit of and the purposes set out in the DIP Orders and shall be disbursed by Borrower as the applicable invoices are approved by the Bankruptcy Court, as further detailed in the DIP Orders and in a manner consistent with the Approved Budget.
- 5.21. **Post-Closing Requirements**. Comply with the requirements set forth on <u>Schedule</u> 5.21.

6. NEGATIVE COVENANTS.

Each Debtor covenants and agrees that, until termination of the Loan Commitment and all of the Obligations are Paid in Full, such Debtor will not, and will not permit any of its Subsidiaries to, do any of the following:

- 6.1. <u>Indebtedness</u>. Create, incur, assume, suffer to exist, guarantee, or otherwise become or remain, directly or indirectly, liable with respect to any Indebtedness, except Permitted Indebtedness.
- 6.2. <u>Liens</u>. Create, incur, assume, or suffer to exist, directly or indirectly, any Lien on or with respect to any of its assets, of any kind, whether now owned or hereafter acquired, or any income or profits therefrom, except for Permitted Liens, CrossFirst Liens and the Carve-Out.
- 6.3. <u>Restrictions on Fundamental Changes</u>. Other than as required or permitted hereunder or under the DIP Orders:
- (a) Enter into any merger, consolidation, reorganization, or recapitalization, or reclassify its Capital Stock;

- (b) Liquidate, wind up, or dissolve or suffer any liquidation or dissolution;
- (c) Dispose of all or any substantial part of its assets, in one transaction or a series of transactions; or
- (d) Issue any additional shares of Capital Stock of any Debtor or any of its Subsidiaries.
- 6.4. <u>Disposal of Assets.</u> Convey, sell, lease, license, assign, transfer, or otherwise dispose of any of the assets of any Debtor or any Subsidiary, including any Capital Stock, except Permitted Dispositions.
- 6.5. <u>Change Name</u>. Change any Debtor's or any Subsidiaries' name, organizational identification number, state of organization, or organizational identity without DIP Lender's prior written consent.
- 6.6. <u>Nature of Business</u>. Make any change in the principal nature of their business as of the type conducted by Debtors on the Effective Date.

6.7. Restricted Payments; Certain Payments of Indebtedness.

- (a) Declare or make, or agree to pay or make, directly or indirectly, any Restricted Payment.
- (b) Make, distribute or agree to pay, make or distribute, directly or indirectly, any payment or other distribution (whether in cash securities or other property) of or in respect of principal of, or interest, premium or fees on, any Indebtedness, including any sinking fund or similar deposit, on account of the purchase, redemption, retirement, acquisition, cancellation or termination of any Indebtedness, except any such payments or prepayments (i) required to be made under the DIP Orders, and (ii) expressly authorized by DIP Lender in writing.
- 6.8. **Change of Control**. Cause, permit, or suffer, directly or indirectly, any Change of Control.
- 6.9. **Consignments**. Consign any of their Inventory or sell any of their Inventory on bill and hold, sale or return, sale on approval, or other conditional terms of sale.
- 6.10. <u>Accounting Methods</u>. Modify or change their fiscal year or their method of accounting without the prior written consent of DIP Lender or enter into, modify, or terminate any agreement currently existing, or at any time hereafter entered into with any third party accounting firm or service bureau for the preparation or storage of Debtors' or their Subsidiaries' accounting records without said accounting firm or service bureau agreeing to provide DIP Lender information regarding Debtors' and their Subsidiaries' financial condition.

6.11. **Investments**.

- (a) Except for Permitted Investments, directly or indirectly, make or acquire any Investment, or incur any liabilities (including contingent obligations) for or in connection with any Investment.
- (b) Permit its Subsidiaries to, form any additional Subsidiaries or enter into any joint venture without the prior written consent of DIP Lender.
- 6.12. Transactions with Affiliates. Except as set forth in Schedule 6.12, directly or indirectly enter into or permit to exist any transaction with any Affiliate of any Debtor; provided, the foregoing restriction shall not apply (to the extent that the same are not inconsistent with the orders of the Bankruptcy Court or with the Bankruptcy Code and are reflected in the Approved Budget) to transactions that (a) are in the ordinary course of Debtors' business, (b) are upon fair and reasonable terms, (c) if they involve one or more payments by any Debtor or any Subsidiaries in excess of \$50,000, any such payments and transactions are fully disclosed to DIP Lender, (d) are no less favorable to Debtors or their Subsidiaries, as applicable, than would be obtained in an arm's length transaction with a similarly situated non-Affiliate, and (e) are not prohibited under this Agreement (including, without limitation, Permitted Dispositions, Permitted Investments and Permitted Indebtedness).
- 6.13. <u>Inventory and Equipment with Bailees</u>. Store any material portion of the Inventory or Equipment of any Debtor at any time now or hereafter with a bailee, warehouseman, or similar party.
- 6.14. <u>Compliance with the Approved Budget</u>. Contract for, purchase or make any expenditure (including any Capital Expenditure) except in accordance with the Approved Budget and for replacements of fixed assets or improvements to the extent made from insurance proceeds arising from the damage, destruction, or loss of Equipment or other capital assets, to the extent such expenditure(s) is approved by DIP Lender in its sole discretion.
- 6.15. Restrictive Agreements. Other than as in accordance with the DIP Orders, incur or permit to exist any agreement or other arrangement that prohibits, restricts or imposes any condition upon the ability of any Debtor to create, incur or permit to exist any Lien upon any of its property or assets; provided that the foregoing shall not apply to (i) restrictions and conditions imposed by Applicable Law or by any Loan Document, (ii) restrictions or conditions imposed by any agreement relating to secured Indebtedness permitted by this Agreement if such restrictions or conditions apply only to the property or assets securing such Indebtedness, (iii) customary provisions in leases restricting the assignment or subleasing thereof, and (iv) licenses or contracts which by the terms of such licenses and contracts prohibit the granting of Liens on the rights contained therein.
- 6.16. <u>Amendment of Material Documents</u>. Amend, modify or waive: (a) (i) its Governing Documents, (ii) any Material Contract, or (iii) any agreement or document governing its Indebtedness, or (b) the subordination, payment or maturity provisions of any Indebtedness subordinated to the Obligations.
- 6.17. **Bankruptcy Related Negative Covenants**. No Debtor will seek, consent to, or permit to exist any of the following:

(a) (Reserved);

- (b) Any order which authorizes the assumption or rejection of any executory contract or unexpired lease of any Debtor, without DIP Lender's prior written consent, such consent not to be unreasonably withheld, delayed or conditioned;
- (c) Any modification, stay, vacation or amendment to any DIP Order to which DIP Lender has not consented in writing, such consent not to be unreasonably withheld, delayed or conditioned;
- (d) A priority claim or administrative expense or unsecured claim against any Debtor (now existing or hereafter arising or any kind or nature whatsoever, including, without limitation, any administrative expense of the kind specified in Sections 105, 326, 328, 330, 331, 364(c), 503(a), 503(b), 506(c), 507(a), 507(b), 546(c), 546(d), 726 or 1114 of the Bankruptcy Code) equal or superior to the priority claim of DIP Lender in respect of the Obligations;
- (e) Any Lien on any Collateral having a priority equal or superior to the Lien securing the Obligations, other than the Carve-Out;
- (f) Any order which authorizes the return of any of the Debtors' property pursuant to Section 546(h) of the Bankruptcy Code, provided that the foregoing shall not prohibit any Debtor from returning any consigned inventory to consignors on such terms as are satisfactory to DIP Lender;
- (g) Any order which authorizes the payment of any Indebtedness (other than the Prepetition Bridge Loans, payments permitted pursuant to the DIP Orders or otherwise consented to by DIP Lender) incurred prior to the Petition Date; or
- (h) Any order seeking authority to take any action that is prohibited by the terms of this Agreement or the other Loan Documents or refrain from taking any action that is required to be taken by the terms of this Agreement or any of the other Loan Documents.

7. EVENTS OF DEFAULT.

If any of the following events (each, an "Event of Default") shall occur and be continuing:

- (a) If any Debtor fails to pay within three (3) Business Days after the same becomes due and payable, or declared due and payable, all or any portion of the Obligations; provided that any payment of Obligations to be made from the proceeds of any 363 Sale shall be made directly from the applicable purchaser to the DIP Lender concurrently with such 363 Sale (and any failure to so pay such Obligations shall constitute an immediate Event of Default); or
- (b) If any representation, warranty, certification or statement of fact made or deemed made by or on behalf of any Debtor herein, in any other Loan Document or in any document delivered in connection herewith or therewith shall be incorrect or misleading in any material respect when made or deemed made; or

- (c) If any Debtor, or any Subsidiary, fails to perform or observe any covenant or other agreement contained in any of <u>Sections 2.6, 5.1, 5.3, 5.5, 5.12, 5.13, 5.15, 5.16, 5.18, 5.19, 5.21</u> and 6.1 through 6.17 of this Agreement; or
- (d) If Borrower, or any Subsidiary of Borrower, fails to perform or observe any covenant or other agreement contained in this Agreement, or in any of the other Loan Documents in each case, other than any such covenant or agreement that is the subject of another provision of this Section 7 (in which event such other provision of this Section 7 shall govern), and such failure continues for a period of five (5) days after the earlier of (i) the date on which such failure shall first become known to any Financial Officer or Responsible Officer of any Debtor or (ii) written notice thereof is given to Borrower by DIP Lender; or
- (e) If any material portion of any Debtor's or any Subsidiary's assets is attached, seized, subject to a writ or distress warrant, or is levied upon, or comes into the possession of any third Person and the same is not discharged before the earlier of thirty (30) days after the date it first arises or five (5) days prior to the date on which such property or asset is subject to forfeiture; or
- (f) Any Debtor or any Subsidiary shall fail to make any payment (whether of principal or interest and regardless of amount) in respect of any Material Indebtedness incurred Post-Petition when and as the same shall become due and payable (after giving effect to the expiration of any grace or cure period set forth therein); or
- (g) Except for defaults occasioned by the Effect of Bankruptcy and defaults resulting from obligations with respect to which the Bankruptcy Code prohibits a Debtor or a Subsidiary from complying or permits a Debtor or a Subsidiary not to comply, a default, breach or condition occurs under any other agreement, document or instrument entered into Post-Petition, to which the applicable Debtor or Subsidiary, is a party that is not cured within any applicable grace period therefor, and such default or breach results in any Material Indebtedness incurred Post-Petition becoming due prior to its scheduled maturity or that enables or permits (with or without the giving of notice, the lapse of time or both) the holder or holders of any such Material Indebtedness or any trustee or agent on its or their behalf to cause any such Material Indebtedness to become due, or to require the prepayment, repurchase, redemption or defeasance thereof, prior to its scheduled maturity, or cash collateral in respect thereof to be demanded; or
 - (h) A Change of Control shall occur; or
- (i) The determination of any Debtor, whether by vote of such Debtor's Board of Directors or otherwise, to suspend the operation of the business of such Debtor or to liquidate all or substantially all of such Debtor's assets; or
- (j) If any Debtor or any Subsidiary is enjoined, restrained, or in any way prevented by court order from continuing to conduct all or any material part of its business; or
- (k) There is entered against any Debtor or any Subsidiary: (i) a final judgment or order for the payment of money in an aggregate amount exceeding the \$50,000 (to the extent not covered by independent third-party insurance as to which the insurer does not dispute coverage), or (ii) any one or more non-monetary final judgments that have, or could reasonably be

expected to have, individually or in the aggregate, a materially adverse effect on any Debtor or the operation of its business and, in either case: (A) enforcement proceedings are commenced by any creditor upon such judgment or order; or (B) there is a period of thirty (30) consecutive days during which a stay of enforcement of such judgment, by reason of a pending appeal or otherwise, is not in effect; or

- (l) (i) An ERISA Event occurs with respect to a Pension Plan or Multiemployer Plan or (ii) Borrower or any ERISA Affiliate fails to pay when due, after the expiration of any applicable grace period, any installment payment with respect to its withdrawal liability under Section 4201 of ERISA under a Multiemployer Plan; or
- (m) Any provision of any Loan Document shall at any time for any reason be declared to be null and void, or any challenge by or on behalf of any Debtor to the validity of any Loan Document or the enforceability of any Loan Document strictly in accordance with the subject Loan Document's terms or which seeks to void, avoid, limit, or otherwise adversely affect any security interest created by or in any Loan Document or any payment made pursuant thereto; or
- (n) Any challenge by or on behalf of any other Person to the validity of any Loan Document or the enforceability of any Loan Document strictly in accordance with the subject Loan Document's terms or which seeks to void, avoid, limit, or otherwise adversely affect any security interest created by or in any Loan Document or any payment made pursuant thereto, in each case, as to which an order or judgment has been entered adverse to DIP Lender; or
- (o) Any Lien purported to be created under this Agreement or any Security Document shall cease to be, or shall be asserted by any Debtor not to be, a valid and perfected Lien on any Collateral, with the priority required by this Agreement or the applicable Security Document, except as a result of the sale, release or other disposition of the applicable Collateral in a transaction permitted under the Loan Documents or the failure of DIP Lender, through its acts or omissions and through no fault of any Debtor, to maintain the perfection of its Liens in accordance with Applicable Law; or
- (p) The occurrence of any uninsured loss to any portion of the Collateral, which loss could reasonably be expected to have a Material Adverse Change; or
- (q) The imposition of any stay or other order, the effect of which restrains the conduct by the Debtors, taken as a whole, of their business in the ordinary course and that could reasonably be expected to have a Material Adverse Change; or
- (r) The entry of an order in any Chapter 11 Case which amends, supplements, stays, modifies, vacates or reverses any DIP Order or which otherwise materially adversely affects the effectiveness of any DIP Order or the filing of any motion or other pleading requesting such relief that is not timely opposed by the Debtors, in each case, without the express written consent of DIP Lender; or
- (s) (i) The appointment in any Chapter 11 Case of a trustee for any Debtor under Section 1104 of the Bankruptcy Code or (ii) the filing of a motion or other pleading requesting such relief that is not timely opposed by Debtors; or

- (t) (i) The appointment in any Chapter 11 Case of a trustee or of any examiner having expanded powers to operate all or any part of Debtors' business, (ii) the filing of a motion or other pleading requesting such relief that is not timely opposed by Debtors or (iii) the conversion of any Chapter 11 Case to a case under Chapter 7 of the Bankruptcy Code; or
- (u) The entry of any order (other than the DIP Orders) which provides relief from the automatic stay otherwise imposed pursuant to Section 362 of the Bankruptcy Code to any creditor with a claim in excess of \$50,000 (unless otherwise provided in the DIP Order) or which permits any creditor to (i) realize upon, or to exercise any right or remedy with respect to, any Collateral, or (ii) to terminate any license, franchise, or similar agreement, where the exercise of such right or remedy or such realization or termination would reasonably be likely to have a Material Adverse Change; or
- (v) The filing of any application by any Debtor without the express prior written consent of DIP Lender for the approval of any super-priority claim in any Chapter 11 Case which is *pari passu* with or senior to the priority of the claims of DIP Lender for the Obligations, or there shall arise any such super-priority claim under the Bankruptcy Code; or
- (w) The payment or other discharge by any Debtor of any Prepetition Indebtedness, except as expressly permitted hereunder, under the DIP Orders, or in the Approved Budget or by order in any Chapter 11 Case to which order DIP Lender has provided its written consent; or
- (x) The process for the 363 Sale of all of Debtors' assets to the DIP Lender described in the Sale and Bidding Procedures Motion (or, after entered, the Bid Procedures Order) is terminated; or
- (y) Exclusivity between the Debtors and the DIP Lender (or its affiliates) shall have been terminated or any Debtor shall have agreed to any such termination; or
- (z) The Sale Order or the Bid Procedures Order shall cease to be in full force and effect, shall have been reversed, stayed, vacated or subject to stay pending appeal or, without the prior approval of the DIP Lender, shall have been modified or amended; or
- (aa) Any Debtor shall take any action or shall support any other Person in taking any action to restrict or prohibit the DIP Lender (whether in its capacity as the lender under the Prepetition Secured Note or as the lender hereunder) from submitting a "credit bid" for any assets of the Debtors in accordance with Bankruptcy Code 363(k); or
- (bb) Failure by Debtors to disburse the proceeds of the 363 Sale to the DIP Lender (in its capacity as the lender under the Prepetition Secured Note or as the lender hereunder, as applicable) contemporaneously with the closing of the 363 Sale; or
- (cc) The entry of any order in any Chapter 11 Case which provides adequate protection, or the granting by any Debtor of similar relief in favor of any one or more of any Debtor's Prepetition creditors, contrary to the terms and conditions of any DIP Order or the terms hereof; or

- (dd) The failure of Debtors to comply in all material respects with each and all of the terms and conditions of the DIP Orders; or
- (ee) The filing of any motion by any Debtor seeking, or the entry of any order in any Chapter 11 Case: (i) (A) permitting working capital, Post-Petition loans, exit financing, permanent or long-term financing or other financial accommodations for Borrower or any other Debtor from any Person other than DIP Lender, (B) granting a Lien on, or security interest in (other than a Permitted Lien and the Carve-Out) any of the Collateral, other than with respect to this Agreement, (C) except as permitted by this Agreement, permitting the use of any of the Collateral pursuant to Section 363(c) of the Bankruptcy Code without the prior written consent of DIP Lender, (D) permitting recovery from any portion of the Collateral any costs or expenses of preserving or disposing of such Collateral under Section 506(c) of the Bankruptcy Code, or (E) dismissing any Chapter 11 Case or (ii) the filing of any motion by any party in interest or any Creditors' Committee appointed in any Chapter 11 Case seeking any of the matters specified in the foregoing clause (i) that is not dismissed or denied within the later of (x) the soonest available time for a hearing before the Bankruptcy Court seeking the dismissal of such motion, and (y) thirty (30) days of the date of the filing of such motion (or such later date agreed to in writing by DIP Lender); or
- (ff) The filing of any application by any Debtor without the express prior written consent of DIP Lender for an order substituting any assets for all or any portion of the Collateral;
- (gg) The filing of a motion by a Debtor seeking approval of a disclosure statement and a Reorganization Plan, or any action by any Debtor supporting a Reorganization Plan, or the entry of an order confirming a Reorganization Plan, in each case that is not an Acceptable Plan; or
- (hh) (i) The filing of any pleading by any Debtor, or the entry of any order challenging or invalidating the validity, priority, perfection, or enforceability of the Loan Documents, the Obligations, or any Lien granted pursuant to the Loan Documents to secure the Obligations, or (ii) any Lien granted pursuant to the Loan Documents is determined to be null and void, invalid or unenforceable by the Bankruptcy Court or another court of competent jurisdiction in any action commenced or asserted by any other party in interest in any Chapter 11 Case, including, without limitation, the Creditors' Committee; or
- (ii) The filing of any motion, pleading, or proceeding by the Borrower, or the entry of any order, that could reasonably be expected to materially impair the rights or interests of DIP Lender; or
- (jj) Any of the Events of Default (as defined in the DIP Orders) has occurred and is continuing.

8. RIGHTS AND REMEDIES ON DEFAULT.

8.1. **Rights and Remedies**.

- (a) Subject to the terms of the DIP Orders, upon the occurrence, and during the continuation, of an Event of Default, DIP Lender may do any one or more of the following, all of which are authorized by each Debtor:
 - (i) Terminate the Loan Commitment, reduce the Available Commitment Amount to \$0.00, and thereupon the DIP Lender shall no longer have any commitment to make Advances hereunder and such termination of the Loan Commitment shall be irrevocable and immediate;
 - (ii) Declare all or any portion of the Obligations, whether evidenced by this Agreement, by any of the other Loan Documents, or otherwise, due and payable, and thereupon the principal of the Advances and all other Obligations so declared to be due and payable, together with accrued interest thereon and all fees and other obligations of the Debtors accrued under Loan Documents, shall become due and payable immediately, without presentment, demand, protest or other notice of any kind, all of which are hereby waived by the Debtors;

(iii) (Reserved);

- (iv) Terminate this Agreement and any of the other Loan Documents as to any future liability or obligation of DIP Lender, but without affecting any of DIP Lender's Liens in the Collateral and without affecting the Obligations;
- (v) settle or adjust disputes and claims directly with Account Debtors on accounts of Debtors for amounts on terms and in any order that DIP Lender considers advisable, notify any Person owing any Debtor money of DIP Lender's Lien on such funds, and verify the amount of such account. Debtors shall collect all payments in trust for DIP Lender and, if requested by DIP Lender, immediately deliver the payments to DIP Lenders in the form received from the Account Debtor, with proper endorsement for deposit;
- (vi) make any payments and do any acts it considers necessary or reasonable to protect the Collateral and/or its Lien upon the Collateral. During the continuance of an Event of Default, Debtors shall assemble the Collateral if DIP Lender so requests and make it available as DIP Lender so designates. DIP Lender may enter the premises where the Collateral is located, take and maintain possession of any part of the Collateral, and pay, purchase, contest, or compromise any Lien which appears to be prior or superior to DIP Lender's Lien thereon and pay all expenses incurred. Each Debtor grants DIP Lender a license to enter and occupy any of its premises, without charge, to exercise any of DIP Lender's rights or remedies during the continuance of an Event of Default;
- (vii) apply to the Obligations any amount held by DIP Lender owing to or for the credit or the account of any Debtor;

- (viii) ship, reclaim, recover, store, finish, maintain, repair, prepare for sale, advertise for sale, and sell the Collateral. DIP Lender is hereby granted a non-exclusive, royalty-free license or other right to use, during the continuance of an Event of Default, without charge, any Debtor's or any Subsidiaries' labels, patents, copyrights, mask works, rights of use of any name, trade secrets, trade names, trademarks, service marks, and advertising matter, or any similar property as it pertains to the Collateral, in completing production of, advertising for sale, and selling any Collateral and, in connection with DIP Lender's exercise of its rights under this Section, Debtors' and each Subsidiaries' rights under all licenses and all franchise agreements inure to DIP Lender;
- (ix) deliver a notice of exclusive control, any entitlement order, or other directions or instructions pursuant DIP Lender's perfected security interest hereunder and under the DIP Orders in all applicable Collateral (and, to the extent delivered hereunder, pursuant to any Control Agreement);
- (x) demand and receive possession of the Books and Records of Debtors;
 - (xi) set-off any amounts held as cash collateral; and
- (xii) exercise all default rights and remedies available to the DIP Lender under the Loan Documents or at law or equity, including all default remedies under the UCC (including disposal of the collateral (including all Collateral) pursuant to the terms thereof) to obtain the full and indefeasible payment of the Obligations.
- (b) (Reserved).

DIP Lender shall have all other rights and remedies available at law or in equity or pursuant to any other Loan Document or pursuant to the DIP Orders.

None of the Debtors, the Creditors' Committee, or any other party-in-interest shall have the right to contest the enforcement of remedies set forth in the DIP Orders and the Loan Documents on any basis other than an assertion that an Event of Default has not occurred or has been cured within the cure periods expressly set forth in the applicable Loan Documents or the DIP Orders. Debtors shall cooperate fully with the DIP Lenders in their exercise of rights and remedies, whether against the Collateral or otherwise. Each Debtor hereby waives any right to seek relief under the Bankruptcy Code, including under Section 105 thereof, to the extent such relief would restrict or impair the rights and remedies of the DIP Lender set forth in the DIP Orders and in the Loan Documents.

Subject to the DIP Orders, in case any one or more of the covenants and/or agreements set forth in this Agreement or any other Loan Document or in the DIP Orders shall have been breached by any Debtor, then the DIP Lender may proceed to protect and enforce its rights either by suit in equity and/or by action at law, including an action for damages as a result of any such breach and/or an action for specific performance of any such covenant or agreement contained in this Agreement or such other Loan Document. Without limitation of the foregoing, each Debtor agrees that failure to comply with any of the covenants contained herein will cause irreparable harm and that specific

performance shall be available in the event of any breach thereof. The DIP Lender shall be indemnified by Debtors against all liability, loss or damage, together with all reasonable costs and expenses related thereto (including reasonable legal and accounting fees and expenses) in accordance with the terms hereof.

8.2. **Power of Attorney**. Each Debtor hereby irrevocably appoints DIP Lender as its lawful attorney-in-fact, to take any of the following actions during the continuance of an Event of Default: (a) if any Debtor refuses to, or fails timely to execute and deliver any of the documents required to be delivered by it pursuant to the terms hereof, sign the name of the applicable Debtor on any of such documents, (b) endorse the applicable Debtor's name on any checks or other forms of payment or security, sign the applicable Debtor's name on any invoice or bill of lading for any account or drafts against Account Debtors or sign the applicable Debtor's name on any notices to Account Debtors, (c) send requests for verification of Accounts, (d) endorse the applicable Debtor's name on any collection item that may come into DIP Lender's possession, (e) at any time that an Event of Default has occurred and is continuing, make, settle, and adjust all claims under Debtors' policies of insurance and make all determinations and decisions with respect to such policies of insurance, (f) at any time that an Event of Default has occurred and is continuing, take control, in any manner, of any item of payment or proceeds relating to any Collateral, (g) at any time that an Event of Default has occurred and is continuing, prepare, file, and sign any Debtor's name to a proof of claim in bankruptcy or similar document against any Account Debtor, or to any notice of lien, assignment, or satisfaction of lien or similar document in connection with any of the Collateral, (h) at any time that an Event of Default has occurred and is continuing, receive, open and dispose of all mail addressed to any Debtor, and notify postal authorities to change the address for delivery thereof to such address as DIP Lender may designate, (i) at any time that an Event of Default has occurred and is continuing, use the information recorded on or contained in any data processing equipment, computer hardware, and software relating to the Collateral, (j) at any time that an Event of Default has occurred and is continuing, settle and adjust disputes and claims respecting the Accounts, Chattel Paper or General Intangibles directly with Account Debtors, for amounts and upon terms that DIP Lender determines to be reasonable, and DIP Lender may cause to be executed and delivered any documents and releases that DIP Lender determines to be necessary, (k) at any time file UCC-3 assignments reflecting DIP Lender as assignee of any Debtor with respect to any UCC-1 financing statements filed by Debtors in connection with Collateral, (1) at any time that an Event of Default has occurred and is continuing, cause an Account Debtor's insurers to add DIP Lender as loss payee under the relevant insurance policy, (m) at any time that an Event of Default has occurred and is continuing, pay, contest or settle any Lien, charge or adverse claim in, to or upon any or all of the Collateral, or any judgment based thereon, or otherwise take any action to terminate or discharge the same, (n) at any time that an Event of Default has occurred and is continuing, transfer any Collateral into the name of DIP Lender or a third party as the Uniform Commercial Code permits, and (o) do all other acts and things necessary, in DIP Lender's determination, to fulfill each Debtor's obligations under this Agreement. Each Debtor hereby appoints DIP Lender as its lawful attorney-in-fact to sign such Debtor's name on any documents necessary to perfect or continue the perfection of any security interest in the Collateral regardless of whether an Event of Default has occurred until all Obligations have been Paid in Full. DIP Lender's foregoing appointment as the attorney-in-fact for each Debtor, and all of DIP Lender's rights and powers, being coupled with an interest, are irrevocable until all Obligations have been Paid in Full and performed when due (as applicable).

8.3. **Remedies Cumulative**. The rights and remedies of DIP Lender under this Agreement, the other Loan Documents, and all other agreements shall be cumulative. DIP Lender shall have all other rights and remedies not inconsistent herewith as provided under the UCC, by law, or in equity. No exercise by DIP Lender of one right or remedy shall be deemed an election, and no waiver by DIP Lender of any Event of Default shall be deemed a continuing waiver. No delay by DIP Lender shall constitute a waiver, election, or acquiescence by it.

9. APPLICATION OF PROCEEDS ON DEFAULT.

After the occurrence of both an Event of Default and the commencement of an Enforcement Action relating to such Event of Default, subject to the Carve-Out, any and all proceeds of Collateral shall be applied in the following order:

- (a) <u>first</u>, to pay any reasonable and documented costs and expenses (including reasonable and documented attorneys' fees) incurred by DIP Lender or any DIP Lender in connection with the Agreement, any Enforcement Action or any Loan Document;
- (b) <u>second</u>, to pay, first to accrued interest arising hereunder that has not yet been capitalized, and then to the Principal Amount (including interest paid in kind), until Paid in Full;
 - (c) third, to pay the remaining Obligations; and
- (d) <u>last</u>, the balance, if any, after all of the Obligations have been Paid in Full, to the Borrower or as otherwise directed by the Bankruptcy Court or any applicable Final Order.

10. TAXES AND EXPENSES.

10.1. **Taxes**.

(a) DIP Lender shall provide to Borrower on or before the Effective Date a validly executed IRS Form W-9. All payments of principal and interest on the Advances and all other amounts payable hereunder shall be made free and clear of and without deduction for any present or future income, gross receipts, excise, stamp, documentary, sales, use, payroll, employment, property, branch profits or franchise taxes and all other taxes, fees, duties, levies, assessments, withholdings (including backup withholdings) and other charges of any nature whatsoever (including interest, penalties and additions thereon) imposed by any Governmental Authority, excluding taxes imposed on or measured by DIP Lender's net income by the jurisdiction under which DIP Lender is organized or conducts business (other than solely as the result of entering into any of the Loan Documents or taking any action thereunder) (all non-excluded items being called a "Tax" or "Taxes"). If any withholding or deduction from any payment to be made by Borrower hereunder is required in respect of any Taxes pursuant to any Applicable Law, then Borrower will: (i) pay directly to the relevant authority the full amount required to be so withheld or deducted; (ii) promptly forward to DIP Lender an official receipt or other documentation satisfactory to DIP Lender evidencing such payment to such authority; and (iii) pay to DIP Lender for the account of DIP Lender such additional amount or amounts as is necessary to ensure that the net amount actually received by DIP Lender (as reduced by all deductions and withholdings, including deductions and withholdings on such additional payments) will equal the full amount

DIP Lender would have received had no such withholding or deduction been required. If any Taxes are directly asserted against DIP Lender with respect to any payment received by DIP Lender hereunder, DIP Lender may pay such Taxes and Borrower will promptly pay to DIP Lender such additional amounts (including any penalty, interest or expense) as is necessary in order that the net amount received by DIP Lender after the payment of such Taxes (including any Taxes on such additional payments) shall equal the amount DIP Lender would have received had such Taxes not been asserted.

(b) If Borrower fails to pay any Taxes when due to the appropriate taxing authority or fails to remit to DIP Lender, for the account of DIP Lender, the required receipts or other required documentary evidence, Borrower shall indemnify DIP Lender for any incremental Taxes (including any penalty, interest or expense) that may become payable by DIP Lender as a result of any such failure.

10.2. **Reserved**.

10.3. Payment of other Amounts. If Borrower fails to pay any monies (whether Taxes, assessments, insurance premiums, or, in the case of leased properties or assets, rents or other amounts payable under such leases) due to third Persons, or fails to make any deposits or furnish any required proof of payment or deposit, all as required under the terms of this Agreement, then, DIP Lender, in its sole discretion and without prior notice to Borrower, may do any or all of the following: (a) make payment of the same or any part thereof, (b) designate a portion of the Available Commitment Amount as reserves to the extent DIP Lender deems necessary to protect it from the exposure created by such failure, or (c) in the case of the failure to comply with Section 5.8 hereof, obtain and maintain insurance policies of the type described in Section 5.8 and take any action with respect to such policies as DIP Lender deems prudent. Any such amounts paid by DIP Lender shall constitute Advances and any such payments shall not constitute an agreement by DIP Lender to make similar payments in the future or a waiver by DIP Lender of any Event of Default under this Agreement. DIP Lender need not inquire as to, or contest the validity of, any such expense, Tax, or Lien and the receipt of the usual official notice for the payment thereof shall be conclusive evidence that the same was validly due and owing.

11. WAIVERS; INDEMNIFICATION.

- 11.1. <u>Demand; Protest, etc</u>. Each Debtor waives demand, protest, notice of protest, notice of default or dishonor, notice of payment and nonpayment, nonpayment at maturity, release, compromise, settlement, extension, or renewal of documents, instruments, chattel paper, and guarantees at any time held by DIP Lender on which Borrower may in any way be liable.
- as DIP Lender's Liability for Collateral. Borrower hereby agrees that: (a) so long as DIP Lender complies with its obligations, if any, under the DIP Orders, DIP Lender shall not in any way or manner be liable or responsible for: (i) the safekeeping of the Collateral, (ii) any loss or damage thereto occurring or arising in any manner or fashion from any cause, (iii) any diminution in the value thereof, or (iv) any act or default of any carrier, warehouseman, bailee, forwarding agency, or other Person, and (b) all risk of loss, damage, or destruction of the Collateral shall be borne by Borrower.

11.3. **Indemnification**.

- Borrower shall pay, indemnify, defend, and hold the DIP Lender-Related Persons (each, an "Indemnified Person") harmless (to the fullest extent permitted by Applicable Law) from and against any and all claims, demands, suits, actions, investigations, proceedings, liabilities, fees, losses, penalties and damages, and all attorneys', experts' and consultants' fees and disbursements and other costs and expenses actually incurred in connection therewith or in connection with the enforcement of this indemnification (as and when they are incurred and irrespective of whether suit is brought), at any time asserted against, imposed upon, or incurred by any of them (a) in connection with, or as a result of, or related to the execution, delivery, enforcement, performance, or administration (including any restructuring or workout with respect hereto) of this Agreement, any of the other Loan Documents, or the transactions contemplated hereby or thereby, any amendment hereto or thereto, or the monitoring of Borrower's and its Subsidiaries' compliance with the terms of the Loan Documents, and (b) with respect to any investigation, litigation, or proceeding related to this Agreement, any other Loan Document, or the use of the proceeds of the credit provided hereunder (irrespective of whether any Indemnified Person is a party thereto), or any act, omission, event, or circumstance in any manner related thereto (all the foregoing, collectively, the "Indemnified Liabilities"). The foregoing to the contrary notwithstanding, Borrower shall have no obligation to any Indemnified Person under this Section 11.3 with respect to any Indemnified Liability that a court of competent jurisdiction finally determines to have resulted from the gross negligence or willful misconduct of such Indemnified Person. This provision shall survive the termination of this Agreement, the repayment of the Obligations and any assignment by a DIP Lender pursuant to Section 14 (it being understood and agreed that the Borrower's obligations under this provision shall continue to apply to any such assigning DIP Lender with respect to any Indemnified Liabilities that arise from or relate to the period prior to the effectiveness of such assignment). If any Indemnified Person makes any payment to any other Indemnified Person with respect to an Indemnified Liability as to which Borrower is required to indemnify the Indemnified Person receiving such payment, the Indemnified Person making such payment is entitled to be indemnified and reimbursed by Borrower with respect thereto. WITHOUT LIMITATION, THE FOREGOING INDEMNITY SHALL APPLY TO EACH INDEMNIFIED PERSON WITH RESPECT TO INDEMNIFIED LIABILITIES WHICH IN WHOLE OR IN PART ARE CAUSED BY OR ARISE OUT OF ANY NEGLIGENT ACT OR OMISSION OF SUCH INDEMNIFIED PERSON OR OF ANY OTHER PERSON.
- (b) To the fullest extent permitted by Applicable Law, Borrower shall not assert, and Borrower hereby waives, any claim against any Indemnified Person by Borrower on any theory of liability, for special, indirect, consequential, or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement, any other Loan Document, or any agreement or instrument contemplated hereby, the transactions contemplated hereby or thereby, any Advance, or the use of the proceeds thereof. No Indemnified Person shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed by it through telecommunications, electronic, or other information transmission systems in connection with this Agreement or the other Loan Documents or the transactions contemplated hereby or thereby.

(c) All amounts due under this <u>Section 11.3</u> shall be payable immediately after written demand therefor. The agreements in this <u>Section 11.3</u> shall survive Payment in Full of the Obligations, payment of all other amounts payable hereunder, and the termination of this Agreement.

12. NOTICES.

Unless otherwise provided in this Agreement or the DIP Orders, all notices or demands by Debtors or DIP Lender to the other relating to this Agreement or any other Loan Document shall be in writing and (except for financial statements and other informational documents which may be sent by first-class mail, postage prepaid) shall be personally delivered or sent by registered or certified mail (postage prepaid, return receipt requested), overnight courier, electronic mail (at such email addresses as the applicable Debtor or DIP Lender, as applicable, may designate to each other in accordance herewith), or be in an electronic writing sent by electronic mail to Borrower or to the applicable Guarantor in care of Borrower or to DIP Lender, as the case may be, at its address set forth below:

If to Borrower: OTB Acquisition LLC

c/o OTB Holding LLC 3424 Peachtree Road.

Suite 1500

Atlanta, GA 30326

Attn: Jonathan Tibus, Chief Restructuring Officer

E-Mail: jtibus@alvarezandmarsal.com

with a copy to: King & Spalding LLP

1180 Peachtree Street, NE

Suite 1600

Atlanta, GA 30309 Attn: Jeff Dutson

E-Mail: jdutson@kslaw.com

If to DIP Lender: OTB Lender, LLC

c/o Pappas Restaurants Inc.

13939 NW Freeway

Houston, Texas 77040-5115

Eric Swanson and Anna Marchand E-Mail: eswanson@pappas.com and

amarchand@pappas.com

with a copy to: Porter Hedges LLP

1000 Main Street

36th Floor

Houston, TX 77002 Attn: Joshua Wolfshohl Fax No.: (713) 226-6295

E-Mail: jwolfshohl@porterhedges.com

DIP Lender and Debtors may change the address at which they are to receive notices hereunder, by notice in writing in the foregoing manner given to the other parties hereto. All notices or demands sent in accordance with this Section 12, other than notices by DIP Lender in connection with enforcement rights against the Collateral under the provisions of the UCC, shall be deemed received on the earlier of the date of actual receipt or three (3) Business Days after the deposit thereof in the mail (or, with respect to electronic mail, deemed received at the close of business on the Business Day such message was sent, unless an error message or non-delivery notification is received by the sender of such communication). Each Debtor acknowledges and agrees that notices sent by DIP Lender in connection with the exercise of enforcement rights against Collateral under the provisions of the UCC shall be deemed sent when deposited in the mail or personally delivered, or, where permitted by law, transmitted by electronic mail or any other method set forth above.

13. CHOICE OF LAW AND VENUE; JURY TRIAL WAIVER.

- (a) THE VALIDITY OF THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS (UNLESS EXPRESSLY PROVIDED TO THE CONTRARY IN ANOTHER LOAN DOCUMENT IN RESPECT OF SUCH OTHER LOAN DOCUMENT), THE CONSTRUCTION, INTERPRETATION, AND ENFORCEMENT HEREOF AND THEREOF, AND THE RIGHTS OF THE PARTIES HERETO AND THERETO WITH RESPECT TO ALL MATTERS ARISING HEREUNDER OR THEREUNDER OR RELATED HERETO OR THERETO SHALL BE DETERMINED UNDER, GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK, WITHOUT GIVING EFFECT TO THE CONFLICTS OF LAWS PRINCIPLES THEREOF, AND THE BANKRUPTCY CODE.
- (b) THE PARTIES AGREE THAT ALL ACTIONS OR PROCEEDINGS ARISING IN CONNECTION WITH THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS SHALL BE TRIED AND LITIGATED ONLY IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF GEORGIA, AND THE PARTIES CONSENT TO THE NON-EXCLUSIVE JURISDICTION OF SUCH COURTS. EACH DEBTOR AND DIP LENDER WAIVE, TO THE EXTENT PERMITTED UNDER APPLICABLE LAW, ANY RIGHT EACH MAY HAVE TO ASSERT THE DOCTRINE OF FORUM NON CONVENIENS OR TO OBJECT TO VENUE TO THE EXTENT ANY PROCEEDING IS BROUGHT IN ACCORDANCE WITH THIS SECTION 13(b).
- (c) EACH DEBTOR AND DIP LENDER HEREBY WAIVE THEIR RESPECTIVE RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF ANY OF THE LOAN DOCUMENTS OR ANY OF THE TRANSACTIONS CONTEMPLATED THEREIN, INCLUDING CONTRACT CLAIMS, TORT CLAIMS, BREACH OF DUTY CLAIMS, AND ALL OTHER COMMON LAW OR STATUTORY CLAIMS. EACH DEBTOR AND DIP LENDER REPRESENT THAT EACH HAS REVIEWED THIS WAIVER AND EACH KNOWINGLY AND VOLUNTARILY WAIVES ITS JURY TRIAL RIGHTS FOLLOWING CONSULTATION WITH LEGAL COUNSEL. IN THE EVENT OF LITIGATION, A COPY OF THIS

AGREEMENT MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.

14. ASSIGNMENTS; SUCCESSORS.

- 14.1. <u>Successors and Assigns Generally</u>. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that no Debtor may assign or otherwise transfer or delegate any of its rights or obligations hereunder without the prior written consent of DIP Lender. Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in <u>Section 14.3</u> and, to the extent expressly contemplated hereby, the DIP Lender-Related Persons) any legal or equitable right, remedy, or claim under or by reason of this Agreement.
- 14.2. <u>Assignments by DIP Lender</u>. DIP Lender may at any time assign to one or more assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of the Advances at the time owing to it) without the prior written consent of any Debtor.
- 14.3. **Participations**. DIP Lender may at any time, without the consent of, or notice to, any Debtor, sell participations to any Person (each, a "<u>Participant</u>") in all or a portion of DIP Lender's rights or obligations under this Agreement (including all or a portion of the Advances owing to it); <u>provided</u> that (a) DIP Lender's obligations under this Agreement shall remain unchanged, (b) DIP Lender shall remain solely responsible to the other parties hereto for the performance of such obligations, and (c) Borrower shall continue to deal solely and directly with DIP Lender in connection with the rights and obligations under this Agreement.

15. AMENDMENTS; WAIVERS.

- 15.1. <u>Amendments and Waivers</u>. No amendment or waiver of any provision of this Agreement or any other Loan Document, and no consent with respect to any departure by any Debtor therefrom, shall be effective unless the same shall (a) be in writing and signed by DIP Lender and Borrower; provided that if such amendment modifies a provision giving rise to obligations of a Guarantor, then such amendment shall also be signed by the Guarantors and (b) specifically state that it is intended to amend this Agreement or such other Loan Document or waive the right hereunder or thereunder, as applicable, and then any such waiver or consent shall be effective, but only in the specific instance and for the specific purpose for which given.
- 15.2. No Waivers; Cumulative Remedies. No failure by DIP Lender to exercise any right, remedy, or option under this Agreement or any other Loan Document, or delay by DIP Lender in exercising the same, will operate as a waiver thereof. No waiver by DIP Lender will be effective unless it is in writing, and then only to the extent specifically stated. No waiver by DIP Lender on any occasion shall affect or diminish DIP Lender's rights thereafter to require strict performance by Debtors of any provision of this Agreement. DIP Lender's rights under this Agreement and the other Loan Documents will be cumulative and not exclusive of any other right or remedy that DIP Lender may have.

16. GENERAL PROVISIONS.

- 16.1. <u>Section Headings</u>. Headings and numbers have been set forth herein for convenience only. Unless the contrary is compelled by the context, everything contained in each Section applies equally to this entire Agreement.
- 16.2. <u>Interpretation</u>. Neither this Agreement nor any uncertainty or ambiguity herein shall be construed against DIP Lender or any Debtor, whether under any rule of construction or otherwise. On the contrary, this Agreement has been reviewed by all parties and shall be construed and interpreted according to the ordinary meaning of the words used so as to accomplish fairly the purposes and intentions of all parties hereto.
- 16.3. <u>Severability of Provisions</u>. Each provision of this Agreement shall be severable from every other provision of this Agreement for the purpose of determining the legal enforceability of any specific provision.
- 16.4. <u>Counterparts; Electronic Execution</u>. This Agreement may be executed in any number of counterparts and by different parties on separate counterparts, each of which, when executed and delivered, shall be deemed to be an original, and all of which, when taken together, shall constitute but one and the same Agreement. Delivery of an executed counterpart of this Agreement by any electronic method of transmission shall be equally as effective as delivery of an original executed counterpart of this Agreement. The failure by any party to deliver a manually executed counterpart of this Agreement shall not affect the validity, enforceability, and binding effect of this Agreement. The foregoing shall apply to each other Loan Document *mutatis mutandis*.
- 16.5. Revival and Reinstatement of Obligations. If the incurrence or payment of the Obligations by Borrower (or the guarantee of the Obligations by any Guarantor) or the transfer to DIP Lender of any property should for any reason subsequently be declared to be void or voidable under any state or federal law relating to creditors' rights, including provisions of the Bankruptcy Code relating to fraudulent conveyances, preferences, or other voidable or recoverable payments of money or transfers of property (collectively, a "Voidable Transfer"), and if DIP Lender is required to repay or restore, in whole or in part, any such Voidable Transfer, or elects to do so upon the reasonable advice of its counsel, then, as to any such Voidable Transfer, or the amount thereof that DIP Lender is required or elects to repay or restore, and as to all reasonable costs, expenses, and attorneys fees of DIP Lender related thereto, the liability of the applicable Debtor automatically shall be revived, reinstated, and restored and shall exist as though such Voidable Transfer had never been made.

16.6. **<u>Reserved</u>**.

- 16.7. <u>Integration</u>. This Agreement, together with the other Loan Documents, reflects the entire understanding of the parties with respect to the transactions contemplated hereby and shall not be contradicted or qualified by any other agreement, oral or written, before the date hereof.
- 16.8. <u>Marshaling</u>. DIP Lender shall not be required to marshal any present or future collateral security (including but not limited to the Collateral) for, or other assurances of payment

of, the Obligations or any of them or to resort to such collateral security or other assurances of payment in any particular order, and all rights and remedies of DIP Lender under the Loan Documents or otherwise in respect of such collateral security and/or other assurances of payment shall be cumulative and in addition to all other rights and remedies, however existing or arising. To the extent that it lawfully may, Borrower hereby agrees that it will not invoke any law relating to the marshaling of collateral which might cause delay in or impede the enforcement of DIP Lender's rights and remedies under this Agreement or any other Loan Document or under any other instrument creating or evidencing any of the Obligations or under which any of the Obligations is outstanding or by which any of the Obligations is secured or payment thereof is otherwise assured, and, to the extent that it lawfully may, Borrower hereby irrevocably waives the benefits of all such laws.

- 16.9. <u>Press Releases and Related Matters</u>. Borrower consents to the publication by DIP Lender of customary advertising material relating to the transactions contemplated by this Agreement and the other Loan Documents using such party's name, product photographs, logo, or trademark.
- 16.10. **USA PATRIOT Act Notice**. DIP Lender hereby notifies Borrower that pursuant to the requirements of the Patriot Act it is, or may be, required to obtain, verify, and record information that identifies Borrower, which information includes the name and address of Borrower and other information that will allow the DIP Lender to identify Borrower in accordance with the Patriot Act.
- 16.11. No Advisory or Fiduciary Responsibility. In connection with all aspects of each transaction contemplated hereby, each Debtor acknowledges and agrees, and acknowledges its Affiliates' understanding, that: (a) the credit facilities provided for hereunder and any related arranging or other services in connection therewith (including in connection with any amendment, waiver, or other modification hereof or of any other Loan Document) are an arm's-length commercial transaction between Borrower and its Affiliates, on the one hand, and DIP Lender, on the other hand, and each Debtor is capable of evaluating and understanding and understands and accepts the terms, risks, and conditions of the transactions contemplated hereby and by the other Loan Documents (including any amendment, waiver, or other modification hereof or thereof), (b) DIP Lender has not assumed and will not assume an advisory, agency, or fiduciary responsibility in favor of Borrower or any Affiliate of Borrower with respect to any of the transactions contemplated hereby or the process leading thereto, including with respect to any amendment, waiver, or other modification hereof or of any other Loan Document, and DIP Lender shall not have any obligation to Borrower or any Affiliate of the Borrower with respect to the transactions contemplated hereby except those obligations set forth herein and in the other Loan Documents, (c) DIP Lender and its Affiliates may be engaged in a broad range of transactions that involve interests that differ from those of Borrower and the Affiliates of Borrower, and DIP Lender shall not have any obligation to disclose any of such interests by virtue of any advisory, agency, or fiduciary relationship, and (d) DIP Lender has not provided and will not provide any legal, accounting, regulatory, or tax advice with respect to any of the transactions contemplated hereby (including any amendment, waiver, or other modification hereof or of any other Loan Document), and each Debtor has consulted its own legal, accounting, regulatory, and tax advisors to the extent it has deemed appropriate. Each Debtor hereby waives and releases, to the fullest extent permitted

by law, any claims that it may have against DIP Lender with respect to any breach or alleged breach of agency or fiduciary duty.

16.12. **Relationship with DIP Orders**. In the event of any inconsistency between the terms of the DIP Orders and the Loan Documents, the terms of the DIP Orders shall control.

[Signatures are on the following pages.]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed and delivered as of the date first above written.

BORROWER:

OTB ACQUISITION LLC

Name: Jonathan Tibus

Title: Chief Restructuring Officer

GUARANTORS:

OTB HOLDING LLC
OTB ACQUISITION OF NEW JERSEY LLC
MT. LAUREL RESTAURANT OPERATIONS LLC
OTB ACQUISITION OF HOWARD COUNTY LLC
OTB ACQUISITION OF BALTIMORE COUNTY, LLC
OTB ACQUISITION OF KANSAS LLC

Name: Jonathan Tibus

Title: Chief Restructuring Officer

DIP LENDER:

OTB LENDER, LLC

as Payee

Name: Chris Pappas

Title: Authorized Representative

SCHEDULE 1.1

DEFINITIONS

As used in the Agreement, the following terms shall have the following definitions:

- "363 Sale" means sale of all or substantially all of the assets of the Debtors pursuant to Section 363 of the Bankruptcy Code.
- "<u>Acceptable Plan</u>" means a Reorganization Plan (and related disclosure statement) that provides for the indefeasible payment in full and in cash of the Obligations and is otherwise in form and substance satisfactory to the DIP Lender.
 - "Account" means an account (as that term is defined in the UCC).
- "Account Debtor" means any Person who is obligated on an Account, chattel paper, or a general intangible.
- "Advance" and "Advances" means each cash advance or deemed issuance of a Loan pursuant to Section 2.1.
- "<u>Affiliate</u>" means, as applied to any Person, any other Person who Controls, is Controlled by, or is under common Control with, such Person.
- "<u>Agreement</u>" means this Secured Super-Priority Debtor-In-Possession Credit Agreement and all Schedules and Exhibits, as modified, amended, supplemented or restated, and in effect from time to time.
- "Applicable Law" means as to any Person: (a) all laws, statutes, rules, regulations, orders, codes, ordinances or other requirements having the force of law, and (b) all court orders, decrees, judgments, injunctions, notices, binding agreements and/or rulings, in each case of or by any Governmental Authority which has jurisdiction over such Person, or any property of such Person.
- "Approved Budget" means Debtors' statement of sources and uses of cash commencing on the Petition Date and projected through the Maturity Date, broken down by week and including the anticipated uses of the Advances for each week during such period, in the form annexed to the Interim DIP Order, as the same shall be modified or supplemented with the approval of DIP Lender from time to time. The initial Approved Budget and any subsequent modifications or supplements to the initial Approved Budget shall be in form and substance satisfactory to DIP Lender, together with such supporting documentation as may be reasonably requested by DIP Lender.
- "Available Commitment Amount" means, on any date of determination, the lesser of (a) the amount required to make the payments required by the Approved Budget to be made during the applicable Budget Period, less cash on hand and (b) the Loan Commitment Amount, less the Prepetition Bridge Loan Amount, less the aggregate principal amount of all Advances that have been extended (in cash, by net funding or deemed issuance) hereunder pursuant to the DIP Orders; provided that any interest, fees, or expenses that has been added to the Principal Amount hereunder

shall be excluded from the calculation of the Principal Amount and therefore have no impact on the Available Commitment Amount.

"Availability Period" means period beginning on the Effective Date and ending on the Termination Date.

"<u>Bankruptcy Code</u>" means title 11 of the United States Code, as in effect from time to time, or any successor thereto.

"Bankruptcy Court" has the meaning provided in the recitals to this Agreement.

"Benefit Plan" means any "employee benefit plan" (as such term is defined in Section 3(3) of ERISA) established, maintained or contributed to by any Debtor, Debtor Subsidiary or any of their ERISA Affiliates for the benefit of any current or former employee or service provider of any Debtor or any Debtor Subsidiary or which is established, maintained contributed to by any Debtor, Debtor Subsidiary or any of their ERISA Affiliates or to which any of the foregoing has or could have any Liability and which is subject to Section 412 of the Code, Title IV of ERISA, Section 302 of ERISA or under the Consolidated Omnibus Budget Reconciliation of 1985, as amended, and any successor thereto and applicable rulings and regulations thereunder "COBRA."

"Bid Procedures Order" is defined in Section 5.18(c)(iv).

"Board of Directors" means the board of directors of the Debtors or any committee thereof duly authorized to act on behalf of the board of directors.

"Borrower" has the meaning specified therefor in the preamble to the Agreement.

"Borrower Designated Account" means the Deposit Account #xxxxxx9889 of Borrower maintained with the Borrower Designated Account Bank.

"Borrower Designated Account Bank" means CrossFirst Bank.

"Borrowing" means a borrowing hereunder consisting of Advances made on the same day by DIP Lender to Borrower.

"Budget Period" means (a) with respect to the Interim Advance, the period beginning on the Petition Date and ending twenty-eight (28) days after the Petition Date; provided that with respect to the amounts required to fund the Professional Fee Reserve (as defined in the DIP Order), the term "Budget Period" shall refer to the period beginning on the Petition Date through the consummation of the 363 Sale of all of Debtors' assets to the DIP Lender and (b) with respect to the Advance anticipated to be requested upon the entry of the Final DIP Order, the period beginning on the date of such Advance through the consummation of the 363 Sale of all of Debtors' assets to the DIP Lender; provided that if the DIP Lender agrees to make more than two (2) Advances hereunder, the Budget Period with respect to any Advance on or after entry of the Final DIP Order shall be for the increments of time as agreed by DIP Lender.

"Business Day" means any day that is not a Saturday, Sunday, or other day on which banks are authorized or required to close in the State of Georgia or the State of Texas.

"Capital Expenditures" means expenditures made or liabilities incurred for the acquisition of any fixed assets or improvements, replacements, substitutions or additions thereto which have a useful life of more than one year, which, in accordance with GAAP, would be classified as capital expenditures.

"Capital Lease" means a lease that is required to be capitalized for financial reporting purposes in accordance with GAAP.

"Capital Stock" shall mean, as to any Person that is a corporation, the authorized shares of such Person's capital stock, including all classes of common, preferred, voting and nonvoting capital stock, and, as to any Person that is not a corporation or an individual, the membership or other ownership interests in such Person, including, without limitation, the right to share in profits and losses, the right to receive distributions of cash and other property, and the right to receive allocations of items of income, gain, loss, deduction and credit and similar items from such Person, whether or not such interests include voting or similar rights entitling the holder thereof to exercise control over such Person, collectively with, in any such case, all warrants, options and other rights to purchase or otherwise acquire, and all other instruments convertible into or exchangeable for, any of the foregoing.

"Carve-Out" shall have the meaning set forth in the DIP Orders.

"Cash Equivalents" means (a) marketable direct obligations issued by, or unconditionally guaranteed by, the United States or issued by any agency thereof and backed by the full faith and credit of the United States, in each case maturing within 1 year from the date of acquisition thereof, (b) marketable direct obligations issued by any state of the United States or any political subdivision of any such state or any public instrumentality thereof maturing within 1 year from the date of acquisition thereof and, at the time of acquisition, having one of the two highest ratings obtainable from either Standard & Poor's Rating Services, a division of The McGraw Hill Companies, Inc. ("S&P") or Moody's Investors Service, Inc. ("Moody's"), -(c) commercial paper maturing no more than 270 days from the date of creation thereof and, at the time of acquisition, having a rating of at least A-1 from S&P or at least P-1 from Moody's, (d) Deposit Accounts maintained with any bank organized under the laws of the United States or any state thereof so long as the amount maintained with any such other bank is less than or equal to \$250,000 and is insured by the Federal Deposit Insurance Corporation, and (e) Investments in money market funds substantially all of whose assets are invested in the types of assets described in clauses (a) through (d) above.

"Casualty Event" means the damage, destruction or condemnation, including by process of eminent domain or any transfer or disposition of property in lieu of condemnation, as the case may be, of property of any Person or any of its Subsidiaries, including by process of eminent domain or any transfer or disposition of property in lieu of condemnation.

"CERCLA" means the Comprehensive Environmental Response, Compensation, and Liability Act, 42 U.S.C. § 9601 et seq.

"Change of Control" means that (a) Border Holdings LLC fails to own and control directly, 100% of the Capital Stock of Holding, (b) Holding fails to own and control directly, 100% of the

Capital Stock of Borrower, (c) Borrower fails to own and control directly, 100% of each other Guarantor, (c) Jonathan Tibus shall cease to be the chief restructuring officer of the Debtors without the appointment of a successor that is acceptable to the DIP Lender, (d) the management company of any Debtor shall cease to function substantially in the same manner as it did prior to the Petition Date, (e) there shall be a material change in the Board of Directors or other governing body of any Debtor since the Petition Date or (f) a sale of all or substantially all of the assets of any Debtor.

"Chapter 11 Cases" has the meaning provided in the recitals to this Agreement.

"Collateral" has the meaning specified therefor in Section 2.8(a).

"Collateral Access Agreement" means a landlord waiver, bailee letter, or acknowledgement agreement of any lessor, warehouseman, processor, consignee, or other Person in possession of, having a Lien upon, or having rights or interests in any Debtor's or any Subsidiaries' books and records, Equipment or Inventory. DIP Lender shall have the right to request any such Collateral Access Agreement with respect to any leases that are not rejected by Debtors pursuant to an order of the Bankruptcy Court.

"Collateral Accounts" means all commodity accounts, deposit accounts and securities accounts (in each case, as defined in the UCC) of Debtors, including the Borrower Designated Account and shall exclude the Excluded Accounts.

"Collections" means *all* cash, checks, notes, instruments, and other items of payment (including insurance proceeds, proceeds of cash sales, rental proceeds, and Tax refunds).

"Control" means the possession, directly or indirectly, of the power (a) to vote at least a majority of the securities having ordinary voting power for the election of directors (or any similar governing body) of a Person, or (b) to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. The terms "Controlling" and "Controlled" have meanings correlative thereto.

"Control Agreement" means any agreement entered into among a depository institution at which a Debtor maintains a Collateral Account, the applicable Debtor and DIP Lender, that purports to grant DIP Lender control (within the meaning of the UCC) over such Collateral Account.

"Creditors' Committee" means any official committee of unsecured creditors formed, appointed or approved in any Chapter 11 Case pursuant to the Bankruptcy Code.

"CrossFirst Liens" is defined in Section 2.8(b)(ii).

"Debtor" is defined in the preamble.

"<u>Default</u>" means an event, condition, or default that, with the giving of notice, the passage of time, or both, would be an Event of Default.

"Default Rate" means a per annum rate of interest equal to 15.00%.

"Deposit Account" means any deposit account (as that term is defined in the UCC).

"<u>DIP Lender</u>" has the meaning specified therefor in the preamble to the Agreement, and shall include each other Person from time to time party hereto as a DIP Lender, and any other Person made a party to the Agreement in accordance with the provisions of <u>Section 14</u>.

"<u>DIP Lender Designated Account</u>" means any account designated by DIP Lender into which the payment or proceeds of Collateral shall be made.

"DIP Lender Expenses" means all (a) reasonable and documented costs or expenses (including Taxes, and insurance premiums) required to be paid by Debtors or their Subsidiaries under any of the Loan Documents that are paid, advanced, or incurred by DIP Lender, (b) reasonable and documented fees or charges paid or incurred by DIP Lender in connection with DIP Lender's transactions with Debtors, including, fees or charges for photocopying, notarization, couriers and messengers, telecommunication, public record searches (including Tax liens, litigation, and UCC searches and including searches with the patent and trademark office, the copyright office, or the department of motor vehicles), other due diligence, filing, recording, publication, appraisal (including periodic collateral appraisals or business valuations), real estate surveys, real estate title policies and endorsements, and environmental audits, (c) reasonable and documented costs and expenses incurred by DIP Lender in the disbursement of funds to Borrower (by wire transfer or otherwise), (d) reasonable and documented costs and expenses paid or incurred by DIP Lender to correct any default or enforce any provision of the Loan Documents, or in gaining possession of, maintaining, handling, preserving, storing, shipping, selling, preparing for sale, or advertising to sell the Collateral, or any portion thereof, irrespective of whether a sale is consummated, (e) audit fees and expenses of DIP Lender related to any inspections or audits to the extent of the fees and charges (and up to the amount of any limitation) contained in the Agreement, (f) reasonable and documented costs and expenses of third party claims or any other suit paid or incurred by DIP Lender in enforcing or defending the Loan Documents or in connection with the transactions contemplated by the Loan Documents or DIP Lender's relationship with Debtors or any Subsidiary, (g) DIP Lender's reasonable and documented costs and expenses (including reasonable and documented attorneys' fees) incurred in conducting its due diligence and in advising, structuring, drafting, negotiating, reviewing, administering, syndicating, or amending the Loan Documents (including the DIP Orders and all motions and other filings related thereto), (h) DIP Lender's reasonable and documented costs and expenses (including reasonable and documented attorneys, accountants, consultants, and other advisors fees and expenses) incurred in terminating, enforcing (including attorneys, accountants, consultants, and other advisors fees and expenses incurred in connection with the Chapter 11 Cases or in exercising rights or remedies under the Loan Documents), exercising its rights or remedies, or defending the Loan Documents, irrespective of whether suit is brought, or in taking any Remedial Action concerning the Collateral, and (i) the reasonable and documented fees and expenses paid or incurred by DIP Lender if it elects to employ the services of one or more third parties to perform financial audits of the Debtors, to appraise any of the Collateral, to establish electronic reporting for Debtors or to assess the Debtors' business valuation.

"<u>DIP Lender-Related Person</u>" means DIP Lender, together with DIP Lender's Affiliates, officers, directors, employees, attorneys, and agents.

"<u>DIP Order</u>" means, collectively, the Interim DIP Order and the Final Order, or either the Interim DIP Order or the Final DIP Order, as the context may imply.

"<u>Disposition</u>" means any sale, assignment, conveyance, transfer or other disposition (by way of merger, casualty, condemnation or otherwise) by a Debtor to any Person of any assets of any Debtor. "<u>Dispose</u>" has a correlative meaning.

"Dollars" or "\$" means United States dollars.

"Effect of Bankruptcy" means, with respect to any contractual obligation, contract or agreement to which any Debtor is a party, any default, unenforceability or other legal consequences arising solely on account of the commencement of the Chapter 11 Cases (including by the implementation of the automatic stay) or, as a result of any order of the Bankruptcy Court, including any such order rejecting any such contractual obligation, contract or agreement.

"<u>Effective Date</u>" refers to the date on which the conditions precedent set forth in <u>Section 3.1</u> have been satisfied or waived by DIP Lender.

"Enforcement Action" means (a) any action to foreclose a Lien as to any Collateral, (b) any action to take possession of, or sell, or otherwise realize upon, or to exercise any other secured creditor rights or remedies with respect to, any Collateral, including a sale or other disposition after the occurrence and during the occurrence of an Event of Default, (c) taking of any other actions against any Collateral, including the taking of control or possession of, or the exercise of any right of setoff with respect to, any Collateral, (d) the filing of a lawsuit to collect or to require performance owed by any Debtor, (e) the commencement of any legal proceedings or actions against or with respect to any Debtor or any of its property or assets or any Collateral after the commencement of the Chapter 11 Cases, including the seeking of relief from the automatic stay or from any other stay in the Chapter 11 Cases, the conversion of any Chapter 11 Case to a case under Chapter 7 of the Bankruptcy Code, the dismissal of any Chapter 11 Case under Section 1112 of the Bankruptcy Code, the appointment of a trustee under Chapter 7 or Chapter 11 of the Bankruptcy Code or of a responsible officer or any examiner with enlarged powers relating to the operation of the business (powers beyond those set forth in Section 1106(a)(3) and (4) of the Bankruptcy Code) under Section 1106(d) of the Bankruptcy Code, (f) the termination of the Loan Commitment under this Agreement or the acceleration of the Obligations hereunder, in either case, as a result of the occurrence of an Event of Default hereunder, or (g) DIP Lender has ceased or refused to fund any requested Advance under this Agreement that is in accordance with Section 6.14 irrespective of a Default or an Event of Default for longer than three (3) Business Days.

"Environmental Actions" means any complaint, summons, citation, notice, directive, order, claim, litigation, investigation, judicial or administrative proceeding, judgment, letter, or other communication from any Governmental Authority, or any third party involving violations of Environmental Laws or releases of Hazardous Materials from (a) any assets, properties, or businesses of any Debtor, any Subsidiary, or any of their predecessors in interest, (b) from adjoining properties or businesses, or (c) from or onto any facilities which received Hazardous Materials generated by any Debtor, any Subsidiary, or any of their predecessors in interest.

"Environmental Law" means any applicable federal, state, provincial, foreign or local statute, law, rule, regulation, ordinance, code, binding and enforceable guideline, binding and enforceable written policy or rule of common law now or hereafter in effect and in each case as amended, or any judicial or administrative interpretation thereof, including any judicial or administrative order, consent decree or judgment, in each case, to the extent binding on any Debtor or any Subsidiary, relating to the environment, the effect of the environment on employee health, or Hazardous Materials, in each case as amended from time to time.

"Environmental Liabilities" means all liabilities, monetary obligations, losses, damages, punitive damages, consequential damages, treble damages, costs and expenses (including all reasonable fees, disbursements and expenses of counsel, experts, or consultants, and costs of investigation and feasibility studies), fines, penalties, sanctions, and interest incurred as a result of any claim or demand, or Remedial Action required, by any Governmental Authority or any third party, and which relate to any Environmental Action.

"Environmental Lien" means any Lien in favor of any Governmental Authority for Environmental Liabilities.

"Equipment" means equipment (as that term is defined in the UCC).

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and any successor statute thereto.

"ERISA Affiliate" means any Person whose employees are treated as employed by the same employer as the employees of any Debtor or a Debtor Subsidiary under IRC Section 414 or ERISA Section 4001.

"ERISA Event" means any of the following: (a) a Reportable Event with respect to a Pension Plan; (b) the incurrence by any Debtor, Debtor Subsidiary or any of their ERISA Affiliates of any liability with respect to a withdrawal by a Debtor, Debtor Subsidiary or any of their ERISA Affiliates from a Pension Plan subject to Section 4063 of ERISA during a plan year in which it was a substantial employer (as defined in Section 4001(a)(2) of ERISA) or a cessation of operations that is treated as such a withdrawal under Section 4062(e) of ERISA; (c) the incurrence by any Debtor, Debtor Subsidiary or any their ERISA Affiliates of any liability with respect to a complete or partial withdrawal from a Multiemployer Plan or the receipt by any Debtor, Debtor Subsidiary or any of their ERISA Affiliates of notification that a Multiemployer Plan is in reorganization; (d) the filing of a notice of intent to terminate, the treatment of a Benefit Plan amendment as a termination under Sections 4041 or 4041A of ERISA, or the commencement of proceedings by the PBGC to terminate a Pension Plan or Multiemployer Plan; (e) an event or condition that constitutes grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Pension Plan or Multiemployer Plan; or (f) the imposition of any liability under Title IV of ERISA, other than for PBGC premiums due but not delinquent under Section 4007 of ERISA, upon any Debtor, Debtor Subsidiary or any of their ERISA Affiliates.

"Event of Default" has the meaning specified therefor in Section 7.

"Exchange Act" means the Securities Exchange Act of 1934, as in effect from time to time.

"Excluded Accounts" means (a) Debtors' payroll accounts, (b) Debtors' pension and pension reserve accounts, (c) Debtors' employee benefit account(s) and (d) Debtors' withholding tax and fiduciary accounts.

"Facility Fee" has the meaning specified therefor in Section 2.7.

"<u>Final DIP Order</u>" means an order of the Bankruptcy Court, together with all extensions, modifications and amendments thereto, in form, scope and substance satisfactory to DIP Lender, which, among other matters but not by way of limitation, authorizes Debtors to (i) obtain credit and authorizes each Debtor to incur Obligations, (ii) grant Liens under this Agreement and the other Loan Documents, as the case may be, provides for the super priority of DIP Lender's claims (subject to the Carve-Out), which order is a Final Order.

"Final Order" means an order or judgment of the Bankruptcy Court as entered on the docket of the Clerk of the Bankruptcy Court, that has not been reversed, stayed, modified or amended and as to which the time to appeal, petition for certiorari, reargue or seek rehearing has expired and no proceeding for certiorari, reargument or rehearing is pending or if an appeal, petition for certiorari, reargument, or rehearing has been sought, the order or judgment of the Bankruptcy Court has been affirmed by the highest court to which the order was appealed, from which the reargument or rehearing was sought, or certiorari has been denied and the time to take any further appeal or to seek certiorari or further reargument or rehearing has expired.

"<u>Financial Officer</u>" means, with respect to each Debtor, the chief financial officer or the chief restructuring officer that has responsibility for overseeing financial matters of such Debtor, and also the treasurer, assistant treasurer, controller or assistant controller responsible for such Debtor.

"Fiscal Month" means any fiscal month of the Borrower.

"Fiscal Week" means any fiscal week of the Borrower.

"GAAP" means generally accepted accounting principles as in effect from time to time in the United States, consistently applied.

"Governing Documents" means, (a) with respect to any corporation, the certificate or articles of incorporation and the bylaws (or equivalent or comparable constitutive documents with respect to any non-U.S. jurisdiction) of such Person; (b) with respect to any limited liability company, the certificate or articles of formation or organization and operating agreement of such Person; and (c) with respect to any partnership, joint venture, trust or other form of business entity, the partnership, joint venture or other applicable agreement of formation or organization of such Person and any agreement, instrument, filing or notice with respect thereto filed in connection with such Person's formation or organization with the applicable Governmental Authority in the jurisdiction of its formation or organization and, if applicable, any certificate or articles of formation or organization of such Person.

"Governmental Authority" means any federal, state, local, or other governmental or administrative body, instrumentality, board, department, or agency or any court, tribunal,

administrative hearing body, arbitration panel, commission, or other similar dispute-resolving panel or body.

"Hazardous Materials" means (a) substances that are defined or listed in, or otherwise classified pursuant to, any Applicable Laws or regulations as "hazardous substances," "hazardous materials," "hazardous wastes," "toxic substances," or any other formulation intended to define, list, or classify substances by reason of deleterious properties such as ignitability, corrosivity, reactivity, carcinogenicity, reproductive toxicity, or "EP toxicity", (b) oil, petroleum, or petroleum derived substances, natural gas, natural gas liquids, synthetic gas, drilling fluids, produced waters, and other wastes associated with the exploration, development, or production of crude oil, natural gas, or geothermal resources, (c) any flammable substances or explosives or any radioactive materials, (d) asbestos in any form or electrical equipment that contains any oil or dielectric fluid containing levels of polychlorinated biphenyls in excess of 50 parts per million, and (e) and all other substances or wastes of any nature regulated pursuant to any Environmental Law because of their dangerous or deleterious properties, including any material listed as a hazardous substance under Section 101(14) of CERCLA.

"Indebtedness" of any Person means, without duplication: (a) all obligations of such Person for borrowed money, (b) all obligations of such Person evidenced by bonds, debentures, notes, or other similar instruments and all reimbursement or other obligations in respect of letters of credit, bankers acceptances, interest rate swaps, or other financial products, (c) all obligations of such Person under conditional sale or other title retention agreements relating to property acquired by such Person, (d) all obligations of such Person as a lessee under Capital Leases, (e) all obligations or liabilities of others secured by a Lien on any asset of a Person or its Subsidiaries, irrespective of whether such obligation or liability is assumed, (f) all obligations of such Person to pay the deferred purchase price of assets (other than trade payables incurred in the ordinary course of business and repayable in accordance with customary trade practices), (g) all obligations of such Person owing under hedge agreements, (h) all obligations or liabilities of such Person associated with bonds or surety obligations incurred in the ordinary cause of business or required by the laws, rules, regulations and orders of any Governmental Authority, and (h) any obligation of such Person guaranteeing or intended to guarantee (whether directly or indirectly guaranteed, endorsed, comade, discounted, or sold with recourse) any obligation of any other Person that constitutes Indebtedness under any of clauses (a) through (g) above.

"Indemnified Liabilities" has the meaning specified therefor in Section 11.3.

"Indemnified Person" has the meaning specified therefor in Section 11.3.

"Interim Advance" is defined in the recitals to this Agreement.

"Interim DIP Order" means the Interim Order (I) Authorizing the Debtors to Obtain Senior Secured Superpriority Postpetition Financing; (II) Granting (A) Liens and Superpriority Administrative Expense Claims and (B) Adequate Protection; (III) Authorizing Use of Cash Collateral; (IV) Modifying the Automatic Stay; (V) Scheduling a Final Hearing; and (VI) Granting Related Relief entered by the Bankruptcy Court in the Chapter 11 Cases.

"Inventory" means inventory (as that term is defined in the UCC).

"Investment" means, with respect to any Person, any investment by such Person in any other Person (including Affiliates) in the form of loans, guarantees, advances, or capital contributions (excluding (a) commission, travel, and similar advances to officers and employees of such Person made in the ordinary course of business, and (b) bona fide Accounts arising in the ordinary course of business consistent with past practice), purchases or other acquisitions of Indebtedness, Capital Stock, or all or substantially all of the assets of such other Person (or of any division or business line of such other Person), and any other items that are or would be classified as investments on a balance sheet prepared in accordance with GAAP.

"IRC" means the Internal Revenue Code of 1986, as amended from time to time or any successor statute.

"IRS" means the United States Internal Revenue Service.

"Lien" means any interest in an asset securing an obligation owed to, or a claim by, any Person other than the owner of the asset, irrespective of whether (a) such interest is based on the common law, statute, or contract, (b) such interest is recorded or perfected, and (c) such interest is contingent upon the occurrence of some future event or events or the existence of some future circumstance or circumstances. Without limiting the generality of the foregoing, the term "Lien" includes the lien or security interest arising from a mortgage, deed of trust, encumbrance, notice of Lien, levy or assessment, pledge, hypothecation, assignment, deposit arrangement, security agreement, conditional sale or trust receipt, or from a lease, consignment, or bailment for security purposes and also includes reservations, exceptions, encroachments, easements, rights-of-way, covenants, conditions, restrictions, leases, and other title exceptions and encumbrances affecting Real Property.

"<u>Loan Commitment</u>" means commitment of the DIP Lender to make Advances hereunder, as such commitment may be terminated pursuant to <u>Section 8.1</u>.

"<u>Loan Commitment Amount</u>" means \$14,000,000 plus all interest, fees and expenses that will be capitalized under this Agreement.

"<u>Loan Documents</u>" means the Agreement, the Security Documents, any Control Agreements executed and delivered by the any Debtor in favor of DIP Lender, any note or notes executed by Borrower in connection with the Agreement and payable to DIP Lender, and any other agreement entered into, now or in the future, by any Debtor and DIP Lender in connection with the Agreement.

"Material Adverse Change" means (a) a material adverse change in or a material adverse effect upon (in either case, irrespective of whether occurring as a result of a specific event or circumstance or otherwise) the business, financial condition or results of operations of either: (i) Borrower; or (ii) the Debtors and Subsidiaries taken as a whole, (b) a material impairment of any Debtor's ability to perform its obligations under the Loan Documents to which it is a party or of DIP Lender's ability to enforce the Obligations or realize upon the Collateral, or (c) a material impairment of the enforceability of this Agreement and any of the other Loan Documents, the enforceability or priority of DIP Lender's Liens with respect to the Collateral, or the rights and remedies of DIP Lender under this Agreement or any of the other Loan Documents; provided that

the following shall not constitute a Material Adverse Change: (i) the filing of the Chapter 11 Cases or any Effect of Bankruptcy as a result thereof shall not constitute a Material Adverse Change; or (ii) any change resulting directly from the anticipated commencement or commencement and continuation of the Chapter 11 Cases.

"Material Contract" means any agreement, the termination or breach of which would reasonably likely result in a Material Adverse Change under clause (a)(ii) of such definition.

"Material Indebtedness" means Indebtedness (other than the Obligations and the Prepetition Bridge Loans) of any Debtor in an aggregate principal amount exceeding \$100,000.

"Maturity Date" means the earlier of (a) May 30, 2025, (b) the closing date of a 363 Sale, (c) the effective date of any Reorganization Plan confirmed pursuant to an order entered by the Bankruptcy Court, or (d) the acceleration of the maturity of the Obligations pursuant to this Agreement, any other Loan Document or the DIP Orders.

"Milestones" has the meaning specified therefor in Section 5.18(c).

"Mortgages" means, individually and collectively, one or more mortgages, deeds of trust, or deeds to secure debt, executed and delivered by one or more Debtors in favor of DIP Lender, in form and substance satisfactory to DIP Lender, that encumber the real property that is Collateral.

"Multiemployer Plan" means any employee benefit plan of the type described in Section 4001(a)(3) of ERISA to which Debtor, Debtor Subsidiary or any of their ERISA Affiliates makes or is obligated to make contributions, or during the preceding five plan years, has made or been obligated to make contributions or to which any of the foregoing could have any liability.

"Net Cash Proceeds" means, (a) with respect to any sale or disposition by any Debtor of property or assets, the amount of cash received (directly or indirectly) from time to time (whether as initial consideration or through the payment of deferred consideration) by or on behalf of such Person, in connection therewith after deducting therefrom only (i) the amount of any Indebtedness secured by any prior senior Permitted Lien on any asset and the Carve-Out (other than (A) Indebtedness owing to DIP Lender under this Agreement or the other Loan Documents and (B) Indebtedness assumed by the purchaser of such asset) which is required to be, and is, repaid in connection with such disposition, (ii) reasonable expenses related thereto incurred by such Person in connection therewith, and (iii) taxes paid or payable to any taxing authorities by such Person in connection therewith, and (b) with respect to the issuance or incurrence of any Indebtedness by any Debtor, or the sale or issuance by any Debtor of any shares of its Capital Stock, the aggregate amount of cash received (directly or indirectly) from time to time (whether as initial consideration or through the payment or disposition of deferred compensation) by or on behalf of such Person in connection therewith, after deducting therefrom only (i) reasonable expenses related thereto incurred by such Person in connection therewith, (ii) transfer taxes paid by such Person in connection therewith and (iii) net income taxes to be paid in connection therewith (after taking into account any tax credits or deductions and any tax sharing arrangements); in each case of clause (a) and (b) to the extent, but only to the extent, that the amounts so deducted are (A) actually paid to a Person that, except in the case of reasonable out-of-pocket expenses, is not an Affiliate of such

Person or any of its Subsidiaries and (B) properly attributable to such transaction or to the asset that is the subject thereof.

"Notice of Borrowing" is defined in Section 2.2(a).

"Obligations" means all loans, Advances, debts, the principal amount, interest, premiums, liabilities (including all amounts charged to Borrower by DIP Lender), obligations (including indemnification obligations), fees (including the Facility Fee), charges, costs, DIP Lender Expenses, lease payments, guaranties, covenants, and duties of any kind and description owing by Borrower to DIP Lender pursuant to or evidenced by the Loan Documents and irrespective of whether for the payment of money, whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, and including all interest not paid when due and all DIP Lender Expenses that Borrower is required to pay or reimburse by the Loan Documents, by law, or otherwise. In addition, the term "Obligations" shall include all obligations of each Guarantor hereunder as a result of the unconditional and continuing guaranty by each Guarantor of the Obligations. Any reference in the Agreement or in the Loan Documents to the Obligations shall include all or any portion thereof and any extensions, modifications, renewals, or alterations thereof. Without limiting the foregoing and subject to the DIP Orders, the Obligations shall constitute allowed administrative expense claims in the Chapter 11 Cases having priority pursuant to Section 364(c)(1) of the Bankruptcy Code over all administrative expense claims and unsecured claims against the Debtors now existing or hereafter arising, of any kind or nature whatsoever, including, without limitation, all administrative expense claims of the kind specified in Sections 105, 326, 328, 330, 331, 503(b), 506(c), 507(a), 507(b), 546(c), 1113 and 1114 of the Bankruptcy Code.

"Paid in Full" and "Payment in Full" means, with respect to any Obligations, the indefeasible payment in full of such Obligations in cash (or otherwise to the written satisfaction, in such holder's discretion, of the holder thereof), and, in the event any such Obligations are paid over time or modified pursuant to Section 1129 of the Bankruptcy Code (or any similar provision of any other applicable bankruptcy law), shall further mean that the holder thereof shall have received the final payment due on account of such Obligations. For purposes of the foregoing, the "holder" of any applicable Obligations shall be deemed to be the Person entitled to receipt of payment thereof. Notwithstanding the foregoing, the Obligations shall not be deemed to have been "paid in full" until the Loan Commitment has expired or been terminated.

"Participant" has the meaning specified therefor in Section 14.3.

"<u>Patriot Act</u>" means the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, P.L. 107-56, as amended.

"PBGC" means the Pension Benefit Guaranty Corporation.

"Pension Plan" means any "employee pension benefit plan" (as that term is defined in Section 3(2) of ERISA), other than a Multiemployer Plan, that is subject to Title IV of ERISA and is sponsored or maintained by Debtor, Debtor Subsidiary or any their ERISA Affiliates or to which any Debtor, Debtor Subsidiary or any their ERISA Affiliates contributes or has an obligation to contribute, or in the case of a multiple employer or other plan described in Section 4064(a) of

ERISA, has made contributions at any time during the immediately preceding five plan years or for which any of the foregoing has or could have any liability.

"Permitted Disposition" means any of the following:

- (a) sales or other dispositions of Equipment that is substantially worn, damaged, or obsolete in the ordinary course of business consistent with past business practices;
 - (b) sales of Inventory to buyers in the ordinary course of business;
- (c) the licensing, on a non-exclusive basis, of patents, trademarks, copyrights, and other intellectual property rights in the ordinary course of business;
 - (d) sales of Inventory;
 - (e) transactions permitted pursuant to Section 6.3;
- (f) Dispositions expressly approved by the Bankruptcy Court that are de minimis in value or are otherwise agreed by the DIP Lender and Debtors; and
 - (g) Dispositions permitted under <u>Section 6.4</u>.

"Permitted Indebtedness" means each of the following:

- (a) Indebtedness created under the Loan Documents;
- (b) Prepetition Bridge Loans;
- (c) Guarantees by any Debtor of Indebtedness permitted hereunder;
- (d) Indebtedness incurred in the ordinary course of business in connection with the financing of insurance premiums; and
- (e) Indebtedness relating to surety and appeal bonds, performance bonds and other obligations of a like nature incurred in the ordinary course of business or required by the laws, rules, regulations and orders of any Governmental Authority;

<u>provided</u>, that with respect to Indebtedness in clauses (c) through (e), such Indebtedness arises Post-Petition and is permitted pursuant to the Approved Budget.

"Permitted Investments" means (a) Investments in cash and Cash Equivalents, (b) Investments in negotiable instruments for collection, (c) advances made in connection with purchases of goods or services in the ordinary course of business, (d) to the extent approved by DIP Lender, Investments received in settlement of amounts due to any Debtor or any Subsidiary effected in the ordinary course of business or owing to any Debtor or any Subsidiary as a result of Insolvency Proceedings involving an Account Debtor or upon the foreclosure or enforcement of any Lien in favor of any Debtor or any Subsidiary, (e) Investments in Borrower and (f) Investments made by Borrower or any Subsidiaries in Deposit Accounts or Securities Accounts in which the Lien in favor of DIP Lender has been perfected.

"Permitted Liens" means:

- (a) Liens held by DIP Lender to secure the Obligations;
- (b) Liens for unpaid taxes, assessments, or other governmental charges or levies that either (i) are not yet delinquent, or (ii) (A) do not have priority over DIP Lender's Liens and (B) with respect to which the underlying taxes, assessments, or charges or levies are the subject of Permitted Protests;
- (c) To the extent created prior to the Petition Date, any Lien created in favor of the DIP Lender securing only the Prepetition Bridge Loans;
- (d) Liens arising from precautionary UCC filings regarding "true" operating leases or the consignment of goods to Debtors;
- (e) Liens arising by operation of law in favor of warehousemen, landlords, carriers, mechanics, materialmen, laborers, or suppliers, incurred in the ordinary course of Debtors' business and not in connection with the borrowing of money, and which Liens either (i) are for sums not yet delinquent, or (ii) are the subject of Permitted Protests;
- (f) Banker's Liens, rights of set-off or similar rights and remedies arising in the ordinary course of business and burdening only deposit accounts or other funds maintained with a creditor depository institution, provided that (i) no such deposit account is a dedicated cash collateral account or is subject to restrictions against access by the depository in excess of those set forth by regulations promulgated by Board of Governors of the Federal Reserve System, (ii) no such deposit account is intended by any Debtor or any Subsidiaries to provide collateral to the depository institution to secure Indebtedness to such depository institution and (iii) CrossFirst Liens shall not constitute Permitted Liens;
- (g) Non-consensual defects in title (which might otherwise constitute Liens) to the properties of any Debtor or any Subsidiary arising in the ordinary course of such entity's business or incidental to the ownership of their respective properties, including, without limitation, easements, restrictions, servitudes, permits, conditions, covenants, exceptions, zoning restrictions and reservations for purposes of roads, pipelines, transmission lines, transportation lines, distribution lines and other like purposes, provided that no such Liens shall materially detract from the value of marketability of the property subject thereto or materially impair the use or operation thereof in the operation of the business of Debtors;
- (h) Liens incurred in the ordinary course of business on amounts deposited in connection with obtaining worker's compensation or other unemployment insurance; and
- (i) To the extent that, as of the Petition Date, US Foods holds validly perfected, enforceable and unavoidable Liens on any assets of any Debtor, such Liens.

Notwithstanding anything to the contrary herein, the designation of a prior Lien as a "Permitted Lien" shall not operate in any way to preclude or estop the Borrowers or any other party in interest in the Chapter 11 Cases from objecting to such Lien or the obligation that such Lien secures.

"<u>Permitted Protest</u>" means the right of any Debtor or any Subsidiary to protest any Lien (other than any Lien that secures the Obligations), Taxes (other than payroll taxes or taxes that are the subject of a United States federal tax lien), or rental payment, provided that (a) a reserve with respect to such obligation is established on Debtors' books and records in such amount as is required under GAAP, (b) any such protest is instituted promptly and prosecuted diligently by the Debtors, in good faith, and (c) DIP Lender is satisfied that, while any such protest is pending, there will be no impairment of the enforceability, validity, or priority of any of DIP Lender's Liens.

"<u>Person</u>" means natural persons, corporations, limited liability companies, limited partnerships, general partnerships, limited liability partnerships, joint ventures, trusts, land trusts, business trusts, or other organizations, irrespective of whether they are legal entities, and governments and agencies and political subdivisions thereof.

"Petition Date" has the meaning provided in the recitals to this Agreement.

"<u>Post-Petition</u>" means the time period commencing immediately upon the filing of the Chapter 11 Cases.

"Prepetition Bridge Collateral" means the Collateral pledged to the Prepetition DIP Lender in connection with the Prepetition Secured Note.

"<u>Prepetition Bridge Lender</u>" means the DIP Lender, in its capacity as the lender under the Prepetition Secured Note.

"Prepetition Bridge Loans" is defined in the recitals to this Agreement.

"Prepetition Bridge Loan Amount" is defined in the recitals to this Agreement.

"Prepetition Bridge Loan Roll-Up" is defined in Section 2.1.

"<u>Prepetition Indebtedness</u>" means any or all Indebtedness incurred by Debtors prior to the filing of the Chapter 11 Cases.

"Prepetition Secured Note" is defined in the recitals to this Agreement.

"Principal Amount" means, when determined, the principal balance of all outstanding and unpaid Advances (including both cash Advances, deemed Advances and capitalized amounts (whether such capitalized are interest, fees or expenses).

"<u>Professional Fees and Expenses</u>" means (a) amounts payable pursuant to 28 U.S.C. § 1930(a)(6), and (b) professional fees and expenses of attorneys retained by the Debtors or the Creditors' Committee.

"Real Property" means any estates or interests in real property now owned or hereafter acquired by any Debtor or any Subsidiary and the improvements thereto.

"Record" means information that is inscribed on a tangible medium or which is stored in an electronic or other medium and is retrievable in perceivable form.

"Remedial Action" means all actions taken to (a) clean up, remove, remediate, contain, treat, monitor, assess, evaluate, or in any way address Hazardous Materials in the indoor or outdoor environment, (b) prevent or minimize a release or threatened release of Hazardous Materials so they do not migrate or endanger or threaten to endanger public health or welfare or the indoor or outdoor environment, (c) restore or reclaim natural resources or the environment, (d) perform any pre-remedial studies, investigations, or post-remedial operation and maintenance activities, or (e) conduct any other actions with respect to Hazardous Materials authorized by Environmental Laws.

"Reorganization Plan" means a Chapter 11 plan of reorganization or liquidation filed by any Debtor in any Chapter 11 Case.

"Reportable Event" means any of the events set forth in Section 4043(c) of ERISA, other than events for which the thirty-day notice period has been waived.

"Responsible Officer" of any Person shall mean the chief executive officer, president, any vice president, the chief financial officer, chief restructuring officer, or treasurer of such Person or any other officer having substantially the same authority and responsibility as a chief executive officer or president, or a Financial Officer.

"Restricted Payment" means any dividend or other distribution (whether in cash, securities or other property) with respect to any class of Capital Stock of a Person, or any payment (whether in cash, securities or other property), including any sinking fund or similar deposit, on account of the purchase, redemption, retirement, acquisition, cancellation or termination of any Capital Stock of a Person or any option, warrant or other right to acquire any Capital Stock of a Person; provided that "Restricted Payments" shall not include any dividends payable solely in Capital Stock of a Debtor.

"Sale and Bidding Procedures Motion" is defined in Section 5.18(c)(iii).

"Sale Order" is defined in Section 5.18(c)(v).

"Securities Account" means a "securities account" (as that term is defined in the UCC).

"Security Documents" means each security agreement (including Section 2.8 hereof), any Control Agreement, each Mortgage, and all other instruments or documents executed and delivered pursuant to this Agreement or any other Loan Document to secure any of the Obligations, each in form and substance satisfactory to DIP Lender.

"Subsidiary" of a Person means a corporation partnership, limited liability company, or other entity in which that Person directly or indirectly owns or controls the shares of Capital Stock having ordinary voting power to elect a majority of the board of directors (or appoint other comparable managers) of such corporation, partnership, limited liability company, or other entity. Unless the context requires otherwise, "Subsidiary" as used in the Loan Documents refers to a Subsidiary of a Debtor.

"Successor Cases" means any other proceedings superseding or related to the Chapter 11 Cases or in any case under chapter 7 of the Bankruptcy Code upon the conversion of any of the Chapter 11 Cases.

"Tax" and "Taxes" has the meaning specified therefor in Section 10.1(a).

"<u>Termination Date</u>" means the earlier to occur of (a) the Maturity Date and (b) the date of the termination of the Loan Commitment pursuant to <u>Section 8.1</u>.

"<u>UCC</u>" means the Uniform Commercial Code of the State of Delaware, as in effect from time to time, or if the context shall require, the Uniform Commercial Code of the State of New York, or if applicable, the law of the state of formation of the applicable Debtor or the law of the jurisdiction in which the applicable asset that is purported to constitute Collateral hereunder (including any deposit account) is located.

"<u>Unfunded Pension Liability</u>" means the excess of a Pension Plan's benefit liabilities under Section 4001(a)(16) of ERISA, over the current value of that Pension Plan's assets, determined in accordance with the assumptions used for funding the Pension Plan pursuant to Section 412 of the Code for the applicable plan year.

"United States" means the United States of America.

"Variance Report" means a report prepared by Borrower's management which reflects (a) Debtors' actual cash receipts and disbursements, and (b) on a line-item basis the Debtors' actual performance compared to the Approved Budget, in each case for the immediately preceding week and on a cumulative basis for the period after the Effective Date and the percentage variance of Debtors' actual results from those reflected in the then extant Approved Budget, along with management's explanation of such variance.

"Voidable Transfer" has the meaning specified therefor in Section 16.5.

EXHIBIT A

(Reserved)

EXHIBIT B

Form of NOTICE OF BORROWING

Date: [*], 2025

OTB Lender, LLC, as DIP Lender under the below-referenced Credit Agreement c/o Papas Restaurants Inc. 13939 NW Freeway Houston, Texas 77040-5115 Attn: [_____]

This Notice of Borrowing is executed and delivered by **OTB ACQUISITION LLC**, a Delaware limited liability company, as debtor and debtor in possession under Chapter 11 of the Bankruptcy Code (as defined in the Credit Agreement) (the "<u>Borrower</u>"), pursuant to that certain Secured Super-Priority Debtor-In-Possession Credit Agreement, dated as of March [__], 2025 (as amended or otherwise modified from time to time, the "<u>Credit Agreement</u>"), by and among Borrower, OTB Lender, LLC, as DIP Lender (the "<u>DIP Lender</u>") and certain guarantors party thereto. This Notice of Borrowing hereby gives you notice pursuant to Section 2.2 of the Credit Agreement that the undersigned hereby requests an Advance under the Credit Agreement (the "<u>Proposed Advance</u>"), and in that connection sets forth below the information relating to such Proposed Advance as required by Section 2.2 of the Credit Agreement. All capitalized terms used herein but not defined herein have the same meanings herein as set forth in the Credit Agreement.

- (1) The aggregate principal amount of the Proposed Advance is \$[*].
- (2) The funding date of the Proposed Advance is [*], 2025.
- (3) The Budget Period of the Proposed Advance begins on [*], 2025 and ends on [*], 2025[; provided that the Budget Period as it pertains to the Professional Fee Reserve is the entirety of the anticipated Availability Period].
- (4) Attached hereto is a reconciliation of the proposed uses of the proceeds of the Proposed Advance to the Approved Budget for the Budget Period.
- (5) The representations and warranties contained in the Credit Agreement or in the other Loan Documents are true, correct and complete in all material respects on and as of the date hereof and shall be true, correct and complete on and as of the date of such extension of credit, as though made on and as of such date (except to the extent that such representations and warranties relate solely to an earlier date in which case such representations and warranties shall be true and correct as of such earlier date);
- (6) No Default or Event of Default has occurred or is continuing, nor shall either result from the making of the Proposed Advance.
- (7) The Proposed Advance does not exceed the amounts permitted in accordance with Section 2.1(d) of the Credit Agreement and the proceeds of the Advances shall be used in strict compliance with Section 6.14 and the Approved Budget.

	(8)	Each	other	conditi	on pre	eceden	it to 1	the fu	ınding	of t	he F	ropos	sed A	dvan	ce se	t for	th in
Section	3.2 of	the Cre	dit Ag	reemer	nt has	been s	satisfi	ed or	, with	respe	ect to	each	cond	lition	prec	edent	that
describ	es certa	in facts	that r	nust be	e true,	each	such	fact i	s true	as of	f the	date	herec	f; pro	ovide	d tha	t the
paymei	nt of an	y fees	that w	ill be	funde	d with	the	proce	eds of	the	Adv	ance	shall	be p	aid f	rom	such
proceed	ds.																

Very truly yours,
OTB ACQUISITION LLC, as Borrower
By:
Name: Jonathan Tibus

Name: Jonathan Tibus
Title: Chief Restructuring Officer

EXHIBIT 2

Initial DIP Budget

To Be Attached.

Exhibit B

The Tibus Declaration

UNITED STATES BANKRUPTCY COURT NORTHERN DISTRICT OF GEORGIA ATLANTA DIVISION

)
In re:) Chapter 11
OTB HOLDING LLC, et al.,1) Case No. 25-52415 (SMS)
Debtors.) (Joint Administration Requested)
)

DECLARATION OF JONATHAN TIBUS IN SUPPORT OF DEBTORS' MOTION FOR INTERIM AND FINAL ORDERS (I) AUTHORIZING THE DEBTORS TO OBTAIN SENIOR SECURED SUPERPRIORITY POSTPETITION FINANCING; (II) GRANTING (A) LIENS AND SUPERPRIORITY ADMINISTRATIVE EXPENSE CLAIMS AND (B) ADEQUATE PROTECTION; (III) AUTHORIZING USE OF CASH COLLATERAL; (IV) MODIFYING THE AUTOMATIC STAY; (V) SCHEDULING A FINAL HEARING; AND (VI) GRANTING RELATED RELIEF

- I, Jonathan Tibus, pursuant to 28 U.S.C. § 1764, hereby declare and state:
- 1. I am the Chief Restructuring Officer ("CRO") of OTB Holding LLC ("OTB Holding") and all of its direct and indirect wholly-owned subsidiaries as debtors and debtors-in-possession in the above-captioned chapter 11 cases (collectively, the "Debtors" and, together with certain non-Debtor affiliates, "On The Border" or the "Company"). I was appointed CRO of On The Border in January of 2025. I submit this Declaration in support of the Debtors' Motion for Interim and Final Orders (I) Authorizing the Debtors to Obtain Senior Secured Superpriority Postpetition Financing; (II) Granting (A) Liens and Superpriority Administrative Expense Claims

The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number, include: OTB Holding LLC (3213), OTB Acquisition LLC (8500), OTB Acquisition of New Jersey LLC (1506), OTB Acquisition of Howard County LLC (9865), Mt. Laurel Restaurant Operations LLC (5100), OTB Acquisition of Kansas LLC (9014), OTB Acquisition of Baltimore County, LLC (6963). OTB Holding LLC's service address is One Buckhead Plaza, 3060 Peachtree Road, NW, Atlanta, GA 30305.

- and (B) Adequate Protection; (III) Authorizing Use of Cash Collateral; (IV) Modifying the Automatic Stay; (V) Scheduling a Final Hearing; and (VI) Granting Related Relief (the "DIP Motion").²
- 2. I am also a Managing Director at Alvarez & Marsal North America, LLC (together with employees of its affiliates—all of which are wholly-owned by its parent company and employees—its wholly-owned subsidiaries, and independent contractors, "A&M").
- 3. I have been a full-time restructuring advisor for 27 years. I received a bachelor's degree from Florida State University and an MBA from The University of Florida. I am also a Certified Insolvency and Restructuring Advisor with a Certification in Distressed Business Valuation and am a member of the American Bankruptcy Institute, the Association of Insolvency and Restructuring Advisors, and the Turnaround Management Association.
- 4. I have been employed by A&M for 22 years. A&M is a preeminent consulting firm with extensive experience and an excellent reputation for providing high quality, specialized management and restructuring advisory services to debtors and distressed companies. A&M's core services include turnaround advisory services, interim and crisis management, revenue enhancement, claims management, and creditor and risk management advisory services. A&M provides a wide range of debtor advisory services targeted at stabilizing and improving a company's financial position, including, developing or validating forecasts, business plans and related assessments of strategic position; monitoring and managing cash, cash flow, and supplier relationships; assessing and recommending cost reduction strategies; and designing and negotiating financial restructuring packages. Additionally, A&M provides advice on specific

² Capitalized terms used but not otherwise defined herein have the meaning given to them in the DIP Motion.

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 166 of 171

aspects of the turnaround process and helps manage complex constituency relations and communications. A&M is known for its ability to work alongside company management and key constituents during chapter 11 restructurings. Some notable, publicly disclosed restructuring assignments that I have personally advised on include Ignite Restaurant Group, Inc. (including its Joe's Crab Shack and Brickhouse Tavern restaurant brands), California Pizza Kitchen, Kona Grill, Real Mex Restaurants, Inc., Last Call Operating Co., Krystal Company, Quiznos, Max & Erma's, Garden Fresh (d/b/a/ Souplantation and Sweet Tomatoes), and Red Lobster Management LLC (d/b/a Red Lobster).

- 5. Except as indicated otherwise, I have personal knowledge of the matters set forth herein and all facts set forth herein are based on my personal knowledge, discussions with current and former members of the Debtors' senior management and professional advisors, my review of relevant documents, or my opinion based upon my experience and knowledge of the Debtors' operations and financial conditions. In making this Declaration, I have relied in part on information and materials that the Debtors' personnel and advisors have gathered, prepared, verified, and provided to me, in each case under my ultimate supervision, at my direction, and/or for my benefit in preparing this Declaration. I am authorized to submit this Declaration on the Debtors' behalf, and, if called upon to testify, I could and would testify competently to the facts set forth herein.
- 6. The above-captioned debtors-in-possession (the "<u>Debtors</u>") commenced voluntary cases (the "<u>Chapter 11 Cases</u>") under chapter 11 of the Bankruptcy Code by the filing of petitions in the Bankruptcy Court for the Northern District of Georgia (the "<u>Bankruptcy Court</u>") on March 4, 2025 (the "Petition Date").

A&M's Retention

7. I was initially retained on January 9, 2025, as part of a team from A&M in which I served as chief restructuring officer and A&M provided additional personnel to the Company. We were initially tasked with (i) evaluating the Debtors' current business plan; (ii) identifying cost reduction and operation improvement opportunities and (iii) managing cash, cash forecasting and cash reporting requirements.

The Debtors' Need for DIP Financing

8. As described in the *Declaration of Jonathan M. Tibus in Support of Chapter 11*Petitions and First Day Pleadings (the "First Day Declaration") and the DIP Motion, the Debtors require immediate access to additional liquidity to continue operating in the ordinary course during the Chapter 11 Cases while pursuing a sale transaction. Specifically, based on the Debtors' forecast, the Debtors will be unable to generate sufficient levels of operating cash flow in the ordinary course of business to cover the projected costs of the Chapter 11 Cases without debtorin-possession financing.

Negotiation and Terms of Debtor-in-Possession Financing

- 9. The Debtors' prepetition capital structure is described in detail in the First Day Declaration. The Debtors' substantial prepetition secured indebtedness informed the negotiation and terms of the Debtors' proposed \$14 million senior secured super-priority multi-draw term loan credit facility (the "DIP Facility").
- 10. By January 2025, the Debtors required additional working capital to (i) continue operations, and (ii) evaluate and pursue certain strategic alternatives in order to maximize the value of the Debtors' assets for the benefit of all stakeholders. The Debtors were in

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 168 of 171

default under their Prepetition Credit Agreement and CrossFirst was unable to advance additional funds. However, CrossFirst was willing to subordinate its claims and liens to new bridge financing and potential debtor-in-possession financing (subject to a cap). As a result, A&M, on behalf of the Debtors, commenced a process to identify potential bridge funding to provide the Debtors with sufficient liquidity to continue operations while they evaluate and pursue certain strategic alternatives, including a potential chapter 11 filing, to maximize the value of the Debtors' assets for all stakeholders.

- 11. In January 2025, the Debtors also engaged Hilco Corporate Finance, LLC ("<u>Hilco</u>") for a potential marketing process to sell their assets on a going-concern basis, or to consummate another strategic, value-maximizing transaction that would resolve the Debtors' operational and financial challenges. At the Debtors' direction, Hilco approached interested parties to secure a stalking horse bidder for the sale of the Debtors' assets pursuant to Section 363 of the Bankruptcy Code. In total, during the sale process, Hilco contacted 273 strategic and financial potential bidders to serve as a potential stalking horse bidder, of which 31 ultimately negotiated confidentiality agreements and were provided a confidential information memorandum.
- 12. Following negotiations with two parties to provide bridge financing, on February 14, 2025, the Debtors entered into that certain Secured Promissory Note (the "Prepetition Secured Note") with OTB Lender LLC, as payee. Pursuant to the Prepetition Secured Note, OTB Lender LLC advanced the Debtors an amount equal to \$4,000,000 (the "Bridge Financing").
- 13. OTB Lender LLC has now agreed to provide the DIP Facility (in such capacity, the proposed "<u>DIP Lender</u>"), subject to the refinancing of the Prepetition Secured Note using proceeds of the initial borrowing under the DIP Facility upon entry of the Interim Order. Following

Case 25-52415-sms Doc 17 Filed 03/05/25 Entered 03/05/25 03:03:11 Desc Main Document Page 169 of 171

extensive negotiations, the Debtors anticipate that the DIP Lender will also serve as the stalking horse bidder (in such capacity, the "Stalking Horse Bidder") and credit bid the full amount of its claim for the purchase of substantially all of the Debtors' assets pursuant to Section 363 of the Bankruptcy Code.

The DIP Facility Was Negotiated in Good Faith and at Arm's Length

- 14. I assisted the Debtors, together with their other advisors, with the negotiation of the terms and provisions of the DIP Facility. The DIP Facility is the product of lengthy, good faith, arm's length negotiations that continued right up to the Petition Date. The Debtors, with the assistance of their advisors, engaged in multiple rounds of negotiations with the DIP Lender, resulting in improved terms.
- 15. In negotiations with the DIP Lender, the Debtors focused on obtaining the best terms available. All negotiations between the Debtors and the DIP Lender, who were represented by experienced counsel and financial advisors, were conducted in good faith and on an arm's-length basis. The Debtors' management team and legal and financial advisors, including A&M, were actively involved throughout the process. Those negotiations led to improvements in terms for the Debtors, including, among other things, the reduction of certain fees and material modifications to the initially proposed covenants, events of default, milestones, and conditions precedent to borrowing. Notably, the Debtors anticipate that the DIP Lender will act as Stalking Horse Bidder and credit bid their claims under the DIP Facility rather than require repayment of the obligations thereunder at the close of the sale transaction, which will provide a significant benefit to the Debtors and their estates. As a result of these arm's-length negotiations, the parties agreed upon, and the Debtors presently request authority to enter into, the DIP Facility.

The DIP Facility Is the Best Postpetition Financing Arrangements Presently Available to the Debtors

- 16. Based upon my understanding of the Debtors' liquidity needs, the current state of the debt markets, and the non-existence of financing alternatives, I believe the DIP Facility is the best financing option available to the Debtors under the circumstances. I do not believe alternative sources of financing are available to the Debtors (whether unsecured or secured) on better or comparable terms to the DIP Facility. The proposed DIP Facility will provide the Debtors with immediate access to liquidity that is necessary to ensure that the Debtors' businesses are stabilized, chapter 11 administrative costs are paid, and value is preserved during the course of the Chapter 11 Cases. Additionally, I do not believe it would be prudent, or even possible, to administer the Debtors' chapter 11 estates on a "cash collateral" only basis. Without access to the DIP Facility, the Debtors would have extremely limited cash on hand and I do not expect the Debtors to be able to generate sufficient levels of operating cash flow in the ordinary course of business to cover their working capital needs and the projected administrative costs of these chapter 11 cases.
- 17. The DIP Lender also required the refinancing of the Prepetition Secured Note as a condition to their commitment to provide the DIP Facility and to permit the use of Cash Collateral. As set forth in the DIP Motion, upon the refinancing of the Prepetition Secured Note, the DIP Facility will be available to the Debtors to address their working capital needs, so the Prepetition Secured Note refinancing will have no negative impact on the Debtors' liquidity. The refinancing of the Prepetition Secured Note will also not affect the DIP Lender's position as a secured creditor of the Debtors to the detriment of any other stakeholders or the estates.
- 18. Finally, I believe that the financial terms proposed under the DIP Facility are customary and usual for debtor-in-possession financings of this type. Specifically, the

Page 171 of 171 Document

contemplated pricing, fees, interest rate, and other economic terms of the DIP Facility were all

negotiated at arm's length and are generally consistent with the cost of debtor-in-possession

financings in comparable circumstances.

19. Based on the foregoing, it is my belief that the proposed DIP Facility represents the

best option available to address the Debtors' immediate liquidity needs and that the terms and

conditions of the DIP Facility are reasonable and appropriate under the circumstances.

20. Accordingly, I believe that the proposed DIP Facility should be approved on the

terms and conditions described in the DIP Credit Agreements and the Interim DIP Order and that

entry into the DIP Facility reflects a sound exercise of the Debtors' business judgment.

Dated: March 5, 2025

/s/Jonathan Tibus

Name: Jonathan Tibus Title: Managing Director

8